# THE ROLE OF AMYLOID ENHANCING FACTOR IN THE DEVELOPMENT OF AMYLOID A AMYLOIDOSIS

Caroline L. Reid Department of Medicine Division of Experimental Medicine McGill University, Montreal August, 1993

A Thesis submitted to the Faculty of Graduate Studies and Research in partial fufillment of the requirements for the degree of Master of Science.

© Caroline L. Reid, 1993.

Shert Title of Thesis

· .

Amyloid Enhancing Factor in Development of Amyloid A Amyloidosis

#### ABSTRACT

i

Secondary, or AA, amyloidosis is a systemic disease characterized by the extercellular ussue deposition of insoluble, fibrillar amyloid A protein. Aberrant must thism of serum amyloid A protein by reticuloendothelial cells in the presence of an where enhancing factor (AEF) is thought to result in the accumulation of fibrils within the Treatment of nuce with AEF, in conjunction with an inflaminatory stimulus, induce  $\mu$  doid deposition within 48-72 hours. In viso examination of the effect of AEF on spicel. macrophage activation-associated phenotype and functions reveal that AFF has no apparent of fect on the ability of spleen and liver macrophages to phagocytose or kill Listeria moder togenes. It did appear to block enhanced respiratory burst function as well as rest rectrophoretic purification of AEF from crude amyloidotic tissue Productive in  $U_{1} = W_{1} + W_{2} + W_{1}$  Factivity to a region apparently consisting of 3 protein species with ging from 63.2 KDa to 47.9 KDa. Ex vivo studies of macrophage mondate A day ture to a product eluted from this "AEF-active" region did not indicate an inhibition of respiratory being that the action of AEF may be dependent on the presence of additional, possibly inflammation-associated, component(s).

#### ABSTRACT

L'amyloidose secondaire de type AA est une maladie systémique charactérisée par un dépôt tissulaire extracellulaire d'une protéine insoluble et fibrillaire, l'amyloide A. L'accumulation de ces fibrilles résulterait d'un métabolisme aberrant de la protéine précurseur, le "serum amy loid A", par les cellules du système réticuloendothélial en présence de "Amyloid Enhancing Factor" (AEF). Le traitement de souris avec l'AEF en parallèle avec l'induction d'une réponse inflammatoire, provoque l'apparition d'amyloidose en 48 à 72 heures. Nous avons étudié l'effet de l'AEF in vivo sur l'activation du macrophage tant au point de vue phénotypique qu'au point de vue fonctionnel. Les fonctions phagocytaire et bactéricide des macrophages spléniques et hépatiques ne sont pas affectées par l'AEF. Ce dernier, par contre, bloque l'induction du burst respiratoire et provoque une diminution de nombre de cellules exprimant l'antigène majeur d'histocompatibilité de type II à leur surface. La purification de l'AEF à partir de l'électrophorèse de l'extrait tissulaire sur gel de polyacrylamide a circonscrit l'activité AEF à une région comprise entre 63.2 et 47.9 Kd, obtenue par électroelution à partir du gel. Cette région montrant une activité AEF n'inhibe pas le burst respiratoire des macrophages in vitro, tel que démontré in vivo, suggérant que l'action AEF peut être dépendante de la présence d'autres facteurs possiblement associés au processus inflammatoire.

ii

#### PREFACE

The experimental work presented in this thesis appears in chapters 2, 3, and 4. The following option, as described in the "Guideline's Concerning Thesis Preparation" for the Faculty of Graduate Studies and Research at McGill University, has been used.

"Candidates have the option, subject to the approval of their Department, of including, as part of their thesis, copies of the text of a paper(s) submitted for publication, or the clearly-duplicated text of a published paper(s), provided that these copies are bound as an integral part of the thesis. If this option is chosen, connecting texts, providing logical bridges between the different papers, are mandatory.

The thesis must still conform to all other requirements of the "Guidelines Concerning Thesis Preparation" and should be in a literary form that is more than a mere collection of manuscripts published or to be published. The thesis must include, as separate chapters or sections: (1) a Table of Contents, (2) a general abstract in English and French, (3) an introduction which clearly states the rationale and objectives of the study, (4) a comprehensive general review of the background literature to the subject of the thesis, when this review is appropriate, and (5) a final overall conclusion and/or summary.

Additional material (procedural and design data, as well as descriptions of the equipment used) must be provided where appropriate and in sufficient detail (e.g. in appendices) to allow a clear and precise judgement to be made of the importance and originality of the research reported in the thesis.

In the case of manuscipts co-authored by the candidate and others, the candidate is required to make an explicit statement in the thesis of who contributed to such work and to what extent; supervisors must attest to the accuracy of such claims at the Ph. D. Oral defense. Since the task of the Examiners is made more difficult in these cases, it is in the candidate's interest to make perfectly clear the responsibilities of the different authors of co-authored papers."

As required by the Guidelines, Chapter 1 is a general introduction and review of literature relevant to the subject matter of the thesis.

Aberrant amyloid fibril precursor protein metabolism associated with macrophages at the site of amyloid deposition is believed to play an integral role in amyloid fibril formation. Amyloid enhancing factor has been identified in all types of amyloid tissue tested to date due to its ability to induce amyloid fibril deposition at an accelerated rate. Though its precise chemical nature and mechanism of action have not yet been characterized, most work seems to suggest that it exerts its action at the level of the macrophage. Chapter 2 describes experimental data investigating the effects of amyloid enhancing factor on macrophage activation and function during amyloid enhancing factor-induced amyloidogenesis which has been published in the *Journal of Leukocyte Biology*, volume 53, July 1993.

The unavailability of definitive biochemical characterization of amyloid enhancing factor has hampered investigations regarding its seemingly universal tole in amyloid ogenesis in particular mechanism of action. Chapter 3 describes unpublished experimental data concerning the purification of amyloid enhancing factor.

Chapter 4 describes macrophage functional assays involving treatment with the purified amyloid enhancing factor prepared in chapter 3 in an attempt to link results obtained in chapter 2 using a more homogeneous preparation of amyloid enhancing factor.

Chapters 2, 3 and 4 are each complised of an Introduction, Materials and Methods, Results, Discussion, and References. Additional conturbors of chapter 2 are Lise Hébert, Gina Pozzulo, and Francine Gervais. Lise Hébert provided the data concerning splenic macrophage O<sub>2</sub><sup>+</sup> production (not including O<sub>2</sub><sup>+</sup> production data described in chapter 4) Gina Pozzulo provided assistance in analysis of la surface expression FACScan data. Francine Gervais is my supervisor. Chapter 5 includes a general discussion summarizing the conclusions of the thesis, and lists claims to original research. Appendix I contains an actual copy of the manuscript published in the *Journal of Leukoc vte Biology*, volume 53, July 1993 entitled Spleme Macrophage Activation and Functions in Amyloid Enhancing Factor-Induced Secondary Amyloidosis. Study of Phagocytosis, Killing, Respiratory Buist, and MHC Class II Surface Expression. AA amyloidosis is referred to as secondary reactive systemic amyloidosis, or simply secondary amyloidosis, throughout this manuscript.

# TABLE OF CONTENTS

	Page			
Chapter 1: General Introduction and Literature review	1			
Introduction	2			
Amyloid and Amyloidosis				
Amyloid Deposits	2			
Classification of Amyloids and Amyloidosis	3			
Amyloid (AA) Amyloidosis	5			
Experimental Induction of AA Amyloidosis	7			
Serum Amyloid A Protein: The AA Fibril Precursor				
Process of AA Amyloid Deposition	10			
Common Elements of Amyloid Deposits				
Serum Amyloid P Component	11			
Proteoglycans and Glycosaminoglycans	13			
Apolipoprotein E				
Amyloid Enhancing Factor (AEF)	15			
Reticuloendothelial Cells in Amyloidosis	18			
Conclusion	21			
References	26			
Chapter 2: Splenic Macrophage Activation and Functions in Amyloid				
Enhancing Factor-Induced AA Amyloidosis	40			
Abstract	41			
Introduction	42			
Materials and Methods	4.1			
Animals	44			
Preparation of AEF and Induction of Amyloidosis				
Detection of AA Amyloid Deposits	44			
In Vivo Phagocytosis and Bacteriacidal Activity	44			
Spleen Macrophage O <sub>2</sub> - Production	45			
Ia Surface Expression	45			

۷

Results	46		
In Vivo Phagocytosis and Bacteriacidal Activity During AEF-			
Induced AA Amyloidosis	46		
O <sub>2</sub> <sup>-</sup> Production in AEF-Induced AA Amyloidosis			
Ia surface expression During AEF-Induced AA Amyloidosis			
Discussion			
Acknowledgements			
References	57		
Chapter 3: Purification of Amyloid Enhancing Factor	59		
Introduction	60		
Materials and Methods	60		
Animals			
Induction of AA Amyloidosis			
Detection of Amyloid Deposits			
Preparation of AEF Extract			
Preparative Gel Electrophoresis			
Electroelution			
Two Dimensional (2-D) Electrophoresis	63		
Results	64		
SDS-PAGE Analysis of AEF Extract	64		
Isolation of Region Containing AEF Activity	65		
2-D Gel Electrophoretic Resolution of the AEF-Active Region			
Discussion			
References	73		
Chapter 4: Macrophage Respiratory Burst Function in the Presence			
of Purified Amyloid Enhancing Factor	75		
Introduction	76		
Materials and Methods	76		
Animals	76		
Cells	76		
Measurement of Superoxide (O <sub>2</sub> <sup>-</sup> ) Production	77		
Results	77		
Discussion	80		
References			

٧I

Chapter 5: General Discussion and Conclusions	85
Discussion	86
References	80
Claims to Original Research	91
Acknowledgments	92
Appendix 1	AI

# TABLE OF FIGURES

	Page
Figure. 1.1:	
The Fibrillar B-Pleated Sheet Conformation of Amyloid Deposits	4
Table 1.1:	
Nomenclature and Classification of Amyloid and Amyloidosis	6
Figure. 1.2:	
The Biochemical Basis of The Respiratory Burst	21
Figure. 1.3:	
The Pathogenesis of Amyloidosis	25
Figure 2.1:	
In Vivo Phagocytosis and Bacteriacidal Activity	47
Table 2.1:	
O2 <sup>-</sup> Production by Macrophages in AEF-Induced AA Amyloidosis	49
Figure 2.2:	
Modulation of Spleen Macrophage Ia Expression	50
Figure 2.3:	
Restoration of Spleen Macrophage Ia Expression	52
Figure 2.4:	
Detection of Amyloid Deposition by Congo Red Stain	53
Table 2.2:	
Incidence of Amyloid Deposition in Spleen Sections	54
Figure 3.1:	
Separation of AEF Extract on Preparative gels	66
Table 3.1:	
Incidence of AEF Activity Indicated by Amyloid Deposition	67
Figure 3.2:	
Two-Dimensional Resolution of the AEF-Active Region	69
Table 4.1:	
Oxygen Radical Production by Peritoneal Macrophages	78
Figure 4.1:	
Oxygen Radical Production by Peritoneal Macrophages	79

viii

CHAPTER 1 GENERAL INTRODUCTION AND LITERATURE REVIEW 1

#### INTRODUCTION

The term amyloidosis describes a heterogenous collection of systemic diseases characterized by the extracellular tissue distribution of proteinaceous amyloid fibrils of varied chemical diversity. Diseases involving amyloid deposition cover a diverse array including chronic renal dialysis, lipoprotein metabolism and artherosclerosis, inflammatory disorders, cancer, rheumatoid arthritis and Alzheimer's disease.

The mechanisms underlying the pathophysiological processing of normally innocuous, soluble precursor proteins of various amyloids are not well understood. Reticulo-endothelial cells are consistently found associated with amyloid deposits and are thought to play an integral role. In 1966, another key player, namely amyloid enhancing factor (AEF), was identified in amyloid tissue extracts (1). Recent studies demonstrate a causative role for AEF in amyloidogenesis and suggest that it is a common pathogenic link between the diverse forms of amyloidotic disorders (2-5). To date definitive biochemical characterization of AEF has remained elusive thereby limiting its identification to largely functional terms.

The work described here focuses on elucidating the biochemical nature of AEF as well as examining its possible cause and effect relationship to reticulo-endothelial cells found at the site of amyloid deposition.

# **AMYLOID AND AMYLOIDOSIS**

#### **Amyloid Deposits:**

Amyloidosis arises when normally innocuous, soluble proteins polymerize to form insoluble fibrils. The resulting mass of amyloid fibrils is associated with plasma and extracellular matrix proteins as well as proteoglycans. As these amyloid deposits increase in size, they invade the extracellular tissue spaces of vital organs thereby destroying normal tissue architecture and function.

The term amyloid, meaning "starch-like", was first devised in the 1800's by Virchow based on the positive iodine staining reaction exhibited by amyloid deposits of diverse origins (6). Despite the fact that Friedreich and Kekulé later demonstrated that the major component of amyloid deposits is protein (7), the term "amyloid" has endured. It is likely that the carbohydrates, such as heparan sulfate proteoglycan, commonly found in association with amyloid deposits are responsible for the cellulose-like iodine staining reaction first observed by Virchow.

A major focus of amyloid research during the past decade has been the characterization of the individual proteins which form amyloid fibrils. To date at least 15 distinct amyloid peptides and their precursors have been identified to form deposits within specific clinical conditions associated with amyloidosis (8). All amyloid deposits, regardless of which of these proteins is the principal component, share certain definitive tinctorial, morphological, and ultrastructural properties. Early researchers identified amyloid deposits by sulfuric acid and iodine staining techniques (6), metachromatic staining with aniline dyes such as methyl violet and crystal violet (9), and eventually Congo red stain (10). When analyzed by light microscopy without polarizing optics, amyloid deposits in tissue sections appear homogenous, eosinophilic and amorphous (11) With polarizing optics, unstained amyloid deposits exhibit birefringence, indicating the presence of highly ordered structures (12). Upon staining with Congo red, this birefringence is increased intensely, with a characteristic apple green colour, in a manner not seen with other fibrillar structures (i.e. collagen) (12). Typically the presence of amyloid is detected using Congo red stain in conjunction with polarization microscopy.

Electron microscopic study of amyloid deposits have shown a fibrillar structure consisting of fibrils 7-10 nanometers in diameter (13, Fig. 1.1). Infrared and X-ray diffraction spectroscopy of isolated amyloid fibrils indicate that the proteins are arranged in a cross-ß-pleated sheet conformation (14) which is not normally present in non-pathogenic fibril polymers. The tinctorial and optical properties of amyloid fibrils (once stained with Congo red) as well as their physiological inertness within the tissues can be attributed to this ß-pleated sheet conformation (15, 13).

#### Classification of Amyloids and Amyloidosis

From Virchow's initial description of amyloid deposits until the early 1970's, popular theory held that all amyloid deposits were comprised of a common major substance. The classification of systemic amyloid syndromes therefore evolved based on the anatomic distribution of amyloid deposition and the associated clinical findings. Systemic amyloidotic disorders were usually grouped under either "secondary", "familial", or "primary" headings in addition to the rarer localized forms of amyloidosis. With the development of methods for dissolving and fractionating amyloid fibrils extracted from tissues (16, 17), a number of chemically distinct amyloid fibril proteins were identified. These findings made it apparent that grouping amyloidotic disorders according to their clinical nature alone is inappropriate.

At present it is the protein nature of the physiologically insoluble amyloid fibril component of the amyloid deposit which defines the type of amyloidosis and determines the pathogenesis of the disease which results from its deposition. The current nomenclature and classification of amyloids and amyloidosis (based on the chemical nature of the fibril

3



# Fig. 1.1 THE FIBRILLAR, B-PLEATED SHEET CONFORMATION OF AMYLOID DEPOSITS.

A Twisted 8-pleated anti-parallel configuration of paired filaments of an amyloid fibril in AA amyloidosis. Sites of binding of the planar Congo red dye are indicated by the stippled blocks. (Adapted from Glenner 1980).

B Congo red staming of a histiological section prepared from the spleen of a mouse with experimentally-induced AA amyloidosis (by repeated casein injections). Amyloid A deposits are identified by their characteristic birefringence under polarization microscopy.

B

A

protein), as devised at the VIth International Symposium on Amyloidosis in 1991, is summarized in Table 1.1.

#### Amyloid A (AA) Amyloidosis

Two observations have led to the theory that the nature of the pathogenic mechanism resulting in the formation of an amyloid deposit is the same regardless of the biochemical character of the deposit or the clinical setting in which it occurs (18). These observations are that a variety of proteins with fundamentally different amino acid sequences form assue protein deposits with the same morphological, ultrastructural, and tinctorial properties, and that amyloid deposits with identical proteins occur in seemingly unrelated clinical settings. The elucidation of the pathological processes culturinating in an amyloid deposit, regardless of its protein composition, is therefore pertinent to all disorders characterized by amyloid deposition. The studies described here are based on amyloid A, or AA, type amyloidosis experimentally induced in mice and are aimed at elucidating the mechanisms which result in amyloid deposition. The reported findings are therefore relevant to all types of amyloidosis as well as AA amyloidosis in animals and man.

AA amyloidosis (previously termed secondary reactive systemic amyloidosis) is characterized by the deposition of the AA protein usually consisting of 76 amino acids and a molecular weight of 8500 (19-21). The AA precursor molecule, termed serum amyloid A (SAA), circulates in the serum bound to high density lipoprotein (HDL). Although its exact physiological function is unknown, it behaves as an acute phase reactant exhibiting a significant increase in serum concentration levels in clinical situations which produce inflammation (22). It is not surprising then that AA amyloidosis occurs as a complication of chronic inflammatory disorders (such as rheumatoid arthritis), chronic infective disorders (i.e. leprosy, osteomyelitis, and tuberculosis), familial disorders characterized by recurrent inflammatory episodes, and certain malignancies.

Recently, Familial Mediterranean fever (FMF), an autosomal recessive disorder characterized by recurrent episodes of fever, peritonitis, arthritis, and pleuritis as well as a high incidence of AA amyloidosis, has been linked to the alpha-globulin locus on chromosome 16p (23). Other genes involved in the inflammatory process which map to this region include the integrins (a cluster of genes encoding the alpha-subunits of leukocyte surface adhesion receptors). Mutations of one of these genes (i.e. resulting in overadhesiveness) may explain the episodic inflammatory process in FMF. It is unlikely however, that AA amyloidosis associated with FMF is due directly to the identified chromosome 16p locus. Distinct ethnic groups exhibiting major differences in incidence

5

Amyloid Protein	Protein Precursor	Clinical Setting
АА	apoSAA	Reactive (secondary) Familial Mediterranean Fever Familial amyloid nephropathy with urticaria and deafness (Muckle-Wells' syndrome)
AL	kappa, lambda e.g. kIII	Idiopathic (primary), myeloma- or macroglobulinemia-associated
AH ATIR	IgG1 Transthyretin	Familial amyloid polyneuropathy, Portugese Familial amyloid cardiomyopathy, Danish Systemic senile amyloidosis
AApoA1	apoA1	Familial amyloid polyneuropathy, Iowa
AGel	Gelsolin	Familial amyloidosis, Finnish
ACys	Cystatin C	Hereditary cerebral hemmorhage with amyloidosis, Dutch
АВ	ß protein precursor e.g. BPP 695	Alzheimer's disease Down's syndrome Hereditary cerebral hemmorrhage with amyloidosis, Dutch
AB2M	<b>B2-microglobulin</b>	Associated with chronic dialysis
AScr	Scrapie protein precursor 33-35, cellular form	Creuzfeld-Jacob disease etc. Gerstmann-Straüssler-Scheinker syndrome
ACal	Procalcitonin	In medullary carcinomas of the thyroid
AANF	Atrial natriuretic factor	Isolated atrial amyloid
AIAPP	Islet amyloid polypeptide	In islets of Langerhans Diabetes type II, insulinoma
AIns	Insulin	Islet amyloid in the degu (a rodent)
AApoII	apoAII (murine)	Amyloidosis in senescence accelerated mice

# Table 1.1 Nomenclature and Classification of Amyloid and Amyloidosis

'Note: "Amyloid Protein" denotes the major protein component of the amyloid deposit.
Abbreviations-AA: amyloid A protein; SAA: serum amyloid A protein; apo: apolipoprotein; L: immunoglobulin light chain; H: immunoglobulin heavy chain.
BPP 695 indicates the 695 amino acid residue protein.
Scrapie protein precursor 33-35 indicates the 33-35 kilodalton (KDa) forms.
Neither AIns nor AApoII are found in humans. of AA amyloidosis demonstrate homogeneity, at least with respect to chromosomal location, of the locus responsible for FMF (23).

AA amyloidotic disorders are characterized by a common pattern of organ involvement. Amyloid A (AA) fibrils accumulate primarily within the spleen, liver, and kidneys, followed by possible deposition in the adrenal glands, gastrointestinal tract, lymph nodes, pancreas, prostate, thyroid, heart, and even the lungs (24, 25)

# **Experimental Induction of AA Amyloidosis**

Most of the known forms of amyloidosis have only been identified in humans on whom research is limited for obvious ethical reasons. Though there are relatively few developed experimental models of amyloidosis the animal equivalent of amyloid A (AA) amyloidosis seen in association with inflammation is well established and has proven instrumental in the study of human AA amyloidosis. AA amyloidosis can be induced in a variety of species, with the exception of rats, using protocols involving chronic parasitic infection or multiple injections of inflammatory agents (26, 27). Since its original development in 1922 (28, 29), daily parenteral injections of mice with 10% casein in a 0.3 M sodium bicarbonate solution has provided a reliable and reproducible method of inducing AA amyloidosis and has been shown to involve the deposition of AA fibrils structurally similar to those involved in the human disease (30).

Marked genetic variation in resistance and susceptibility to induction has been observed between inbred strains of mice with CBA/J and C57Bl/6J strains among the most susceptible and A/J among the most resistant (31, 32). Furthermore, genetic analysis using segregating populations of mice derived from a cross between CBA/J and A/J mice has suggested that susceptibility to amyloidosis can be controlled by a single dominant gene (33).

In keeping with the contemporary theory that amyloid resulted from immunological derangement, the amvloid inducing-properties of casein were previously believed to be dependent upon its antigenicity (34). Current hypotheses recognize that a basic immunological disturbance could not account for the diversity of proteins seen as amyloids in the different diseases (35). It is likely that casein's amyloid-inducing capacity is related to its relatively large molecular weight (36) and ability to cause significant inflammation leading to increased SAA serum concentrations. It has been suggested that lipopolysaccharide (LPS) contamination may be the mechanism by which casein solutions promote amyloid deposition (37, 38). Analysis of the LPS content of the casein solution prepared in our laboratory however has revealed very low levels of contamination. Furthermore, it has been shown that orally ingested milk proteins enhance amyloid

deposition (28) even though LPS is degraded in the gut. Finally, C3H/HeJ mice (a low LPS responder strain) develop amyloid when treated with casein (32).

#### Serum Amyloid A Protein: The AA Fibril Precursor

SAA was originally identified in patients with and without amyloidosis as a serum protein sharing antigenic determinants with the fribrillar tissue AA protein (39). Subsequent comparative amino acid analysis of both AA and SAA proteins revealed identical amino acid sequences beginning at the the N-terminal ends, with the AA protein being shorter at the C-terminal end (40). These observations suggest that cleavage of an SAA precursor molecule at its carboxy-terminal end could result in the formation of the truncated AA protein. Indeed the transfer of human acute phase HDL into amyloidotic mice has resulted in human SAA being incorporated into mouse amyloid tissue (41).

As mentioned previously, the SAA protein has been identified in a variety of species as an acute phase response (APR) apolipoprotein which, like most APR proteins, is synthesized principally by the liver. During an APR, SAA associates with HDL, in particular the HDL3 fraction, of which it can become a predominant apolipoprotein exceeding apolipoprotein A-I (apo-A-I) in molar ratio (42).

In all species studied to date, SAA is not a single protein but rather a family of several related proteins. The SAA genes reveal strong cross-species conservation among vertebrates and, in humans and mice, are clustered in small homologous regions on chromosomes 11p and 7 respectively (43, 44). In mice three genes (SAA1, 2, and 3) as well as a pseudogene (45), have been characterized. Although SAA1 and SAA2 show 96% homology, SAA3 is substantially divergent. All three genes are transcribed (46) however only SAA1 and SAA2 mRNA products give rise to dramatically elevated levels of hepatic mRNA and circulating, HDL-associated protein during APR. While SAA3 does contribute moderately to hepatic mRNA production, it yields no detectable, translated plasma or tissue product. The liver and, to a lesser extent, the gastrointestinal tract, produce the majority of SAA1 and SAA2 (47). It has been shown that tissues other than the liver and the gastrointestinal tract can, however, express SAA3 mRNA (48).

Three members of the human SAA gene family containing transcribable and translatable sequences have been identified (49, 50) as well as a fourth pseudogene (51). Three hepatic mRNA species code for the six major isoforms purified from the HDL3 plasma fraction of individuals undergoing APR (52, 53). However, the gene family probably contains considerable polymorphism and, more recently, additional allelic and nonallelic variants have been reported (54, 55).

Much work has been done, in both humans and mice, showing that during an inflammatory reaction cytokines produced by activated macrophages, namely interleukin-1 (IL-1), interleukin-6 (IL-6), and tumor necrosis factor, are responsible for regulating the transcription of SAA genes in the liver (56, 57, 58). The magnitude of apo-SAA synthesis appears to be controlled by cytokines acting synergistically (59, 60) involving both NF  $\beta$  and IL-6 response elements identified within the SAA promoter region (61). The mechanisms of synergistic stimulation of SAA production appears to involve enhanced mRNA transcription and subsequent SAA mRNA stabilization (62, 63). During disturbances in homeostasis plasma concentrations of SAA can increase from 100 to 1000 fold within 18-24 hours in mice (22) or 48-72 hours in man (56). During increased secretion SAA can comprise up to 20% of total murine HDL apoproteins (64).

Relatively little information is available regarding the biological significance of apo-SAA. It has been shown to modulate HDL lipid metabolism (65) during an APR, inhibit platelet function (66), exert immunosuppressive activity (67), and inhibit IL-1 and TNFinduced fever (68). In addition, rabbit synovial fibroblasts have been found to produce an SAA-like protein with capacity for autocrine modulation of collagenase production (69) A similar protein, GSAA1, has been described at the gene level in humans (70). It is possible that SAA functions physiologically as a regulatory protein inhibiting a variety of acute phase responses and/or regulating the removal of specific constituents from sites of tissue destruction.

Due to the amino acid sequence identity between murine apo-SAA2 and murine AA fibrils, apo-SAA2 is generally believed to the be the sole precursor of AA fibrils in mice (71). In the first 25 amino acid residues, murine SAA1 and SAA2 differ at only two positions (45, 71). It has been shown that synthetic peptides corresponding to the N-terminal 11 amino-acid-residue segment of SAA2 form amyloid-like fibrils *in vitro*, while the corresponding segment of SAA1 do not. This finding indicates that differences in the N-terminal portion of SAA determine the amyloidogenicity of murine SAA isoforms (72).

It has been suggested that SAA2 is rapidly and selectively removed from the circulating pool of SAA1 and SAA2 due to a dramatic decrease in plasma SAA2/SAA1 ratios (concurrent with a fairly constant SAA2/SAA1 mRNA ratio) during a 20-day course of murine amyloid induction using casein (73). Research done in our lab however did not indicate a selective decrease in serum apo-SAA2 before 38 days of casein injections, therefore well after amyloid deposition had been initiated (74). The relative time difference between these two studies may be due to the fact that the former study (73) measured SAA isotypes by isoelectric focusing versus SDS-PAGE followed by densitometry in the latter (74). It is possible that isoelectric focusing allows for better separation of the two SAA

isotypes leading to a better assessment of the ratio cí each apoprotein. Furthermore, the apoSAA2 isotype is more easily displaced from its HDL carrier than apoSAA1 (75) and elements such as hpopolysaccharide and apoproteins AI and AII have been shown to compete with SAA for binding to the HDL carrier (75-77). Differential binding affinities between the two SAA isotypes as well as competition for binding with other factors may well influence apoSAA ratios as detected by different methods. For example, research was conducted in our lab using mice bred and maintained in specific pathogen free facilities. Reduced concentrations of circulatory lipopolysaccharide may well delay the selective removal of apoSAA2. Should this be the case, it would appear that amyloid deposition occurs regardless of whether apoSAA2 is selectively removed from the circulation.

Though some studies have reported that human SAA1 gene products account for the vast majority (80%-100%) of the AA protein isolated from any one individual (78), it is generally believed that preferential deposition of particular isoforms does not occur in humans. AA-proteins isolated from various individuals have been found to vary from 94 to 43 residues (positions 2-44) indicating that an amyloidogenic region is the N-terminal part of the molecule (79). No N-terminal variation is seen between the different human SAA isoforms possibly explaining why especially amyloidogenic human SAA variants have not yet been clearly indicated (72).

## **Process of AA Amyloid Deposition**

The development of AA amyloidosis has been described as involving two phases: the pre-amyloid phase and the amyloid phase (80, 81). The first phase is comprised of sequential host responses to tissue injury and cell necrosis (i.e. an APR) including macrophage activation and cytokine production, transient hepatic SAA production and association of apo-SAA with HDL3 ultimately resulting in elevated levels of circulating SAA. The proceeding amyloid phase is characterized by SAA precursor processing into AA molecules. Fibrils are assembled from filaments of AA molecules which, together with other constituents, accumulate as bundles of fibrils which constitute fibrous deposits within the tissues. Lateral aggregates of fibrils and filaments have been observed in proximity to cells, oriented perpendicularly to, and merging with the cell membrane. Amyloid fibrils are observed as bundles near cell membranes and more disgursed at greater distances from cells. Organ deposition in AA amyloidosis occurs primarily within the spleen, liver, and kidneys. The earliest site of deposition is the perifollicular area of the spleen in close proximity to phagocytic cells, around the follicles, and extending outwards to the red follicular pulp (82). Hepatic amyloid deposits are initially detectable around Kupffer cells and parenchymal cells. In the kidney, deposition initiates within the mesangial region glomeruli then spreads to the basement membrane, endothelial cells, and renal medulla

In addition to the principal amyloid protein component of amyloid deposits other constituents have been found to be included in amyloid deposits of all varieties. At some point during the process of amyloid formation, these factors, most notably amyloid P protein, glycosaminoglycans, and AEF become involved resulting in their inclusion within the deposit. The importance of these components will be considered below

## **COMMON ELEMENTS OF AMYLOID DEPOSITS**

Besides sharing common, definitive structural properties, all types of amyloid deposits tested to date have been found to include both amyloid P protein (AP), highly sulphated glycosaminoglycans (GAGs) and apolipoprotein E (apo E)

# Serum Amyloid P Component:

During the identification of the various amyloid constituent proteins, an alphaglobulin was found associated with amyloid fibrils. Once isolated, this plasma component, termed amyloid P component (AP), was determined to be a pentagonally structured protein distinct from amyloid fibrils and identical to a normal circulating plasma constituent which was then called serum amyloid P component (SAP) (83). The terms AP and SAP therefore simply denote the different locations (either associated with amyloid tissues or circulating in the serum respectively) of the same protein. Though initially suspected to be absent from intracortical and intraneuronal amyloid deposits characteristic of Alzheimer's disease (AD) (84), AP immunoreactivity has now been detected in neurofibrillary pathology of AD and other neurodegenerative disorders characterized by amyloidotic plaques (85). AP has been detected in all other types of amyloid tested to date.

Studies done in AA amyloidosis in mice have given direct evidence that circulating serum SAP is indeed the precursor of AP within anyloid deposits. In short, when an exogenous source of either human or murine <sup>125</sup>I-abelled SAP is injected into AA amyloidotic mice, it is deposited within the liver, spleen and kidneys in a concentration directly proportional to the extent of AA fibril deposition (86). AP is stacked on AA deposits, therefore is not an integral part of the amyloid fibril, and may amount to 10-15% of the total AA fibril mass. Its deposition has been found to be coincidental with the appearance of AA fibrils (86).

SAP, human C-reactive protein (CRP) and other animal homologues such as Syrian hamster female protein (FP) are all members of a family of pentraxin proteins characterized by a discoid conformation of 5-10 noncovalently bound subunits arranged as two stacked pentraxins. Though a considerable amount of sequence homology exists between these

proteins they remain immunologically distinct SAP has been found in all vertebrate species investigated including the mouse, rabbit, pig, goat, sheep, donkey, human, cow, marine toad, plaice, flounder, and dogfish (87). Structurally, human AP consists of 10 identical alpha-glycoprotein subunits each of approximately 23 000 with a total molecular weight of 230 000 (88, 89). Murine SAP monomers migrate at a relative molecular weight of 31 000 (90) and possess approximately 10% carbohydrate (91).

Synthesis of SAP occurs in the liver. In man, the half-life of AP is 7.8-8.5 hours, suggesting a rapid continuous production to maintain a stable and significant serum level of 40-50 ug/ml (92). SAP is increased slightly in chronic active disorders including certain malignancies and rheumatoid arthritis (93).

In mice, SAP acts as an acute phase protein (90). There are marked, genetically determined differences in resting SAP levels within inbred strains of mice yet all strains appear to reach the same serum concentration (~170-240 ug/ml) within 24-48 hours after an inflammatory stimulus (94). *In vivo* and *in vitro* studies examining the effects of IL-1 (95), IL-6 (96), and TNF (97) have established that these molecules are involved in the mediation APR protein induction. Crude macrophage supernatants and purified IL-1 have been shown to regulate SAP synthesis specifically (98) and recombinant IL-6 and IL-1 are known to exhibit synergistic induction of hepatocyte SAP synthesis *in vitro* (99). Hepatocyte cultures exhibit sensitivity to cyclohexamide treatment with respect to SAP synthesis, yet relative resistance with respect to SAA synthesis, suggesting that different regulation by protein factors occurs in each case (100). Inhibition of SAP synthesis by actinomycin-D and colchicine indicates that regulation may occur both pre-translationally and during secretion (101).

While the biological function of SAP is not known, the stable evolutionary conservation of overall molecular structure and binding specificty within its family of pentraxin proteins suggests that it plays an important physiological role (56). AP has been reported to be a component of the microfibrils that are the structural units of elastin fibers in connective tissues (102) as well as a normal matrix glycoprotein of glomerular basement membrane (103). In addition, SAP has been shown to bind amyloid fibrils (104), fibronectin (105), heparan sulphate and dermatin sulphate (106), and C4-binding protein (105) in a Ca<sup>2+</sup>-dependent manner and macrophages via their mannose 6-phosphate receptor (107). It possesses a high affinity, Ca<sup>2+</sup>-dependent binding site for the 4,6-cyclic pyruvate acetal of the galactose residues present in agarose as the galactan homopolymer and in the cell walls of a few bacteria (108). SAP has also been reported to interact with immobilized single or double-stranded DNA in a Ca<sup>2+</sup>-dependent manner (109). Furthermore, SAP is able to displace H-1 and interact with chromatin (110). It has been proposed that SAP may solubilize extracellular DNA and help clear it from the circulatory system. In addition, SAP undergoes self-aggregation in the presence of calcium (111) Heparin and other glycosaminoglycans have been shown to inhibit this polymerization (106). It has therefore been suggested that heparin may influence the aforementioned interaction between C4-t inding protein and SAP *in vivo*, thereby altering the distribution of SAP and C4-binding protein in the serum. In this case SAP may act as an important regulatory protein in the complement and/or coagulation systems (111, 112)

The role that SAP may play in the pathological process of amyloid deposition has yet to be elucidated. Studies of differences in SAP induction in amyloid susceptible and -resistant inbred strains of mice indicate that sustained high levels of SAP are consistent with susceptibility (113). It has been postulated that AP associated with microfibrils at the periphery of elastic fibers may bind amyloid fibril material and encourage its deposition at this site (102). Alternatively, SAP may influence amyloid deposition by binding to fibrils thereby protecting them from enzymatic degradation. SAP has indeed been shown to inhibit the activity of enzymes including porcine pancreatic enzyme, and *Pseudomonas* and human elastase (114, 115).

#### Proteoglycans and Glycosaminoglycans .

Proteoglycans (PGs) are a diverse group of extracellular matrix proteins comprised of a glycosaminoglycan (GAG) attached to a protein core GAGs are sulphated carbohydrate moieties of which there are four types, namely heparin/heparan sulphate, chondroitin sulphate/dermatan sulphate, keratin sulphate, and hyaluronic acid (116) PGs are distinguished by their GAG content.

Highly sulphated GAGs have been demonstrated in all forms of amyloid examined to date (117). In AA amyloidosis, glycosaminoglycans and amyloid peptides have been shown to be deposited coincidentally (118) and in amounts directly proportional to the extent of amyloid deposition (119). This association between GAGs and AA amyloid deposits has been consistently observed regardless of the nature of inflammatory stimulus used to induce deposition, the organ site of deposition, or the length of time between induction and fibril deposition (118). Furthermore, studies have shown that the major type of proteoglycan occuring in AA deposits in experimentally induced amyloidosis is the basement membrane proteoglycan constituent heparan sulphate proteoglycan (HSPG) (120). Though cationic staining studies have been interpreted to show that both the protein core and the carbohydrate core are intimately associated with AA fibrils (121), alternative research comparing amyloid associated glycosaminoglycans before and after B-climination or enzymatic digestion seem to indicate that glycosaminoglycans associated with amyloid fibrils isolated from AA amyloidotic human liver do not appear as part of intact proteoglycans (122). HSPG has been described in at least five forms of amyloid including AA, AL, the B-amyloid of Alzheimer's disease, the prion amyloids in Creutzfeldt-Jacob disease, Gerstmann Straussler Syndrome, experimental scrapie, and the amyloid associated with type II diabetes (120, 123-125). In addition, the dermatan sulphate proteoglycan has been detected in AA amyloidosis and B-amyloidosis in Alzheimer's disease (122,126).

The significance of consistent codeposition of GAGs with amyloid fibrils remains a mystery. As previously mentioned, the SAA2 isoform is the sole precursor to AA amyloid in mice. Studies with different GAGs and murine SAA1 or SAA2 showed that only SAA2 and heparan sulphate interacted so as to markedly increase B-sheeting in SAA2 (127) The association of SAA2 with the GAG also protects the SAA2 from proteolytic digestion (128). This observation suggests that GAG or HSPG may play a role in both the pathogenesis and the structure of the amyloid fibrils. One potential way in which this might occur is through an ionic interaction between the amyloid peptide, or its precursor, and the sulphate moieties on the GAG. By sequence analysis, potential GAG binding sites have been identified on SAA (129).

# Apolipoprotein E

More recently, apolipoprotein E (apo E) has been detected in association with a wide variety of amyloid fibrils including AA,  $\beta$ -protein, scrapie protein, immunoglobulin light chain, and cystatin C amyloids (130, 131). Apo E is a 299 amino acid protein with a molecular weight of 34 000 (132). Besides its structural role in a number of lipoprotein particles, apo E is known to regulate lipoprotein metabolism. In addition apo E has been suggested to function as neurotrophic factor (133) as well as an immunoregulatory molecule (134). Apo E is produced by most organs with its greatest concentrations occuring in the liver and brain. A range of cell types, including macrophages, can produce apo E (132). Apo E is the principal lipoprotein in cerebrospinal fluid, where its levels have been observed to increase during certain central nervous system disorders (135). Apo E binds GAGs (136). It is possible that the consistent detection of apo E in amyloid deposits is simply the result of non-specific binding to GAGs.

At this point it is not clear whether the common presence of AP, GAGs, and apo E in such a variety of amyloids is the result of their integral role in the pathological process of amyloid deposition. Some investigators have felt that amyloid deposits represent a "sink" into which a variety of components including GAGs (137), and possibly SAP and apo E, are drawn. Indeed a diversity of proteins have been detected in some amyloids, for example vitronectin (138) or alpha-antichymotrypsin (139). To date, however, SAP, heparan sulphate, and apo E remain the sole non-amyloid protein constituents common to all amyloid deposits examined, regardless of the type of amyloid deposit or the clinical situation in which the amyloid deposit occurs. It has been shown that amyloid fibrils can be formed in vitro in the absence of AP and GAGs (11) suggesting that a fibrillogenic protein and appropriate environmental conditions could be the exclusive requirements for the formation of amyloid deposits. It is possible however, that AP, GAGs and apo E function as 'pathological molecular chaperones' (130). 'Molecular chaperones' are a newly recognized family of proteins, found in all cell types, that interact with other polypeptides specifically so that they assume a functional tertiary structure, preventing them from undergoing incorrect interactions (141). 'Pathological chaperones' have therefore been defined as a group of unrelated proteins that mediate B-pleated amyloid formation of polypeptide fragments without becoming integral components of the amyloid fibril (130). As such pathological chaperones AP, GAGs, and apo E may function by binding specifically to amyloidogenic protein surfaces that are formed during aberrant protein degradation thus preventing further proteolysis. Alternatively, they may be essential to establish or maintain conditions amenable to fibril formation or prevention of immediate resorption of fibrils before formation of bundles of fibrils and accumulation of amyloid deposits (129).

Much of the recent work concerned with elucidating the role of GAGs and SAP in amyloid deposition has focussed on the possible, fundamental involvement of the basement membrane. The basement membrane is a specialized form of extracellular matrix which provides an underlying structural support for epithelial cells as well as neuronal and mesenchymal elements (142). Basement membranes are composed of several major components including fibronectin, laminin, type IV collagen, and proteoglycans, each of which have been identified in AA amyloid deposits (143). Furthermore, AP is a normal constituent of basement membrane (103). Basement membrane disruption is observed in both AA amyloids and amyloid associated with Alzheimer's disease (144, 145). This basement membrane dysfunction, coupled with the inclusion of basement membrane components in amyloid deposits, has led to the postulation that amyloid peptides and/or their precursors could influence basement membrane organization or that basement membrane components could influence amyloid peptide organization into fibrils (146).

#### **Amyloid Enhancing Factor (AEF)**

It has become increasingly evident that a component isolated from amyloidotic tissue, termed amyloid enhancing factor (AEF), may also be included under the heading of "common elements of amyloid deposits". AEF was originally described in the context of AA amyloidosis (147), however its activity has now been identified in tissue extracts of at

15

least four forms of amyloid including AA, AL,  $\beta$ -amyloid, and transthyretin amyloid (148, 149) implicating its probable relevance in all types of amyloid. It has been postulated that AEF may act as a common pathogenic link between the diverse forms of amyloid (150).

Though definitive biochemical characterization of AEF has not yet been accomplished, AEF can be identified by its ability to induce amyloid deposition 24 to 48 hours following coadministration with a single inflammatory stimulus (in animals undergoing sustained inflammation, amyloid deposition is not usually detected before 15 days) (150). Treatment of experimental animals with AEF alone (i.e. in the absence of an inflammatory stimulus) causes neither inflammation nor amyloidogenesis. AEF activity has been detected in cell suspensions from whole or partial sections of spleen or lymph node, and spleen, liver, kidney, heart, and bone marrow extracts from pre-amyloidotic and amyloidotic animals (151-154, 27, 4). It has been extremely difficult to separate AEF activity from the AA protein (155), however neither purified amyloid proteins nor commercially purified GAG preparations possess AEF activity (3, 119). Furthermore, when mice are treated with AEF of human origin in conjunction with an inflammatory stimulus, the AA deposited is of murine origin (150). Additionally, when mice are injected with murine AEF and human SAA, the amyloid fibrils are of human origin (41). These observations support the concept that AEF is an entity separate from the major amyloid fibril protein component or the GAG component of amyloid deposits.

Studies have shown that the concentration of AEF increases markedly during inflammation (5) and in doing so alters the metabolic processing of at least one acute phase reactant, apoSAA (156). Most research with regards to the origin of AEF suggests that it is derived from and exerts its action at the level of the macrophage. Indeed, it is likely derived from a cell common to both the liver and the spleen (i.e. reticuloendothelial cells) since splenectomized animals can generate AEF activity in their livers upon appropriate stimulation (157). AEF appears approximately 24-48 hours before the histological detection of amyloid fibrils (3). The kinetics of amyloid deposition in experimental animals using standard induction protocols or following the passive transfer of AEF, are identical with the exception of the lag period necessary for the appearance of AEF when using standard regimens (157). The lag phase for AEF induction is obviated during passive transfer. AEF activity has been indicated to be maintained for at least four weeks following intravenous injection, prior to clearance by reticuloendothelial cells (4).

The importance of AEF in amyloidosis has been clearly demonstrated in studies of amyloidosis susceptible inbred strains of mice. For example, work done in our lab was able to show that, following standard AA amyloidosis induction, genetically amyloid resistant A/J mice did not exhibit amyloid deposition nor did their spleens contain detectable AEF activity. When these mice received AEF prepared from the spleens of amyloidotic, susceptible C57Bl/6 mice however, they developed amyloidosis at an accelerated rate. It was therefore deduced that resistance to amyloidosis in A/J mice can be attributed to their inability to produce sufficient AEF and that susceptibility to AA amyloidosis correlates with the increased concentration of AEF (158).

Methods of extraction of AEF activity vary from laboratory to laboratory. Isolation has been achieved by a variety of methods including homogenizing amyloidotic tissues in glycerol/TRIS-buffered solutions with further precipitation of AEF from the supernatant by dialysis against PBS (4), fractionation of the precipitated proteins by chromatographic methods (4, 154), precipitation of AEF activity from homogenized organs with ammonium sulphate followed by fractionation of activity with Sephadex, Biogel P-60 and benzamidine-Sepharose 6B affinity chromatography (5, 159), and primary extraction in high salt (KCl) concentrations (4). None of these methods have allowed for the complete purification of any specific component or components with AEF activity. The fact that AEF is relatively insoluble in physiological buffers has dramatically hindered the purification process. Work done in our lab has taken the purification of crude AEF extract prepared according to the first method described above one step further by solubilizing AEF activity in a 6% urea/1% B-mercaptoethanol/1 mM EDTA buffer amenable to further purification procedures.

Most research examining the nature of AEF has assumed that it is a single, unique molecule. AEF has been assigned a number molecular sizes ranging from 5 .5 kDa (160) to 55.1 kDa (L. Hébert, unpublished observations) the smallest being attributed to ubiquitin (160). Accumulating evidence indicating different molecular sizes with respect to various AEF preparations suggests that the AEF activity detected in each type of amyloid deposit may not be attributable to a single, common molecule but rather to a group of molecules, perhaps with unrelated physiological functions, but sharing some exceptional characteristics which lead to their AEF activity in each type of amyloidosis.

Reports have claimed that a substance with AEF activity, detected in amyloid-laden tissue from both AA amyloidotic mice and patients with Alzheimer's disease, is actually ubiqutin (160, 161). Immunoblotting of crude AEF extracts, prepared in our lab from each of these sources, did not indicate the presence of ubiquitin (unpublished observations). Furthermore, when mice received an i.v. injection of ubiquitin in conjunction with a s.c. injection of silver nitrate as an inflammatory stimulus, we did not detect splenic AA fibril deposition within 72 hours. Differing AEF extraction protocols or AEF activity assessment regimes may explain apparent ubiquitin-associated AEF activity. For example, the studies confirming ubiquitin's AEF activity (160, 161) involved mice undergoing an inflammatory response induced by s.c. silver nitate injections for 4 consecutive days. Animals were not sacrificed until 24 hours after receiving the final silver nitrate injection. Our AEF bioassay is conducted treating mice with a single s.c. injection of silver nitrate administered concomitantly to AEF injection. Animals are sacrificed 48 to 72 hours later. Continuous inflammatory stimulation of mice with multiple injections of silver nitrate may well result in the induction of some factor, such as endogenous AEF, which may use ubiquitin as a co-factor to produce accelerated AA amyloid deposition. Ubiquitin, when injected in this fashion, may then exhibit AEF-like bioactivity.

AEF has been proposed to promote amyloid deposition in a number of ways, none of which have been proven. As previously discussed, it is not clear whether selective removal of the AA amyloid precursor occurs during the development of amyloidosis in mice and does not appear to occur in humans. Data collected in our lab has shown there is no change in circulating serum apoSAA 1/apoSAA2 ratios following accelerated amyloid induction by injection of AEF. If proven conclusively, this would suggest that AEF does not affect the SAA2 precursor at a stage proceeding the normal association of acute phase reactants with their HDL carriers (74). One hypothesis suggests that AEF could act as a nidus for fibril deposition (119) which disrupts local GAG synthesis or clearance thereby perpetuating amyloid deposition. AEF has also been suggested to alter the activity of lysosomal glycosidases (162) or may itself be a leukocyte-derived thicl/serine protease (163). The clarification of the role of AEF in the development of amyloidosis as well as the elucidation of its biochemical nature in AA amyloidosis are the focus of this thesis.

# **RETICULOENDOTHELIAL CELLS IN AMYLOIDOSIS**

Fibril deposition in amyloidosis has been consistently observed in close association with reticuloendothelial cells (RE) *in situ* (164) suggesting that these cells play a significant role in fibril deposition. Both histological and experimental evidence points to the importance of phagocytic cells in amyloid fibril formation, particularily in inflammationassociated AA amyloidosis, in which mononuclear phagocytes are involved in the regulation of precursor production, formation of fibrils, and intraphagosomal digestion of fibrils (11, 165).

Normal peripheral blood monocytes are known to be capable of mediating SAA degradation *in vitro* through the action of cell surface enzymes. This degradative pathway is thought to involve initial proteolytic cleavage producing AA as an intermediate followed by complete proteolysis into smaller fragments (166, 167). Studies aimed at elucidating the exact relationship between RE cells and amyloid fibril deposition have led to the discovery that mononuclear leukocytes from healthy individuals are capable of mediating complete

SAA degradation whereas the cells of amyloidotic patients produce an intermediate protein fragment both the same size and immunologically cross reactive with AA (166). In addition, normal murine Kupffer cells have been found to degrade SAA completely, whereas Kupffer cells from amyloidotic animals produce the AA intermediate causing accumulation of the AA protein (168). These findings have led to the appropriate stat tissue amyloid deposition may be due to innappropriate or incomplete SAA degradation by cells of the monocyte/macrophage lineage, resulting in the accumulation of AA.

It has been suggested that a pathological modulation in macrophage-associated SAA degradation develops during the prolonged, pre-amyloid phase of standard AA-amyloid induction protocols. Induction of endogenous AEF is thought to occur during the preamyloid period (3). In keeping with this, treatment with an exogenous source of AEF in conjunction with an inflammatory stimulus would incur a more immediate macrophage modulation causing accelerated amyloid deposition. The theory that AEF may be the effector behind altered macrophage function ultimately resulting in AA deposition is supported by data comparing the kinetics of AA intermediate accumulation and the appearance of AEF activity (168, 3) in which detectable AEF activity necessarily precedes AA fibril formation. It is logical that a specific, pathological factor such as AEF must be present to cause incomplete SAA metabolism. If activated RE cells alone were responsible for the appearance of AA, then AA should be deposited in most inflammatory conditions and at the sites of inflammation (18). Amyloid deposition as a complication of inflammation, however, is rare and initial amyloid deposits, at least experimentally, are always found in precise anatomical locations within the spleen and liver but never at the sites of inflammation. In addition, both clinical and experimental observations show that serum SAA levels may be elevated for prolonged periods in the absence of amyloid deposition.

It has been established then that the mononuclear phagocyte system plays a central role in amyloidosis in that it provides cytokine signals that regulate expression of some amyloid precursors as well as the proteolytic enzymes that convert precursors to fibril proteins. It is therefore important to review the phenotypic and functional characteristics of the macrophage, particularily in the context of inflammation. Some of the work presented in this thesis involves the evaluation of the functional state of activated, splenic macrophages (including phagocytosis, bactericidal and respiratory burst activity, and MHC Class II expression) during AEF-induced AA amyloidosis Consequently, we will review some of the principles specifically related to those elements of tissue macrophage response

The mononuclear phagocyte (or reticuloendothelial cell) system represents a group of widely distributed, multifunctional, and morphologically heterogenous cells.

Originating from bone marrow stem cell precursors, all cells within this lineage pass through monoblastic then promonocytic developmental stages prior to release into the blood as monocytes. Following a short circulation period, monocytes migrate into tissues to form the macrophages of the lungs, liver, spleen, lymphoid organs, pleural and peritoneal cavities, bones and so on, where they undergo final, tissue-specific differentiation (169). In general, macrophages are key players in both the immune and inflammatory reponses. They respond to a variety of stimuli with aggregation, activation of movement, secretion, endocytosis, activation of respiration, and production of mediators including those that perpetuate inflammation, and are capable of phagocytosis and processing antigen (170, 171). During sustained inflammation macrophages become "activated", resulting in enhancement or initiation of certain functions including phagocytosis, bactericidal and respiratory burst activity, and antigen presentation (172-175). In the context of chronic inflammation, the activated macrophage is therefore relevant to the development of AA amyloidosis.

When mononuclear phagocytes encounter invading microorganisms or other appropriate stimuli, they undergo a "respiratory burst" in which they consume and convert oxygen into the potent microbicidal agents superoxide anion (O<sub>2</sub>), hydrogen peroxide  $(H_2O_2)$ , hydroxyl radical, and singlet oxygen. These oxidants are reactive enough to destroy most biological molecules and are responsible for much of the damage inflicted by phagocytes on both microorganisms and surrounding tissues at sites of infection or inflammation. The metabolic pathway responsible for reactive oxidant production is initiated within a few seconds of plasma cell membrane exposure to any one of a large number of stimuli, among the most effective of which are several that are likely to be present at inflammatory sites including opsonized microorganisms, complement fragment C5a, leukotriene B<sub>4</sub>, and N-formylated oligopeptides of bacterial origin. Activation of this membrane-associated pathway is characterized by an abrupt increase in oxygen uptake devoted to the production of a series of compounds including superoxide  $(O_2)$ , hydrogen peroxide (H<sub>2</sub>O<sub>2</sub>), and other oxygen-containing compounds, all of which are highly reactive. In addition, augmented oxidation of glucose occurs via the hexosemonophosphate shunt (HMPS) (176). The biochemical basis for the respiratory burst is described in in detail in Figure 1.2. Reactive oxidants are essential to the increased microbicidal capacity of activated macrophages (177).

In addition to its direct effector role in microbicidal destruction, the macrophage contributes to regulation of specific immune responses. This complex function includes

20

## Fig. 1.2. <u>THE BIOCHEMICAL BASIS OF THE RESPIRATORY BURST</u> Adapted from Babior *et al* (173).



A) The Hexose Monophosphate (HMP) Shunt. Glucose oxidation through the HMP s init accelerates during the respiratory burst due to increased production of NADP<sup>+</sup>, the availability of which limits the activity of the HMP shunt. NADP<sup>+</sup> is produced through the actions of (a) activated NADPH oxidase and (b) a glutathione dependent system that uses NADPH to detoxify  $H_2O_2$  as follows:

(a)  $2O_2 + NADPH$  NADPH Oxidase  $2O_2^* + NADP^+ + H^+$ (b) NADPH + H<sub>2</sub>O<sub>2</sub> Glutathione perovidase/reductase NADP + 2 H<sub>2</sub>O Each glucose metabolized via the HMP shunt reduces two molecules of NADP to NADPH via the following set of reactions:

**Glucose-6-P + NADP** glucose-6-P dehydrogenase **6-phosphogluconate + NADPII 6-phosphogluconate + NADP** 6-phosphogluconate dehydrogenase

#### Ribulose-5-P + $CO_2$ + NADPH

Thus NADPH necessary for the continued operation of the respiratory burst is replenished

**B)** The biochemical basis for the respiratory burst is the activation of NADPH oxidase dormant in resting cells, which catalyzes  $O_2$  reduction at the expense of NADPH Oxygen consumption and  $O_2^-$  production occuring during the respiratory burst are accounted for entirely by this reaction.  $H_2O_2$  is then produced primarily via enzymatically catalyzed dismutation of  $O_2^-$  by superoxide dismutase.

C) The reactive oxidants responsible for most oxygen-mediated damage by phagocytes are generated from  $O_2^-$  and  $H_2O_2$  produced in the respiratory burst.  $O_2^-$  and  $H_2O_2$  are converted by a complex series of secondary reactions to a) oxidizing radicals and b) oxidized halogens. Examples of each of these are:

a) the OH• (hydroxyl radical) produced primarily by the so called "modified Harber-Weiss" reaction

 $O_2^- + Me^{n+1}$   $Me^n + O_2$ 
 $Me^n + H_2O_2$   $Me^{n+1} + OH^{\bullet} + OH$ 
 $O_2^- + H_2O_2$   $O_2 + OH^{\bullet} + OH^{-1}$ 

where  $Me^{n+1}$  represents the oxidized form of a trace metal. b) oxidation of ubiquitous Cl<sup>-</sup> ions to hypochlorous acid by H<sub>2</sub>O<sub>2</sub> generated during the respiratory burst:

Cl<sup>-</sup> + H<sub>2</sub>O<sub>2</sub> myeloperoxidase HOCL + H<sub>2</sub>O

response, foreign antigens are internalized by macrophages and cleaved into peptide fragments which are returned to the cell surface as peptides bound to class II MHC molecules. Appropriate T lymphocytes expressing clonally derived receptors are then activated through recognition of the macrophage surface complex between the foreign peptide and the MHC molecule (178).

Class II MHC molecules are heterodimers of two highly polymorphic membrane glycoproteins of 33 to 35 kDa (the *a*-chain) and 27 to 29 kDa (the *B*-chain). Each consists of two extracellular domains, a connecting peptide, a transmembrane domain, and a cytoplasmic tail. In the mouse there are two class II MHC, or Ia, molecules, A and E, which are encoded by a number of genes (Aa, AB, Ea, and EB respectively) residing in the MHC complex (179).

Macrophage Ia antigen expression is not constitutive, but is subject to regulation by cytokines. Research has shown that Ia expression is induced primarily by interferon (IFN)-gamma (180), however tumor necrosis factor-*a* (181), IL-4 (182), and granulocytemacrophage colony-stimulating factor (183) also exhibit this capability. Furthermore, it appears that MHC class II expression in the macrophage is regulated mainly by transcription, either positively or negatively. Control of transcription has been directly linked to rapid transductional sequences initiated by IFN-gamma as well as to the binding of proteins to specific regions upstream of the promoter in macrophages (184). The enhanced expression of murine macrophage Ia antigens may be used as a marker for activation (179, 184).

Some of the studies described in this manuscript involve the quantification of spleen macrophages expressing Ia as a means of monitoring their activation state. To achieve this, a double antibody label system specific for Mac.1 and Ia surface antigens was used. Mac.1 is a differentiation antigen known to be involved in leukocyte adhesion reactions. It is a member of a family of structurally and functionally related leukocyte surface glycoproteins which also includes lymphocyte function-associated 1 (LFA-1) and p150,95 antigens. Each of these antigens are characterized by *a*1 and B1 noncovalently associated subunits and are highly conserved in both mouse and man (185). Mac.1 is present on myeloid lineage cells but absent from lymphoid cells. In mice, it is expressed principally on macrophages, regardless of their activation state, but can also be detected on granulocytes and natural killer cells at a significantly lower surface density (186).

By labelling spleen cell isolates with Ia and Mac.1 specific antibodies prior to analysis by flow cytometry we were able to distinguish between macrophages expressing Ia antigen on their surface (Mac.1+Ia<sup>+</sup>), B cells (Mac.1-Ia<sup>+</sup>) and surface Ia negative macrophages (Mac.1+Ia<sup>-</sup>). As indicated previously, granulocytes and natural killer cells express relatively small quantities of Mac.1 therefore do not stain intensely enough for detection as Mac.1<sup>+</sup> in our system.

#### CONCLUSION

To summarize, AA amyloidosis is common to a host of seemingly unrelated disorders. Clinically it can be a complication of long-standing inflammation, familial disorders, and various forms of malignancy and can be induced experimentally in animals through chronic inflammation. A common protein being deposited in such different clinical settings suggests there is a common pathogenic mechanism occurring in each case (81). The AA protein deposited in AA amyloidosis is thought to be derived from the N-terminal end of the naturally occuring circulating serum apolipoprotein SAA (40). ApoSAA is an acute phase serum protein of unknown function whose serum concentration increases both rapidly and substantially during any inflammatory process (22, 56). Activation of a variety of cells including polymorphonucleocytes and reticuloendothelial cells is an integral part of any inflammatory reaction. Upon activation, these cells release a range of inflammatory cytokines including IL-1, IL-6, and tumor necrosis factor, which are responsible for upreg lating SAA gene transcription (56, 57, 58) such that apoSAA forms a substantial part of HDL during inflammation (64). The relative rarity of amyloidosis as a complication of inflammatory disease indicates however, that factors other than the availability of sufficient amounts of precursor protein are essential if amyloid deposition is to occur.

Little is known of the processing events which lead from the circulating apoprotein precursor to the AA fragment characteristic of the amyloid deposit. The *in situ* association between reticuloendothelial cells and amyloid fibrils is well known (164). Furthermore, both mononuclear leukocytes from healthy individuals and normal murine Kuppfer cells have been found to completely degrade SAA *in vitro* most likely through the action of cell surface enzymes. Cells from amyloidotic patients and mice, on the other hand, have been shown to produce an AA intermediate (166-168). These observations are among those supporting the popular hypothesis that tissue amyloid deposition is the result of innappropriate or incomplete SAA degradation by cells of the monocyte/macrophage lineage, resulting in the accumulation of AA.

The theory that AEF may be the effector behind altered macrophage function ultimately resulting in AA deposition is supported by data comparing the kinetics of AA intermediate accumulation and the appearance of AEF activity (168, 3) in which detectable AEF activity necessarily precedes AA fibril formation. Furthermore, AEF dramatically shortens the time necessary for the experimental induction of amyloidosis and can be AEF activity necessarily precedes AA fibril formation. Furthermore, AEF dramatically shortens the time necessary for the experimental induction of amyloidosis and can be detected in tissues 24-48 hours before the appearance of amyloid (3). The kinetics of amyloid deposition using standard induction protocols, or following passive transfer of AEF, are identical with the exception of the lag phase necessary for the appearance of AEF (157).

It is clear that both augmented levels of precursor protein synthesis and generation of AEF actvity, in the presence of macrophages, are necessary for fibrillogenic degradation of SAA into AA fibrils. It has yet to be determined why the AA peptides fold into the cross-ß-pleated sheet conformation (14) which is not normally present in non-pathogenic fibril polymers. The common presence of SAP, glycosaminoglycans, and apolipoprotein E in the variety amyloid deposits tested to date may well influence fibril formation. Figure 1.3 gives a general outline of the elements thought to be involved in the the pathogenesis amyloidosis.

Both reticuloendothelial cells and AEF are obviously integral to the development of amyloidosis, yet their exact roles and interactions have yet to be characterized. The lack of specific biochemical characterization of AEF has limited any research done in this area. The studies presented in this thesis have therefore been designed with two major objectives in mind: the purification of AEF to homogeneity and the characterization of the relationship between AEF, macrophages, and amyloid deposition using a solublilized AEF preparation produced in our lab. Studies of this nature are essential to our understanding of the pathogenesis of amyloidosis.



Fig. 1.3. The Pathogenesis of Amyloidosis

Ĵ

#### REFERENCES

(1). Werdelin, O., Ranlov, P. 1966. Amyloidosis of mice produced by transplantation of spleen cells from casein-treated mice. *Acta Pathol .Microbiol. Scand.*; 68: 1.

(2). Shirahama, E.S., Miura, K., Ju, S-T., *et al.* 1990. Amyloid enhancing factor-loaded macrophages in amyloid fibril formation. *Lab Invest.*; 62: 61.

(3) Axelrad, M., Kisilevsky, R. 1979. Biological characterization of amyloid enhancing factor. In: <u>Amyloid and Amyloidosis</u>. Proc. 3rd. Symposium on Amyloidosis (Glenner, G.G. Costa, P.P. deFreitas, F.F., Eds.) p. 527. Excerpta Med.: Amsterdam.

(4). Axelrad, M., Kisilevsky, R., Willmer, J., *et al.* 1982. Further characterization of amyloid enhancing factor *Lab. Invest*; 47: 139.

(5). Ali-Khan, Z, Quirion, R., Robitaille, Y., *et al.* 1988. Evidence for increased amyloid enhancing factor activity in Alzheimer brain extract. *Acta. Neuropathol.* 77: 82.

(6). Virchow, R. 1855. Zur Cellulose-Frage. Arch. Pathol. Anat. Physiol. 8: 140.

(7). Friedreich, N., Kekulé, A. 1859. Zur amyloidfrage. Arch. Pathol. Anat. Physiol. Klin. Med. 16: 50.

(8) Husby, G., Arakı, S., Benditt, E.P., *et al.* 1990. The 1990 guidelines for nomenclature and classification of amyloid and amyloidosis. In <u>Amyloid and Amyloidosis</u> 1990. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) p. 7. Kluwer:Dordrecht.

(9). Rokitansky, C. 1842. On the abnormalities of the liver. Handbuch der pathologischen anatomie. 3:11. in Bramüller & Seidel (Eds.) Wein.

(10) Bennhold, H. 1922. Eine spezifische amyloïdfarbung mit kongorot. *Muench. Med.*. *Wochenschr.* 69:1537.

(11). Glenner, G.G., Page, D.L. 1976. Amyloid, amyloidosis, and amyloidogenesis. Int. Rev. Exp. Pathol. 15:1.

(12). Divry, P., Florkin, M. 1927. C. R. Soc. Biol. 97:1808-10.

(13). Glennar, G.G. 1980. Amyloid deposits and amyloidosis- the ß-fibrillosis. N. Engl. J. Med. 302: 1283, 1333.

(14). Termine, J.D., Eanes, E.D. Ein, D., *et al.* 1972. Infrared spectroscopy of human amyloid fibrils and immunoglobulin proteins. *Biopolymers* 11: 1103.

(15). Cooper, J.H. 1974. Selected amyloid staining as a function of amyloid composition and structure: histochemical analysis of the alkaline Congo red, standardized toluidine blue, and iodine methods. *Lab. Invest.* 31:232.

(16) Pras, M., Schubert, M., Zucker-Franklin, D., *et. al.* 1968. The characterization of soluble amyloid prepared in water. *J. Clin Invest.* 47:924.
(17) Glenner, G.G., Harada, M., Isersky, C. 1972. The purification of amyloid fibril proteins. *Prep. Biochem.* 2:39.

(18) Kisilevsky, R. 1987. From arthritis to Alzheimer's disease: current concepts on the pathogenesis of amyloidosis. *Can J. Physiol Pharmacol* 65:1805-

(19) Sletten, K., Husby, G., Natvig, J.B. 1976. The complete amino acid sequence of an amyloid fibril protein AA of unusual size (64 residues). *Biochem Biophys Res. Commun.* 69:19.

(20) Ein, D., Kimura, S., Terry, W D., *et al.* 1972. Amon acid sequence of an amyloid fibril protein of unknown origin. *J. Biol. Chem.* 247: 5653

(21) Enksen, N., Ericsson, L.H., Pearsall, N., *et al.* 1976. Mouse amyloid protein-AAhomology with non-immunoglobulin protein of human and monkey amyloid substance. *Proc. Natl. Acad. Sci. U.S.A.* 73:964.

(22) McAdam, K.P.W.J., Sipe, J.D. 1976. Murine model for human secondary amyloidosis: genetic variability of the acute phase serum protein SAA response to endotoxin and casein. *J. Exp. Med.* 144:1121.

(23) Shohat, M., Bu, X., Shohat, T., *et al.* 1992. The gene for Familial Mediterranean
Fever in both Armenians and non-Ashkenazi jews is linke dto the alpha-globin complex on
16p: evidence for locus homogeneity. *Am J Hum Genet* 51:1349

(24) Levin, M., Franklin, E.C., Frangine, B., *et al.* 1972. The amino acid sequence of a major nonimmunoglobulin component of some amyloid fibrils. *J. Clin. Invest.* 51:2773
(25) Nomure, S., Kumagai, N., Kanoh, T., *et al.* 1986. Pulmonary amyloidosis

associated with systemic lupus erythematosus. Arthr. Rheum. 29 680

(26) Janigan, D.T., Druet, R.L. 1966. Experimental amyloidosis. Role of antigenicity and rapid induction. *Amer. J. Path.* 48:1013.

(27) Kisilevsky, R., Axelrad, M., Corbett, W., *et al.* 1977 The role of inflammatory cells in the pathogenesis of amyloidosis *Lab. Invest.* 37:544

(28) Kuczynski, M.H., 1922. Edwin GoldmannsUntersuchungen über cellare Vorgange im Gelfolge des Verdauungsprozesses auf Grund nachgelassener Praparate dargestellt und durche neue Versuche ergantz. Virchows Arch path Anat 239:185

(29) Kuczynski, M.H., 1923. Weitere Beitrage zur Lehre vom Amyloid (über die Rückbildung des Amyloids). *Klin. Wschr.* 2:2193.

(30) Skinner, M., Shirahama, T., Benson, M.D., *et al.* 1977 Murine amyloid protein AA in casein-induced experimental amyloidosis. *Lab Invest* 36:420.

(31) Cohen, A.S., Cathcart, E.S. 1972. Casein-induced experimental amyloidosis. *Meth* Achievm. exp. Path. 6:207.

(32) Lavie, G., Flechner, E., Gallo, G., *et al.* 1979. Variations in the susceptibility of different strains of mice to the development of amyloidosis and the production of SAA. *Ann. NY. Acad Sci* 387:518.

(33) Wolgethan, J.R., Cathcart, E.S., 1979. Amyloid in A/J mice is determined by a single dominant gene. *Nature* 278:453.

(34) Janigan, D.T., Druet, R.L. 1966. Experimental amyloidosis. Role of antigenicity and rapid induction. *Amer. J. Path.* 48:1013.

(35) Kisilevsky, R. 1987. From arthritis to Alzheimer's disease: current concepts in the pathogenesis of amyloidosis. *Can. J. Physiol. Pharmacol.* 65:1805.

(36). Letterer, E. 1968. Experimental amyloidosis with particular reference to parabosis and to qualitative and quantitative changes of the amyloid-inducing substances in mice and guinea pigs (amyloidosis): In Excerpta Medica Foundation, Amsterdam.

(37) Barth, W.F., Gordon, J.K., Willerson, J.T. 1968. Amyloidosis induced in mice by *Escherichia coli* endotoxin. *Science* 162:694.

(38) Barth, W.F., Willerson, J.T., Asofsky, R., *et al.* 1969. Experimental murine amyloid. III. Amyloidosis induced with endotoxins. *Arth. Rheum.* 12:615.

(39) Levin, M., Pras, M., Franklin, E.C. 1973. Immunologic studies of the major nonimmunoglobulin protein of amyloid. *J. Exp. Med.* 138.

(40) Benditt, E.P., Eriksen, M., Hermodsen, M.A., *et al.* 1971. The major proteins of human and monkey amyloid substance: common properties including unusual N-terminal amino acid sequences. *FEBS Lett.* 19:169.

(41) Husebekk, A., Skogen, B., Husby, G., *et al.* 1985. Transformation of amyloid precursor SAA to protein AA and incorporation in amyloid fibrils *in vivo*. *Scand*. *J. Immunol*. 21:283.

(42) Coetzee, G.A., Strachan, A.F., van der Westhuyzen, *et al.* 1986. Serum amyloid Acontaining human high density lipoprotein 3. Density, size, and apolipopprotein composition. *J. Biol. Chem.* 261:9644.

(43) Sack, G.H., Talbot, C.C., Seuanez, H., *et al.* 1989. Molecular analysis of the human serum amyloid A (SAA) gene family. *Scand. J. Immunol.* 29:113.

(44) Yamamoto, K-J., Goto, N., Kosaka, J., et. al. 1987. Structural diversity of the murine SAA genes: evolutionary implications. J. Immunol. 139:1683.

(45) Lowell, C.A., Potter, D.A., Stearman, R.S., *et al.* 1986. Structure of the murine SAA gene family: gene conversion. *J. Biol. Chem.* 261:8442.

(46) Lowell, C.A., Stearman, R.S., Marrow, J.F. 1986. Transcriptional regulation of SAA gene expression. *J. Biol. Chem.* 261:8453.

(47) Ramadori, G., Sipe, J.D., Colten, H.R. 1985. Expression and regulation of the murine serum amyloid A (SAA) gene in extrahepatic sites. *J Immunol.* 135:3645.

(48) Benditt, E.P., Meek, R.L. Expression of the third member of the serum amyloid A gene family in mouse adipocytes. *J. Exp. Med.* 169:1841.

(49) Woo, P., Sipe, J.D., Dinarello, C.A., *et al.*, 1987. Structure of a human serum amyloid A gene and modulation of expression in transfected L cells. *J Biol Chem.* 262:15790.

(50) Betts, J.C., Edbrooke, M.R., Thakker, R.V., *et al.* 1991. The human acute-phase serum amyloid A gene family: structure, evolution and expression in hepatoma cells. *Scand. J. Immunol.* 34:471.

(51) Sack, G.H., Talbot, C.C. 1992. The human serum amyloid A locus SAA4 is a pseudogene. *Biochem. Biophys. Res. Comm.* 183:362.

(52) Kluve-Beckerman, B., Dwulet, F.E., Benson, M.D. 1988. Human serum amyloid A. Three hepatic mRNAs and the corresponding proteins in one person. *J. Clin. Invest.* 82:1670.

(53) Strachan, A.F., de Beer, F.C., van der Westhuyzen, D.R., et al. 1988. Identification of three isoform patterns of human SAA protein. *Biochem. J.* 250:203.

(54) Steinmetz, A., Vitt, H., Motzny, S., et al. 1990. Genetic isofocusing variant of human serum amyloid A. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., et al., Eds.) p. 20. Kluwer:Dordrecht.

(55) Beach, C.M., de Beer, M.C., Sipe, J.D., *et al.* 1992 Human serum amyloid A protein. Complete amino acid sequence of a new variant. *Biochem. J.* 282:615.

(56) Pepys, M.B., Baltz, M.L. 1983. Acute phase proteins with special reference to C-reactive protein and related proteins (pentraxins) and serum amyloid A protein. Adv. Immunol. 34:141.

(57) Selinger, M.J., McAdam, K.P., Kaplan, M.M., et al. 1980. Monokine-induced synthesis of serum amyloid A protein by hepatocytes. *Nature* 285: 498.

(58) Ganapathi, M.K., Schultz, D., Mackiewicz, A., et al. 1988. Heterogenous nature of the acute phase response: differential regulation of human serum amyloid A, C-reactive protein and other acute phase proteins by cytokines in Hep 3B cells. J. Immunol. 141:564.
(59) Ganapathi, M.K., Rzewnicki, D., Samols, D., et al. 1991. Effect of combinations of cytokines and hormones on synthesis of serum amyloid A and C-reactive protein in HEP 3B cells. J. Immunol. 147:1261.

(60) Sipe, J.D., Rokita, H., Bartle, L.M., *et al.* 1991. The IL-1 receptor antagonist simultaneously inhibits SAA and stimulates fibrinogen synthesis *in vivo* and *in vitro*: a proposed mechanism of action. *Cytokine* 3:497.

(61) Woo, P., Betts, J., Edbrooke, M. 1990. The human serum amyloid A genes and their regulation by inflammatory cytokines. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) p. 13. Kluwer:Dordrecht.

(62) Lowell, C.A., Stearman, R.S., Marrow, J.F. 1986. Transcriptional regulation of SAA gene expression. *J. Biol. Chem.* 261:8453.

(63) Sipe, J.D. 1992. Amyloidosis. Ann. Rev. Biochem. 61:947.

(64) Hoffman, J.S., Benditt, E.P. 1982. Secretion of serum amyloid protein and assembly of serum amyloid protein-rich high density lipoprotein in primary mouse hepatocyte culture. *J. Biol. Chem.* 257: 10518.

(65) Kisilevsky, R. 1991. Serum amyloid A (SAA), a protein without a function: some suggestions with reference to cholesterol metabolism. *Med. Hyp.* 35:337.

(66) Zimlichman, S., Danon, A., Nathan, I., *et al.* 1990. Serum amyloid A, an acute phase protein, inhibits platelet activation. *J. Clin. Med.* 116:180.

(67) Benson, M.D., Aldo-Benson, M.J. 1979. Effect of purified protein SAA on immune response *in vitro*: Mechanisms of suppression. *J. Immun*<sub>2</sub>*l*. 122: 2077.

(68) Shainkin-Kestenbaum, R., Berlyne, G., Zimlichman, S., *et al.* 1991. Acute phase protein, serum amyloid A, inhibits IL-1- and TNF-induced fever and hypothalamic PGE<sub>2</sub> in mice. *Scand. J. Immunol.* 34: 179.

(69) Mitchell, T.I., Coon, C.I., Brinckerhoff, C.E. 1991. Serum amyloid A (SAA3) produced by rabbit synovial fibroblasts treated with phorbol esters or interleukin 1 induces synthesis of collagenase and is neutralized with specific antiserum. *J. Clin. Invest.* 87:1177.

(70) Sack, G., Conover, C.C. 1989. The human serum amyloid A (SAA)-encoding gene GSAA1: nucleotide sequence and possible autocrine-collagenase-inducer function. *Gene*. 84:509.

(71) Hoffman, J.S., Ericsson, L.H., Eriksen, N., *et al.* 1984. Murine tissue amyloid protein AA. NH<sub>2</sub>-terminal sequence identity with only one of the two serum amyloid protein (apo-SAA) gene products. *J. Exp. Med.* 159:641.

(72) Westermark, G.T., Engström, U., Westermark, P. 1992. The N-terminal segment of protein AA determines its fibrillogenic property. *Biochem. Biophys. Res. Comm.* 182:27.
(73) Meek, R.L., Hoffman, J.S., Benditt, E.P. 1986. Amyloidogenesis: one serum amyloid A isotype is selectively removed from the circulation. *J. Exp. Med.* 163:499.
(74) Hébert, L., Gervais, F. 1990. Apo-SAA1/apo-SAA2 isotype ratios during casein-

and amyloid-enhancing-factor-induced secondary amyloidosis in A/J and C57Bl/6J mice. *Scand. J. Immunol.* 31:167.

(75) Husebekk, A., Skogen, B., Husby, G. 1988. High-density lipoprotein has different binding capacity for different apoproteins. The amyloidogenic apoproteins are easier to displace from high density lipoprotein. *Scand. J. Immunol.* 28:653.

(76) Tobias, P.S., Ulevitch, R.J. 1983. Control of lipopolysaccharide-high density lipoprotein binding by the acute phase protein(s). J. Immunol. 131:1913.

(77) Ulevitch, R.J., Johnston, A R., Weinstein, D.B. 1981. New function for high density lipoprotein. Isolation and characterization of a bacterial lipopolysaccharide-high density lipoprotein complex formed in rabbit plasma. *J. Cli. Invest.* 67:827.

(78) Liepniecks, J.J., Leagre, C., Kluve-Beckerman, B., *et al.* 1990. Predominance of one SAA isotype (SAA1) in human reactive amyloid. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) pp. 511. Kluwer:Dordrecht.

(79) Westermark, G.T., Sletten, K., Westermark, P. 1989. Massive vascular AAamyloidosis: a histiologically and biochemically distinctive subtype of reactive systemic amyloidosis. *Scand. J. Immunol.* 30:605.

(80) Telium, G. 1964. Pathogenesis of amyloidosis. The two phase theory of local secretion. Acta. Pathol. Microbiol. Scand. 61:21.

(81) Kisilevsky, R. 1983. Amyloidosis: a familiar problem in the light of current pathogenic developments. *Lab. Invest.* 48:60.

(82) Cohen, A.S., 1965. The constitution and genesis of amyloid. In <u>International Review</u> of <u>Experimental Pathology</u>. (Richter, G.W., Epstein, M.A. Eds.) New York Academic Press. vol. 4. pp. 159.

(83) Cathcart, E.S., Shirahama, T., Cohen, A.S. 1967. Isolation and identification of a plasma component of amyloid. *Biochem. Biophys. Acta* 147:392.

(84) Westermark, P., Shirahama, T., Skinner, M., et al. 1982. Immunohistochemical evidence for the lack of amyloid P-component in some intracerebral amyloids. Lab. Invest. 46:457.

(85) Kalaria, R.N., Galloway, P.G., Perry, G. 1991. Widespread serum amyloid P immunoreactivity in cortical amyloid deposits and the neurofibrillary pathology of Alzheimer's disease and other degenerative disorders. *Neuropathol. Appl. Neurobiol.* 17:189.

(86) Baltz, M.L., Caspi, D., Rowe, I.F., et al. 1986. Pathogenic mechanisms and precursor product relationships in murine amyloidosis. In: <u>Amyloidosis</u> (Glenner, G.G., Osserman, E.F., Benditt, E.P., et al. Eds.) New York, Plenum Press. pp. 101.
(87) Pepys, M.B., Dash, A.C., Fletcher, T.C., et al. 1978. Analogues in other mammals and in fish of human plasma proteins: C-reactive protein and amyloid-P component. *Nature* 273:168.

(88) Pinteric, L., Painter, R.H. 1979. Electron microscopy of serum amyloid protein in the presence of calcium; alternative forms of assembly of pentagonal molecules in twodimensional lattices. *Can. J. Biochem.* 57:727.

(89) Painter, R.H., De Escallon, I., Massey, A., et al. 1982. The structure and binding characteristics of serum amyloid P protein. Ann. N. Y. Acad. Sci. 389:199.

(90) Pepys, M.B., Baltz, M.L., Gomer, K., *et al.* 1979. Serum amyloid P component is an acute phase reactant in the mouse. *Nature* 278:259.

(91) Baltz, M.L., deBeer, F.C., Feinstein, A., et al. 1982. Phylogenic aspects of C-reactive protein and related proteins. Ann. N. Y. Acad. Sci. 389:49.

(92) Skinner, M., Sipe, J.D., Yood, R.A., et al. 1982. Characterization of P-component
[AP] isolated from amyloidotic tissue: half-life studies of human and murine AP. Ann. N.
Y. Acad. Sci. 389:190.

(93) Skinner, M., Vaitukaitis, J.L., Cohen, A.S., *et al.* 1979. Serum amyloid P component levels in amyloidosis, contective disease, infection and malignancy as compared to normal serum. *J. Lab. Clin. Med.* 94:633.

(94) Mortensen, R.F., Beisel, K., Zeleznik, N.J., *et al.* 1983. Acute phase reactants of mice. II. Strain dependence of serum amyloid P-component (SAP) levels and response to inflammation. *J. Immunol.* 130:885.

(95) Ramadori, G., Sipe, J.D., Dinarello, C.A., *et al.* 1985. Pretranslational modulation of acute phase hepatic protein synthesis by murine recombinant interleukin-1 (IL-1) and purified human IL-1. *J. Exp. Med.* 162:930.

(96) Gauldie, J., Richards, C., Harnish, D., *et al.* 1987. Arterferon B2/B-cell stimulating factor type 2 shares identity with monocyte-derived hepatocyte stimulating factor and regulates the major acute phase protein response in liver cells. *Proc. Natl. Acad. Sci. USA* 84:7251.

(97) Perlmutter, D.H., Dinarello, C.A., Punsal, P.I., *et al.* 1986. Cachetin/tumor necrosis factor regulates hepatic acute phase gene expression. *J. Clin. Invest.* 78:1349.

(98) Le, P.T., Mortensen, R.F., 1984. Mouse hepatocyte synthesis and induction of the acute phase reactant: serum amyloid P-component. *In Vitro* 20:505.

(99) Lin, B-F., Ku, N-O., Zahedi, K., *et al.* 1989. Induction of hepatocyte synthesis of the mouse acute phase protein serum amyloid P-component (SAP) by IL-1 and IL-6. *Ann. NY Acad. Sci.* 557:534.

(100) Tatsuta, E., S'pe, J.D., Shirahama, T., *et al.* 1983. Different regulatory mechanisms for serum amyloid A and serum amyloid P synthesis by cultured mouse hepatocytes. J. *Biol. Chem.* 258:5414.

(101) Le, P.T., Muller, M.T., Mortensen, R.F. 1982. Acute phase reactants of mice. I. Isolation of serum amyloid P-component (SAP) and its induction by a monokine. J. *Immunol.* 129:665.

(102) Breathnach, S.M., Melrose, S.M., Bhogal, B., et al. 1981. Amyloid P component is located on elastic microfibrils in normal human tissue. Nature 293:652.

(103) Dyck, R.F., Lockwood, C.M., Kershaw, M., et al. 1980. Amyloid P component is a constituent of normal human basement membrane. J. Exp. Med. 152:1162.

(104) Baltz, M.L., Dyck, R.F., Pepys, M.B. 1980. Amyloid P-component in mice injected with casein: identification in amyloid deposits and in the cytoplasm of hepatocytes. *Immunol.* 41:59.

(105) deBeer, F.C., Baltz, M.L., Holford, S., et al. 1981. Fibronectin and C4-binding protein are selectively bound by aggregated amyloid P component. J. Exp. Med. 154:1134.
(106) Hamazaki, H. 1987. Ca<sup>2+</sup>-mediated association of human serum amyloid P component with heparan sulphate and dematan sulphate. J. Biol. Chem. 262:1456.
(107) Siripont, J., Tebo, J.M., Mortensen. 1988. Receptor mediated binding of the acute-

phase reactant mouse serum amyloid-P component (SAP) to macrophages. *Cell. Immunol.* 117:239.

(108) Hind, C.R.K., Collins, P.M., Baltz, M.L., *et al.* 1985. Human serum amyloid P component, a circulating lectin with specificity for the cyclic 4,6-pyruvate acetal of galactose. Interactions with various bacteria. *Biochem. J.* 225:107.

(109) Pepys, M.B., Butler, P.J.G. 1987. Serum amyloid P component is the major calcium-dependent specific DNA binding protein of the serum. *Biochem. Biophys. Res. Commun.* 148:308-313.

(110) Butler, P.J.G., Tennent, G.A., Pepys, M.B. 1990. Pentraxin-chromatin interactions. Serum amyloid P component specifically displaces H1-type histones and solubilizes native long chromatin. J. Exp. Med. 172:13.

(111) Schwaibe, R.A., Dahlbäck, B., Nelsestuen, G. 1990. Independent association of serum amyloid P component, protein S, and complement C4b with complement C4bbinding protein and subsequent association of the complex with membranes. J. Biol. Chem. 265:21757.

(112) Schwalbe, R.A., Dahlbäck, B., Nelsestuen, G. 1991. Heparin influence on the complex of serum amyloid P component and complement C4b-binding protein. *J. Biol. Chem.* 266:12896.

(113) Baltz, M.L., Gomer, K., Davies, A.J.S., *et al.* 1980. Differences in the acute phase responses of serum amyloid P-component (SAP) and C3 to injections of casein or bovine

serum albumin in amyloid-sensitive and -resistant mouse strains. *Clin. Exp. Immunol.* 39:355.

(114) Li, J.J., McAdam, K.P.W.J. 1984. Human amyloid P component: an elastase inhibitor. *Scand. J. Immunol.* 20:219.

(115) Vachino, G., Heck, L.W., Gelfand, J.A., *et al.* 1988. Inhibition of human neutrophil and *Pseudomonas* elastases by the amyloid-P component: a constituent of elastic fibers and amyloid deposits. *J. Leuk. Biol.* 44:529.

(116) Ruoslahti, E. 1988. Structure and biology of proteoglycans. Ann. Rev. Cell Biol. 4:229.

(117) Snow, A.D., Willmer, J., Kisilevsky, R. 1987. Sulphated glycosaminoglycans: a common constituent of all amyloids. *Lab. Invest.* 56:120.

(118) Snow, A.D., Kisilevsky, R. 1985. Temporal relationship between glycosaminoglycan accumulation and amyloid deposition during experimental amyloidosis: A histochemical study. *Lab. Invest.* 53:37.

(119) Snow, A.D., Kisilevsky, R., Stephens, C., *et al.* 1987. Characterization of tissue and plasma glycosaminoglycans during experimental AA amyloids and acute inflammation. Qualitative and quantitative analysis. *Lab. Invest.* 56:665.

(120) Snow, A.D., Kisilevsky, R., Wight, T.N. 1988. Immunolocalization of heparan sulphate proteoglycans to AA amyloid deposition sites in spleen during experimental amyloidosis. In: <u>Amyloid and Amyloidosis</u>. (Isobe, T., Araki, S., Uchino, F., *et al.* Eds.) pp.87. New York, Plenum Press.

(121) Snow, A.D., Kisilevsky, R. 1988. A close ultrastructural relationship between sulphated proteoglycans and AA amyloid fibrils. *Lab. Invest.* 57:687.

(122) Magnus, J.H., Kolset, S.O., Husby, G. 1991. High molecular glycosaminoglycans in AA type amyloid fibril extracts from human liver. *Ann. Rheum. Dis.* 50:562.

(123) Snow, A.D., Mar, H., Nochlin, D., Kimata, K., *et al.* 1988. The presence of heparan sulphate proteoglycans in neuritic plaques and congophilic angiopathy in Alzheimer's disease. *Am. J. Pathol.* 133:456.

(124) Snow, A.D., Wight, T.N., Nochlin, D., *et al.* 1990. Immunolocalization of heparan sulphate proteoglycans to the PrP amyloid plaques of Gerstmann-Straussler syndrome, Creutzfeld-Jakob disease and scrapie. *Lab. Invest.* 63:601.

(125) Young, I.D., Ailles, L., Aubin, S., et al. 1991. The basement form of heparan sulfate proteoglycan is part of human IAPP amyloid deposits in the islets of Langerhans. In Amyloid and Amyloidosis 1990. (Natvig, J. B., Forre, O., Husby, G., et al., Eds.) pp. 334. Kluwer:Dordrecht.

(126) Snow, A.D. 1990. Peripheral distribution of a dermatan sulphate proteoglycan (decorin) in amyloid plaques and their presence in neurofibrillary tangles of Alzheimer's disease. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) pp. 757. Kluwer:Dordrecht.

(127) McCubbin, W.D., Kay, C.M., Narindrasorasak, S., et al. 1988. Circular dichroism and fluoresence studies on two murine serum amyloid A proteins. *Biochem J.* 256:775.
(128) Kisilevsky, R. 1989. Glycosaminoglycans and amyloid proteins. In:

Cerebrovascular Diseases: Sixteenth Princeton Conference. (Ginsberg, M.D., Dietrich, W.D. Eds.) pp.223. Raven Press, New York.

(129) Kisilevsky, R. 1989. Theme and variations on a string of amyloid. *Neurobiol. Aging* 10:499.

(130) Wisniewski, T., Frangione, B. 1992. Apolipoprotein E: a pathological chaperone protein in patients with cerebral and systemic amyloid. *Neuro*. *Lett.* 135:235.

(131) Namba, Y., Tomonaga, M., Kawasaki, H. et al. 1991. Apolipoprotein E

immunoreactivity in cerebral deposits and neurofibrillary tangles in Alzheimer's disease and kuru plaque amyloid in Creutzfeld-Jacob disease. *Brain Res.* 541:163.

(132) Mahley, R.W. 1988. Apolipoprotein E: cholesterol transport protein with expanding role in cell biology. *Science* 240:622.

(133) Boyles, J. K., Zoellner, C.D., Anderson, L.J. *et al.* 1989. A role for Apolipoprotein E, Apoprotein A, Apoprotein A-1, low density lipoprotein receptors in cholesterol transport during regeneration and remyelination of the rat sciatic nerve. *J. Clin Invest.* 83:1015.
(134) Cuthbert, J.A., Lipsky, P.E. 1984. Modulation of human lymphocyte response by low density lipoprotein (LDL): enhancement but not immunosupression is mediated by LDL receptors. *Proc. Natl. Acad. Sci. U.S.A.* 81:4539.

(135) Carlsson, J., Armstrong, V.W., Reiber, H. et al. 1991. Clinical relevance of the quantification of apolipoprotein E in cerebrospinal fluid. *Clin Chem. Acta.* 196:167.
(136) Mε'hley, R.W., Weisgraber, K.H., Innerarity, T.L. 1979. Interaction of plasma lipoproteins containing apolipoproteins B and E with heparin and cell surface receptors. *Biochem. Biophys. Acta* 575:81.

(137) Benditt, E.P. 1989. What role(s) may extracellular matrix, particularly heparan sulphate, play in amyloid of Alzheimer's disease. *Neurobiol. Aging* 10:506.
(138) Dahlbäck, K., Löfberg, H., Dahlbäck, B. 1987. Immunohistochemical demonstration of vitronectin in association with elastin and amyloid deposits in human kidney. *Histochem.* 87:511.

35

(139) Abraham, C.R., Shirahama, T., Potter, H. 1990. Alpha-1-antichymotrypsin is associated solely with amyloid deposits containing the beta-protein- amyloid and cell localization of alpha-1-antichymotrypsin. *Neurobiol. Aging* 11:123.

(140) Glenner, G.G. 1980. Amyloid deposits and amyloidosis- the beta-fibrilloses. N. Eng. J. Med. 302:1283-92, 1333.

(141) Rothman, J.E. 1989. Polypeptide chain binding proteins: catalysts of protein folding and related processes in cells. *Cell* 59:591.

(142) Timpl, R., Dziadek, M. 1986. Structure, development, and molecular pathology of basement membranes. *Int. Rev. Exp. Pathol.* 29:1.

(143) Lyon, A.W., Narindrasorasak, S., Young, I.D., *et al.* 1991. Co-deposition of basement membrane components during the induction of murine splenic AA amyloid. *Lab. Invest.* 64:785.

(144) Schultz, R.T., Pintha, J.V., McDonald, T., et al. 1985. Ultrastructural studies of vascular lesions in experimental amyloidosis in mice. Am. J. Pathol. 119:138.

(145) Perlmutter, L.S., Chui, H.C., Saperia, D., *et al.* 1990. Microangiopathy and the colocalization of heparan sulfate proteoglycan with amyloid in senile plaques of Alzheimer's disease. *Brain Res.* 508:13.

(146) Kisilevsky, R. 1990. Heparan sulfate proteoglycans in amyloidogenesis: an epiphenomenon, a unique factor, or the tip of a more fundamental process? *Lab. Invest.* 63:589.

(147) Werdelin, O., Ranlov, P. 1966. Amyloidosis in mice produced by transplantation of spleen cells from casein-treated mice. Acta Pathol. Microbiol. Scand. 68:1.

(148) Morisette, C., Gervais, F. Induction of amyloid deposition by subfractions of the amyloid enhancing factor extracted from brain of patients with Alzheimer's disease. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) pp. 303. Kluwer:Dordrecht.

(149) Sipe, J.D., De Beer, F.C., Pepys, M., et al. 1990. Report of special sessions on bioassays and standardization of amyloid proteins and precursors. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., et al., Eds.) pp. 883. Kluwer:Dordrecht.

(150) Varga, J., Flinn, M.S., Shirahama, T., et al. 1986. The induction of murine amyloid with human splenic extract. Probable role of amyloid enhancing factor. *Virchows Arch. B* 51:177.

(151) Axelrad, M. A., Kisilevsky, R., Beswetherick, S. 1975. Acceleration of amyloidosis by syngeneic spleen cells from normal donors. *Am. J. Pathol.* 78:277.
(152) Hardt, F., Ranlov, P. 1976. Transfer amyloidosis. *Int. Rev. Exp. Pathol.* 16:273.

(153) Keizman, I., Rimon, A., Sohar, E., *et al.* 1972. Amyloid accelerating factor.
Purification of a substance from human amyloidotic spleen that accelerates the formationof casein-inducing murine amyloid. *Acta Pathol. Microbiol. Scand* (A) 233:172.
(154) Baltz, M.L., Caspi, D., Hind, C.R.K., et al. 1986. Isolation and characterization of amyloid enhancing factor. In <u>Amyloidosis</u>. (Glenner, G.G., Osserman, E.F., Bendutt, E.P., *et al.*, Eds.) pp. 115. Plenum; NY

(155) Hol, P.R., Snel, F.W.J.J., Niewold, Th. A., et al. 1986. Amyloid enhancing factor (AEF) in the pathogenesis of AA-amyloidosis in the hamster. Virchows Arch. B 52:273.
(156) Deal, C.L., Sipe, J.D., Tatsuta, E., et al. 1982. The effect of amyloid enhancing factor (AEF) on the acute phase serum amyloid A (SAA) and serum amyloid P (SAP) response to silver nitrate. Ann. NY Acad. Sci. 389:439.

(157) Kisilevsky, R., Boudreau, L. 1983. The kinetics of amyloid deposition. I. The effects of amyloid-enhancing factor and splenectomy. *Lab. Invest* 48:60.

(158) Gervais, F., Hébert, L., Skamene, E. 1988. Amyloid-enhancing factor: production and response in amyloidosis-susceptible and -resistant mouse strains. J Leuk Biol.
43:311.

(159) Alizadeh-Khiavi, K., Ali-Khan, Z. 1988. Biochemical nature and cellular origin of amyloid enhancing factor (AEF) as determined by anti-AEF antibody. *Br. J. Exp. Pathol.* 69:605.

(160) Alizadeh-Khiavi, K., Normand, J., Chronopoulos, S., *et al.* 1992. Amyloidenhancing factor activity is associated with ubiquitin. *Virchows Archiv. A Pathol. Anat.* 420:139.

(161) Alizadeh-Khiavi, K., Normand, J., Chronopoulos, S., *et al.* 1991. Alzheimer's disease brain-derived ubiquitin has amyloid-enhancing factor activity: behaviour of ubiquitin during accelerated amyloidogenesis. *Acta Neuropathol.* 81.280.

(162) Kisilevsky, R., Benson, M., Axelrad, N.M., *et al* 1979. The effect of a liver protein synthesis inhibitor on plasma SAA levels and a model of accelerated amyloid deposition. *Lab. Invest.* 41:206.

(163) Abankwa, G.V., Ali-Khan, Z. 1988. Induction of murine amyloid enhancing factor and its biological properties in murine alveolar hydatidosis. *Br. J. Exp. Pathol.* 69:123.
(164) Gueft, B., Ghidoni, J.J. 1963. The site of formation and ultrastructure of amyloid. Am, J. Path, 43:837.

(165) Shirahama, T., Cohen, A.S. 1975. Intralysosomal formation of amyloid fibrils. Am. J. Pathol. 81:101.

(166) Lavie, G., Zucker-Franklin, D., Franklin, E.C. 1980. Elastase-type proteases on the surface of human blood monocytes: possible role in amyloid formation. *J. Immunol.* 125:175.

(167) Skogen, B., Thorsteinsson, L., Natvig, J.B. 1980. Degradation of protein SAA to an AA-like fragment by enzymes of the monocytic origin. *Scand. J. Immunol.* 11:533.
(168) Fuks, A., Zucker-Franklin, D. 1985. Impaired Kupffer cell function precedes development of secondary amyloidosis. *J. Exp. Med.* 161:1013.

(169) Van Furth, R. 1988. Phagocytic cells: Development and distribution of mononuclear phagocytes in normal steady state and inflammation. In <u>Inflammation. Basic Principles and</u> <u>Clinical Correlates.</u> (Gallin, J.I., Goldstein, I.M., Snyderman, R. Eds.) pp. 281. Raven Press : NY.

(170) Adams, D.O., Hamilton, T.A. 1988. Phagocytic cells: cytotoxic activities of macrophages. In <u>Inflammation. Basic Principles and Clinical Correlates.</u> (Gallin, J.I., Goldstein, I.M., Snyderman, R. Eds.) pp. 471. Raven Press : NY.

(171) Unanue, E.R., Allen, P.M. 1987. The basis for the immunoregulatory role of macrophages and other accessory cells. *Science* 236:551.

(172) Berche, P., Gaillard, J-L., Sansonetti, P.J. 1987. Intracellular growth of *Listeria monocytogenes* as a prerequisite for *in vivo* induction of T-cell mediated immunity. *J. Immunol.* 138:2266.

(173) Babior, B.M. 1984. Oxidants from phagocytes: agents of defense and destruction. *Blood* 64:959.

(174) Nathan, C.F. 1983. Secretory products of macrophages. J. Clin. Invest. 79:319.

(175) Ezekowitz, R.A.B., Austyn, J., Stahl, P.D., *et al.* 1982. Surface properties of the Bacillus-Calmette-Guerin-activated mouse macrophage. Reduced expression of mannose specific endocytosis, Fc receptors, and antigen F4/80 accompanies induction of Ia. *J. Exp. Med.* 154:60.

(176) Rossi, F., Bellavite, P., Berton, G., *et al.* 1985. Mechanism of production of toxic oygen radicals by granulocytes and macrophages and their function in the inflammatory process. *Path. Res. Pract.* 180:136.

(177) Sasada, M., Johnston, R.B.Jr. 1980. Macrophage microbicidal activity: correlation between phagocytosis-associated oxidative metabolism and the killing of candida by macrophages. J. Exp. Med. 152:85.

(178) Unanue, E.R. 1984. Antigen presenting function of the macrophage. Ann. Rev. Immunol. 2:395.

(179) Lafuse, W.P. 1991. Molecular biology of murine MHC class II genes. Crit. Rev. Immunol. 11:167.

(180) Steeg, P.S., Moore, R.N., Johnson, H.M., *et al.* 1982. Regulation of murine Ia antigen expression by a lymphokine with immune interferon activity. *J. Exp. Med.* 56:178.
(181) Chang, R.J., Lee, S.H. 1986. Effects of interferon-gamma and tumor necrosis factor-alpha on the expression of an Ia antigen on a murine macrophage cell line. *J. Immunol.* 137:2853.

(182) Stuart, P.M., Zlotnik, A., Woodward, J.G. 1988 Induction of class I and class II MHC antigen expression on murine bone-marrow derived macrophages by II-4 (B cell stimulatory factor 1).*J. Immunol* 140:1542.

(183) Falk, L.A., Wahl, L.M., Vogel, S.N. 1988. Analysis of Ia antigen expression in macrophages derived from bone marrow cells cultured in granulocyte-macrophage colony-stimulating factor or macrophage-colony stimulting factor. *J. Immunol* 140:2652.
(184) Adams, D.O., Hamilton, T.O. 1987. Molecular transductional mechanisms by which IFN and other signals regulate macrophage development. *Immunol. Rev* 97:5.
(185) Sanchez-Madrid, F., Nagy, J., Robbins, E., *et al* 1983. A human leukocyte differentiation antigen family with distinct alpha subunits and a common beta subunit: the lymphocyte function-associated antigen (LFA-1), the C3bi complement receptor

(OKM1/Mac.1), and the p150,95 molecule. J. Exp. Med. 158:1785.

(186) Ho, M.K., Springer, T.A. 1982. Mac.1 antigen: quantitative expression in macrophage populations and tissues, and immunofluorescent localization in the spleen. J. *Immunol.* 128:2281.

# CHAPTER 2

SPLENIC MACROPHAGE ACTIVATION AND FUNCTIONS IN AMYLOID ENHANCING FACTOR-INDUCED AA AMYLOIDOSIS Study of Phagocytosis, Killing, Respiratory Burst, and MHC Class II Surface Expression

# ABSTRACT

Secondary, or AA, amyloidosis is a systemic disease characterized by the extracellular tissue deposition of insoluble fibrillar amyloid A protein. Aberrant metabolism of serum amyloid A protein by reticuloendothelial cells is thought to result in the accumulation of fibrils within the tissue. Treatment of mice with amyloid enhancing factor (AEF) in conjunction with an inflammatory stimulus (i.e. AgNO3) induced amyloid deposition within 48-72 hours. The activation state of a macrophage largely defines its enzymatic capabilities. The following studies examine the effect of AEF on spleen macrophage activation using both functional and phenotypic assays. We found that while AEF, in the presence or absence of AgNO<sub>3</sub> has no apparent effect on the ability of spleen and liver macrophages to phagocytose or kill L. monocytogenes, it appears to block enhanced respiratory burst function (as measured by O2<sup>-</sup> production) observed with AgNO3 alone. AEF therefore seems capable of inhibiting certain macrophage activationassociated functions while not affecting others. Our activation phenotype studies, using surface Ia expression, reveal that AEF blocks the increase in number of splenic macrophages expressing Ia seen with AgNO3 alone Treatment with IFN-gamma was found to restore decreased Ia expression in animals given AEF+ AgNO3 but did not prevent amyloid A fibril deposition.

## **INTRODUCTION**

Amyloidosis is a generic term for a diverse group of diseases characterized by the extracellular tissue deposition of insoluble protein fibrils, often resulting in failure of the involved organs. AA amyloidosis is typified by the deposition of amyloid A protein (AA) derived from the carboxy-terminal cleavage of its acute phase serum precursor: serum amyloid A (SAA) (1, 2). AA amyloidosis appears in conjunction with certain malignancies, chronic inflammatory disorders, or prolonged infection (3). Experimental AA amyloidosis is induced in laboratory animals by sustained inflammation or chronic bacterial or parasitic infections (4).

Amyloid enhancing factor (AEF) is believed to play an integral role in the pathogenesis of amyloidosis. Though its complete biochemical characterization remains elusive, AEF is routinely extracted from the tissue of amyloidotic animals where its activity is evident approximately 24-48 hours before the histological detection of amyloid fibrils (5). The hallmark characteristic of AEF is its ability to accelerate AA fibril deposition. While most experimental induction protocols require several weeks for the development of detectable amyloid deposits, administration of AEF in conjunction with an inflammatory stimulus results in fibril deposition within 24 to 72 hours (6). The ability of an exogenous source of AEF to shorten the induction period required for amyloid deposition suggests that it acts by precluding the lag time required for the endogenous production of AEF (7). Furthermore, genetic analysis of amyloid resistant and susceptible strains of mice suggests that susceptibility to the development of experimental AA amyloidosis is controlled by a single dominant gene (8). This observation, in conjunction with the finding that resistance in A/J mice can be overcome by i.v. injection of an exogenous source of AEF (9), further supports the role of AEF as a major determinant in the development of AA amyloidosis.

Reticuloendothelial cells have long been recognized in close association with amyloid deposits *in situ* (10). Normal peripheral blood monocytes are known to be capable of mediating SAA degradation *in vitro* through the action of cell surface enzymes (11, 12). Studies aimed at elucidating the exact relationship between macrophages or monocytes and the deposition of amyloid fibrils have led to the discovery that mononuclear leukocytes from healthy individuals are capable of mediating complete SAA degradation whereas the cells of amyloidotic patients produce an intermediate protein fragment, both the same size as AA and immunologically cross-reactive with AA (11). In addition, normal murine Kupffer cells have been found to degrade SAA completely, whereas Kupffer cells from amyloid-induced animals produce the AA intermediate (13). This observation has led to the hypothesis that tissue amyloid deposition may be due to inappropriate or incomplete SAA degradation by cells of the monocyte/macrophage lineage, resulting in the accumulation of AA.

In light of the obvious importance of both AEF and reticuloendothelial cells in the formation of amyloid deposits, the aim of the present study was to elucidate the relationship between AEF and macrophages at the primary site of amyloid deposition in AA amyloidosis. Our working hypothesis was that an alteration in the normal processes of macrophage activation resulting from the production of endogenous AEF could cause aberrant SAA degradation by macrophages. Circumstantial evidence supporting our hypothesis is provided by the results of previous studies which showed that the appearance of AEF was correlated with the triggering of fibril accumulation in vivo (5). Furthermore, preliminary work done in our lab detected a decrease in splenic macrophage surface MHC Class II (Ia) expression which coincided with the appearance of endogenous AEF during induction of experimental AA amyloidosis by a chronic inflammation protocol (unpublished results). AEF could interfere with macrophage degradative processes by blocking activation resulting in the inability of the macrophage to completely degrade the SAA. The work presented here therefore examines macrophage activation on a functional level as well as on a phenotypic level following the induction of AA amyloidosis using AEF in combination with silver nitrate (AgNO3) as an inflammatory stimulus.

Certain macrophage-specific activities such as phagocytosis, bactericidal function, and respiratory burst functions are enhanced during activation (14-16) and are therefore good functional indicators of macrophage activation status. Our experiments compare the *in vivo* ability of liver and spleen cells to phagocytose and kill *Listeria monocytogenes* as well as the *in vitro* oxidative burst response (as indicated by O2<sup>-</sup> production) of splenic macrophages between amyloidotic animals treated with AgNO3 in conjunction with AEF and animals treated with AgNO3 alone. In addition, we examined the differential expression of Ia antigen by macrophages isolated from the spleens of mice treated with AEF and AgNO3 using flow cytometry and immunofluorescence analysis. The enhanced expression of MHC Class II molecules may be used as a marker of macrophage activation (17,18). Our aim was to determine whether or not treatment with AEF and silver nitrate causes a change in macrophage surface Ia expression over treatment with the inflammatory stimulus alone and, if so, could any such change reflect an alteration in the degradation pattern of the SAA precursor leading to AA amyloid fibril deposition.

# MATERIALS AND METHODS Animals

Specific pathogen free (SPF) C57B1/6NHsd male mice, 8-12 weeks old, were obtained either from Harlan Sprague Dawley (NHsd; Indianapolis, IN) or bred in our own SPF facilities. Food and water were provided *ad libitum*. Animals were sacrificed by CO<sub>2</sub> narcosis followed by cervical dislocation.

# Preparation of Amyloid Enhancing Factor and Induction of Amyloidosis

AEF was prepared from the spleens of C57BL/6 mice, subcutaneously injected for 21 days with 13% casein (United States Biochemical, Cleveland, Ohio). AEF was extracted from homogenized spleens in ice-cold 30% glycerol-10 mM Tris-HCl, pH 7.5 buffer as previously described (6). 0.5 mg of the lyophilized AEF product (an amount previously determined to induce significant amyloid deposition within 24-48 hours [9]) was resuspended in 0.2 ml sterile phosphate buffered saline (PBS) and administered via i.v. injection concomitant with 0.5 ml s.c. of 2% AgNO3 solution.

Spleens from untreated, healthy mice were subjected to the same extraction protocol as above. The resulting product was used as a control, AEF-free extract. Control groups consisted of animals treated with AEF-free extract/AgNO3 or with 0.5 ml 2% AgNO3 solution or with 0.5 mg AEF alone. They were housed in the same facilities and were sacrificed after the same time periods as experimental mice.

# **Detection of AA Amyloid Deposits**

At least a portion of each spleen was fixed in buffered 10% formalin, embedded in parrafin, sliced in 5 um sections, then stained with alkaline Congo Red (19). The presence of amyloid deposits was determined microscopically by the detection of green birefringence characteristic of AA fibrils under polarized light.

# In Vivo Phagocytosis and Bactericidal Activity

Three days after the injection of AEF alone, AgNO3 alone, or both AEF and AgNO3, experimental animals were inoculated with  $8 \times 10^6$  CFU *L. monocytogenes*. The livers and spleens of infected animals were tested for their ability to phagocytose and kill *L. monocytogenes* by determining the number of viable bacteria in individual organs 30 min. or 6 hours respectively post-infection (14). Spleens and livers were removed aseptically and homogenized in sterile saline for about 1 minute with a Tri-R Stir-R homogenizer (Tri-R instruments, Rockville Centre, NY.). Serial 10-fold dilutions of these homogenates were

plated on tryptose agar (1%; Bacto-agar; Difco Laboratories; Detroit, MN.) and colony counts were routinely executed following 18-24 hour incubations at 37 C. Five animals per group were tested.

# Spleen Macrophage O2<sup>•</sup>Production

Responses were examined in splenic macrophages from naive mice as well as mice treated with AEF alone, AgNO3 alone, or AEF/AgNO3 3 days prior. Internal experimental positive controls consisted of proteose peptone (PP)-elicited peritoneal macrophages.

Spleen cells were flushed from isolated spleens with ice-cold RPMI 1640 supplemented with 0.3 mg/ml glutamine and 50.0 ng/ml gentamicin. Red blood cells were lysed by hypotonic shock on ice. Cell differentiation was determined on cytospin slides (Shandon, Sewickley, PA.) stained with Giemsa (DiffQuik, Canlab, Montreal, Que.). Each cell suspension was adjusted to  $1.5 \times 10^6$  macrophages /ml in RPMI 1640 +10% fetal calf serum. Macrophages were plated in 6- or 24-flat bottom well plates (Linbro, Flow Laboratories, Maclean, VA.) and purified by adherence overnight at 37 C, 5% CO<sub>2</sub>. Each cell suspension was assayed in triplicate. In all instances, cell viability as established by trypan blue exclusion test was > 90-95%. The cell monolayers were > 90% macrophages as determined morphologically by Giemsa stain and phagocytosis of *L. monocytogenes*.

 $O_2^-$  production was measured following treatment with PMA (0.5 ug/10<sup>6</sup> cells) for 75 minutes by the method of Cohen et al. (20) The protein content of each well was then determined following cell lysis with 0.5 <u>N</u> NaOH using the B10-Rad protein assay kit (21). Results are reported as nmoles  $O_2^-/mg$  protein/75 minutes ± SEM. Statistical analysis was done by a two-tailed Mann-Whitney U test with a present probability level of a =0.05. Mean values represent data obtained from three experiments.

### Ia Surface Expression

After aseptic removal, each spleen was flushed with ice-cold RPMI 1640 supplemented with glutamine and gentamicin. Following lymphoid cell purification on Lympholyte-M (Cedarlane Laboratories, Hornby, Ontario), the cells were washed by centrifugation ( $4^{\circ}$ C) and resuspended in 10 ml ice-cold RPMI 1640. Enumeration of cells and determination of viability (by stain exclusion) was accomplished using a hemocytometer and Turks or trypan blue stain respectively. Slides for cell differential counts were prepared as previously described. Following resuspension at a final concentration of 1-2 x 10<sup>6</sup> cells/ml with 0.5% bovine serum albumin (BSA) in PBS, 1x 10<sup>6</sup> cells were aliquoted to 5 ml polystyrene round-bottom tubes (Becton Dickinson labware, Montreal, Que.) for immunofluorescent labelling. Following incubation with goat IgG (Sigma, St. Louis, MO.) to reduce non-specific binding by blocking Fc receptors, cells were labelled successively with monoclonal anti-murine Mac.1 antibody (clone M1/70, Boehringer-Mannheim Biochemical, Laval, Que.), phycoerythrin-conjugated goat F(ab')<sub>2</sub> anti-rat IgG (Caltag Laboratories, San Francisco), and fluorescein isothiocyanate-conjugated murine anti-Ia<sup>k</sup> IgG (Cedarlane Laboratories). Each incubation with labelling antibody (at 1/20 dilution as determined optimal by serial titration) was carried out for 15 min. on ice. The appropriate control labellings were prepared to correct for autofluorescence, non-specific binding, and inhibition of binding due to steric hindrance for each experiment group. Labelled cells were fixed in ice-cold paraformaldehyde and kept at 4 ° C until analyzed on a FACScan flow cytometer with the Lysis II program (Becton Dickinson) the following day. When indicated, IFN-gamma treatment consisted of 5000 units of rat recombinant IFN-gamma (Amgen, Thousand Oaks, CA.) administered i.p. on days 0 and 2 of the experimental design.

# RESULTS

# In Vivo Phagocytosis and Bactericidal Activity During AEF-Induced AA Amyloidosis

In order to determine whether or not the ability of splenic and hepatic macrophages to phagocytose and kill *L. monocytogenes* is modified during AA fibril deposition, groups of mice were injected with AEF and AgNO3 3 days prior to i.v. inoculation with *L. monocytogenes*. Control groups consisted of naive mice and mice injected with AEF or AgNO3 alone.

Phagocytosis by spleen and liver cells was measured 30 minutes following the i.v. injection of  $2 \times 10^8$  bacteria *L. monocyto genes*. The development of amyloid deposits due to AEF/AgNO3 treatment was not found to be associated with modified bacterial uptake as compared to naive animals or animals treated with AEF or AgNO3 alone (Fig. 2.1). Approximatly 95% of the bacterial inoculum was taken up by the spleen and liver in the first 30 minutes following injection.

The bactericidal capacity of macrophages (measured 6 hours after the injection of *L. monocytogenes*) was not significantly modulated in the liver of AEF/AgNO3 treated animals despite amyloid deposition after 3 days when compared to naive animals or animals treated with AEF or AgNO3 alone (Fig. 2.1). All groups were shown to be capable of clearing approximatly 85-90% of the bacterial load within the first 6 hours following infection. No significant bactericidal activity was observed during the same time period in the spleen of both control and experimental animals.



Fig. 2.1: In vivo phagocytosis and bactericidal activity measured in the liver of mice treated with AEF/AgNO3 as compared with mice treated with AgNO3 or AEF alone. Mice were treated with AEF and/or AgNO3 three days prior being infected with  $2 \times 10^4$  bacteria L. monocytogenes. The in vivo phagocytic ability of the liver cells was determined 30 min. following infection while the *un vivo* bactericidal activity was measured 6 hours later, as described in materials and methods. Normal mice displayed similar phagocytic and bactericidal activities observed in mice treated with AEF and/or AgNO3.

# **O2**<sup>-</sup> Production in AEF-Induced AA Amyloidosis

In order to determine whether the respiratory burst functions of splenic macrophages is altered during AA fibril deposition due to AEF, O2<sup>-</sup> production by cells isolated from mice 3 days following treatment with AEF/AgNO3 or either agent alone was measured (Table 2.1). Splenic macrophages from mice injected with AgNO3 alone (a non-amyloidogenic treatment) released a significantly higher concentration of O2<sup>-</sup> metabolite (following stimulation *in vitro* with PMA) than those obtained from the untreated control group. Remarkably, when animals were treated with both AEF and AgNO3, the ability of their spleen macrophages to release O2<sup>-</sup> was significantly decreased as compared to cells isolated from the AgNO3 treatment group. These results suggest that in an amyloid-inducing treatment, AEF partially blocks AgNO3-induced O2<sup>-</sup> production by spleen macrophages.

This pattern of partial inhibition also occurred at the level of H<sub>2</sub>O<sub>2</sub> production. Again, the amyloidogenic treatment was accompanied by a decrease in the AgNO<sub>3</sub> -induced release of the relevant metabolite. Data obtained for H<sub>2</sub>O<sub>2</sub> release however, did not reach significance.

# Ia Surface Expression During AEF-Induced AA Amyloidosis

The injection of a combination of AEF and AgNO3 (as an inflammatory stimulus) is known to induce amyloid fibril deposition in mice within 24-48 hours (5). AgNO3 alone was injected into mice in a parallel group n order to provide a basis of comparison of the level of activation of splenic macrophages harvested from animals subjected to an inflammatory stimulus in the absence of AEF. Using a double immunofluorescence protocol, we determined the level of surface Ia antigen expression on Mac.1<sup>+</sup> spleen cells by flow cytometry, as a means of monitoring the activation state of the macrophages, three days following treatment of mice in vivo. Results presented in Figure 2.2 reveal that maximal surface expression of Ia antigens by splenic macrophages occurred in response to the injection of AgNO3 alone which does not produce amyloid deposits in the tissues. When this treatment was combined with in injection of amyloid-inducing AEF, however, there was a complete abrogation of the augmented macrophage Ia expression. Control preparations of AEF-free spleen extracts did not interfere with the inflammation-induced Ia expression suggesting that the observed down-regulation in surface Ia is a specific effect of AEF. AEF treatment without concomitant AgNO3 treatment, caused a slight but insignificant increase in Ia expression relative to untreated animals (Fig. 2.2).

# Table 2.1:

O2 Production by Macrophages in AEF Induced AA Amyloidosis

Treatment Group	O2- Production <sup>+</sup>	
	(nmoles/mg protein/75 i ± SEM	
Control	92 ± 9.20	
AEF alone	$111 \pm 1760$	
Silver Nitrate alone	$507 \pm 52.64$	
AEF and Silver Nitrate	$218 \pm 32.29 * t$	

\*O2-Production was measured according to the method described by Cohen et al.(20) Results obtained represent means  $\pm$  SEM of 3 experiments (each done in triplicate) \*\*significantly less (a<.05) than value obtained with silver nitrate alone



Fig. 2.2. Modulation of spleen macrophage Ia expression from normal steady state in mice rendered amyloidotic by treatment with AEF/AgNO3 as compared with mice treated with AgNO3 or AEF alone, or a control, AEF-free spleen extract/AgNO3. Spleen cells were isolated from mice three days following the indicated treatments. Cells were labelled with Ia and Mac.1-specific antibodies, as described in Materials and Methods, prior to analysis by flow cytometry. This double antibody label system distinguishes macrophages expressing Ia antigen on their surface (Mac.1+Ia<sup>+</sup>) from B cells (Mac.1-Ia<sup>+</sup>) and surface Ia negative macrophages (Mac.1+Ia<sup>-</sup>). Granulocytes and natural killer cells, which express only small quantities of Mac.1 antigen, are weakly Mac.1<sup>+</sup> relative to macrophages (22) and therefore do not stain intensely enough to be included in Mac.1<sup>+</sup> determinations. Scatter gates were set to exclude dead cells and debris. Results are expressed as a percentage of Mac.1<sup>+</sup> cells which are Ia<sup>+</sup>. As indicated on the x-axis, analysis of spleen cells obtained from untreated, control mice revealed that 16.5% of Mac.1<sup>+</sup> spleen cells are Ia<sup>+</sup> under normal steady state (NSS) conditions 1-3 groups (of at least 3 mice each) were tested for each treatment.

Having established that AEF may indeed downregulate the normal macrophage response to activating agents, at least at the Ia level, we examined the possibility that the observed downregulation may be related to a change in the degradation pattern of SAA leading to AA amyloid fibril deposition. By combining an injection of IFN-gamma (in a dose previously determined to be sufficient to induce increased la expression by splenic macrophages in C57BI/6 mice [unpublished results]) with our AEF/AgNO3 treatment protocol, we were able to restore Ia expression to levels seen with AgNO3 alone (Fig. 2.3). Histological examination of portions of spleen retained from mice treated with all three agents however, reveal that despite the IFN-gamma-induced restoration of macrophage Ia expression in mice administered AEF and AgNO3, splenic amyloid fibril deposition still occurred (Fig. 2.4, Table 2.2). This finding suggests that AEF-induced downregulation in splenic macrophage activation observed at the level of la expression is not related to aberrant SAA degradation leading to AA amyloid fibril deposition.

## DISCUSSION

Based on studies reported in the literature (11, 13), we postulate that the degradation of amyloid precursors by macrophages is a critical event which prevents the deposition of amyloid fibrils in normal tissues. The process of macrophage activation involves a regulated series of changes in physiology which culminate in the ability to execute complex enzymatic functions (18, 22). The ability of a macrophage to perform certain specific functions, possibly including the degradation of SAA, may therefore be dependent on its activation state. In light of the fact that the appearance of endogenous AEF is directly correlated with the deposition of amyloid fibrils (5) and our preliminary observation that splenic macrophage surface Ia expression is decreased during the induction of amyloid fibril deposition (unpublished results), we suspected that AEF may indeed affect the ability of the macrophage to be appropriately activated for the degradation of the amyloid precursor, leading to the deposition of amyloid fibrils.

Since the mechanism by which SAA and macrophages interact remains unknown, we focussed our studies on functions known to be enhanced in the activated macrophage. Macrophage phagocytic and microbicidal activity was monitored by the *in vivo* uptake and killing of the facultative intracellular pathogen *L. monocytogenes* which has been shown to be processed by macrophages (14). Our results indicate that AEF, whether administered alone or in conjunction with AgNO3, does not affect either capacity.

Primed and activated macrophages exhibit an increased release of reactive oxygen metabolites (23). Treatment of animals with an inflammatory agent, such as AgNO3,



Fig. 2.3 Restoration of spleen macrophage Ia expression in mice treated with AEF/AgNO3 by *in vivo* IFN-gamma treatment. Spleen cells isolated from mice 3 days following *in vivo* treatments indicated in the figure (described in Materials and Methods) were prepared and analysed as described in Fig. 2.1. 1-3 groups (of at least three mice each) were tested for each treatment.



Fig. 2.4. Detection of amyloid deposition by Congo Red stain. Amyloid fibril deposits within spleen sections obtained from an AEF/AgNO3 treated animal exhibit characteristic green birefringence under polarized light (panel A) once stained with Congo Red. Spleen sections from normal animals do not exhibit birefringence under polarized light (Panel C). Panels B and D show Congo Red stained spleen sections from amyloidotic and untreated mice respectively, as observed under a normal microscope. Spleen sections were prepared as described in Materials and Methods.

# Table 2.2 Incidence of Amyloid Deposition in Spleen Sections from Mice Treated with AEF, Silver Nitrate, and IFN-gamma

Trearment Group"	Amyloid Deposition**
AEF alone	0/3
Silver nitrate	0/3
AEF/silver nitrate	9/9
Control AEF-free extract	0/3
AEF/IFN-gamma	0/3
Silver nitrate/IFN-gamma	0/6
AEF/silver nitrate/IFN-gamma	6/6

\*Treatment groups as described in Materials and Methods. \*\*sections were stained with Congo Red as described in Materials and Methods. Amyloid fibril deposition was detected by characteristic green birefringence under polarized light.

results in an enhanced macrophage oxidative metabolism (15) Our experiments measuring reactive oxygen metabolite production by splenic macrophages reveal that AEF dampens O2<sup>+</sup> production by AgNO3-induced macrophages and may similarly affect H<sub>2</sub>O<sub>2</sub> production. At this point, one can only speculate as to the mechanism by which AEF modulates O2<sup>+</sup> production. It is concervable, however, that AEF-inhibition occurs at the level of membrane components of the phagocyte superoxide generating system, for example by interfering with NADPH oxireductase cytochrome C activity. When AEF was administered in the absence of AgNO3 no significant modification in metabolite release was observed as compared to steady state, suggesting that the down-regulatory action of AEF is dependent upon a concomitant inflammatory stimuli. It is probable that AEF either acts in concert with other factors induced or released during an inflammatory response or, exerts an inhibitory effect only on macrophages rendered responsive through inflammatory stimuli. This would explain why AEF triggers AA fibril deposition only when administered in conjunction with an inflammatory stimulus.

Having thus determined that AEF may selectively affect some macrophage activation-associated functions but not others we chose to examine the effect of AEF on macrophage activation at the phenotypic level. Increased MHC class II (la) surface expression is considered a marker of macrophage activation (18,24,25) and is therefore useful for monitoring macrophage activation status. We first compared surface Ia expression by splenic macrophages harvested from mice subjected to amyloid depositinducing and non-inducing treatment protocols. Our data indicate that AgNO3 treatment alone, ineffective in causing amyloidosis, results in splenic maciophage activation as reflected by enhanced Ia expression over normal steady state. This increased expression is dampened or blocked in the presence of AEF The absence of any such "block" in la expression in mice treated with AEF-free spleen extract (obtained from control, untreated mice) in conjunction with AgNO3 suggests that decreased Ia expression is attributable to AEF and not some contaminant within our AEF preparation. Furthermore, it is interesting to note that AEF treatment alone revealed a slight increase in Ia expression levels over normal steady state. As indicated by our respiratory burst studies, AEF seems to exert an inhibitory effect only in the event of a concomitant inflammatory stimulus Again, this observation may be related to the fact that AEF is amyloidogenic only when administered in conjunction with an inflammatory stimulus.

Having thus linked decreased splenic macrophage Ia expression to AEF, we looked at whether the observed suppression of macrophage Ia expression in the presence of AEF played an integral role in altered SAA degradation leading to AA fibril deposition. By counteracting AEF-induced down-regulation of splenic macrophage Ia expression with IFN-gamma, we were able to show that the effect of AEF on macrophage activation status during the induction of amyloidosis at the level of Ia expression is not necessary for the production and deposition of AA fibrils in the spleen. It is possible that IFN-gamma increases Ia expression via a different pathway than that which may be blocked by AEF. Studies on the regulation of macrophage surface Ia expression suggest at least four activation-associated transductional signal cascades (i.e. cAMP cascade, Ca<sup>++</sup> and protein kinase C cascades) may be important for its regulation (18,26). It is therefore conceivable that Ia expression could be restored by IFN-gamma despite continued inhibition of an alternative cascade by AEF. Should this alternative transduction cascade also mediate SAA degradation (i.e. by transcriptional regulation of SAA metabolic enzyme genes), the end result could be that AEF blockage of one macrophage activation pathway (causing aberrant SAA degradation) could persist despite replenished Ia expression due to IFN-gamma.

These studies have established that, in the presence of an inflammatory stimulus, AEF selectively modifies certain aspects of the activated splenic macrophage. It is probable that AEF induces amyloid fibril deposition by altering the macrophage function responsible for normal or aberrant degradation of SAA. Studies such as those described here will prove most revealing once the exact nature of this function has been characterized.

## **ACKNOWLEDGEMENTS**

I would like to thank Mrs. Ping Wang for her technical help. This work was supported by a grant from the Medical Research Council of Canada (MT-10250).

# REFERENCES

 (1). Hoffman, J.S., Ericsson, L.H., Eriksen, N., Walsh, K.A., Benditt, E.P. 1984. Murine tissue amyloid protein AA. NH2-terminal sequence identity with only one of the two serum amyloid protein (apo-SAA) gene products. J. Exp. Med. 159: 641-646.
 (2). Meek, R.L., Hoffman, J.S., Benditt, E.P. 1986. Amyloidogenesis: one serum amyloid A isotype is selectively removed from the circulation. J Exp. Med. 163: 499-510.
 (3). Glenner, G.G. 1980. Amyloid deposits and amyloidosis- the beta-fibrilloses. I N Engl. J. Med. 302: 1283-1292.

(4). Cohen, A.S., Cathcart, E.S. 1972. Casein-induced experimental amyloidosis. Review of the cellular and immunological aspects. *Methods Achiev. Exp. Pathol.* 6<sup>,</sup> 207-242.
(5). Axelrad, M., Kisilevsky, B. 1979. Biological characterization of amyloid-enhancing factor. In: <u>Amyloid and Amyloidosis</u>. Proc. 3rd. Symposium on Amyloidosis(Glenner, G.G., Costa, P.P., deFreitas, F.F., Eds.) Excerpta Med.: Amsterdam, p. 527-533.
(6). Axelrad, M., Kisilevsky, R., Willmen, J., Chen, S.J., Skinner, M. 1982. Further

characterization of amyloid-enhancing factor. *Lab. Invest.* 47: 139-146.

(7). Kisilevsky, R., Boudreau, L. 1983. Kinetics of amyloid deposition. The effects of amyloid-enhancing factor and splenectomy. *Lab. Invest.* 48: 53-59.

(8). Wohlgethan, J.R., Cathcart, E.S. 1979. Amyloid in A/J mice is determined by a single gene. *Nature*, 278: 453-454.

(9). Gervais, F., Hebert, L., Skamene, E. 1988. Amyloid-enhancing factor: production and response in amyloid-susceptible and resistant strains. *J. Leuk. Biol.* 43: 311-316.

(10). Gueft, B., Ghidoni, J.J. 1963. The site of formation and ultrastructure of amyloid. *Am. J. Path.* 43: 837-854.

(11). Lavie, G., Zucker-Franklin, D., Franklin, E.C. 1980. Elastase-type proteases on the surface of human blood monocytes: possible role in amyloid formation. *J. Immunol* 125: 175-180.

(12). Skogen, B., Thorsteinsson, L., Natvig, J.B. 1980. Degradation of protein SAA to an AA-like fragment by enzymes of the monocytic origin. *Scand. J. Immunol* 11: 533.

(13). Fuks, A., Zucker-Franklin, D. 1985. Impaired Kuppfer cell function precedes development of secondary amyloidosis. *J. Exp. Med* 161: 1013-1028.

(14). Berche, P., Gaillard, J-L., Sansonetti, P.J. 1987. Intracellular growth of *Listeria* monocytogenes as a prerequisite for *in vivo* induction of T-cell mediated immunity. J. *Immunol.* 138:2266.

(15). Babior, B.M. 1984. Oxidants from phagocytes: agents of defence and destruction. *Blood* 64:959.

(16). Nathan, C.F. 1983. Secretory products of macrophages. J. Clin. Invest. 79:319.

(17). Ezekowitz, R.A.B., Austyn, J., Stahl, P.D., Gordon, S. 1982. Surface properties of the Bacillus Calmette-Guerin-activated mouse macrophage. Reduced expression of mannose specific endocytosis, Fc receptors, and antigen F4/80 accompanies induction of Ia. J. Exp. Med. 154: 60-76.

(18). Adams, D.O., Hamilton, T.O. 1987. Molecular transductional mechanisms by which IFN and other signals regulate macrophage development. *Immunol. Rev.* 97: 5-27.

(19). Puchtler, H., Sweat, F., Levine, M. 1962. On the binding of Congo Red by amyloid. J. Histochem. Cytochem. 10: 355-364.

(20). Cohen, H.J., Chovaneic, M.E. 1978. Superoxide generation by digitonin stimulated guinea pig granulocytes. A basis for a continuous assay monitoring superoxide production and for the study of the activation of the generating system. J. Clin. Invest. 61:1081.

(21). Bradford, M. 1976. A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein dye binding. *Anal. Biochem.* 72:248.
(22). Ho, M.K., Springer, T.A. 1982. Mac.1 antigen: quantitative expression in macrophage populations and tissues, and immunofuorescent localization in the spleen. *J. Immunol* 128 (5): 2281-2286.

(23). Johnston, R.B.Jr. 1981. Enhancement of phagocytosis-associated oxidative metabolism as a manifestation of macrophage activation. *Lymphok*. 3:33.

(24). Lafuse, W.P. 1991. Molecular biology of murine MHC Class II genes. Crit. Rev. Immunol. 11 (3, 4): 167-194.

(25). Nibbering, P.H., Langermans, J.A.M., Van de Gevel, J.S., Van der Hulst, M.B., Van Furth, R. 1991. Nitrite production by activated murine macrophages correlates with their toxoplasmastatic activity, Ia antigen expression, and production of H<sub>2</sub>O<sub>2</sub>.

Immunobiol. 184(1): 93-105.

(26). Uhing, R.J., Adams, D.O. 1989. Molecular events in the activation of murine macrophages. *Agents and Actions*. 26: 9-14.

CHAPTER 3 PURIFICATION OF AMYLOID ENHANCING FACTOR

. -

\_ \_ \_

#### **INTRODUCTION**

Though first detected in amyloid tissue extracts in 1966 (1), the biochemical nature of AEF remains a mystery. At present AEF is defined most clearly in functional terms as a non-inflammatory, non-amyloidogenic substance that, when co-administered with a single inflammatory stimulus, shortens the pre-amyloid phase of amyloidosis to result in splenic amyloid deposition within 24 to 48 hours. Under conditions of sustained inflammation in animals, deposition takes approximately 15 days (2). The consistent detection of AEF in several forms of amyloidosis including AA, AL, β-amyloid, and transthyretin amyloid (3,4) suggests that AEF activity may be a common link between the diverse forms of amyloid (2).

The purification of AEF is essential to the progression of research into the pathogenic mechanisms leading to the formation of amyloid deposits. A significant portion of my work has concentrated on identifying the particular protein responsible for AEF activity. Positive identification of the exact protein obviously preceeds complete biochemical characterization of AEF including amino acid sequencing or preparation of AEF-specific antibodies. Besides opening the door to endless research possibilities such as the development of tissue culture systems as *ex-vivo* models of AA amyloid generation or methods for monitoring pathological conditions associated with the increased risk for amyloidosis, discoveries of this nature are essential to the study of AEF production kinetics, potency, and sites of action.

To provide sufficient raw material, AA amyloidosis was induced in large batches of C57Bl/6 mice. Mass amounts of a relatively crude preparation of AEF was then extracted from the amyloidotic spleen of these mice and separated on multiple acrylamide-SDS preparative gels. Protein electroeluted from specific regions of these gels was then assayed for AEF activity allowing us to define the specific area of the gels containing AEF activity. Two dimensional (2-D) electrophoretic analysis of protein contained within this "AEF-active" area has identified 3 proteins of interest not expressed in "AEF-free" controls.

#### MATERIALS AND METHODS

#### Animals

C57Bl/6J male mice were obtained from Jackson Laboratories (Bar Harbor, ME) or bred in our own facilities. Food and water were provided *ad libitum*. Animals were sacrificed by CO<sub>2</sub> narcosis followed by cervical dislocation.

# **Induction of AA Amyloidosis**

AA amyloidosis was induced in C57Bl/6 mice by subcutaneous injection with 0.5 ml 13% casein (United States Biochemical, Cleveland, Oh.) prepared in sterile 0.3 M NaHCO<sub>3</sub>, pH 7.0 (Gibco Laboratories, Grand Island, NY) for 21 consecutive days. The NaHCO<sub>3</sub> and the casein solutions contained less than 0.02 ng hpopolysaccharide/ml as measured by the E-toxate (Limulus ameobocyte lysate) detection and semi-quantitation test (Sigma, St. Louis, MO). Spleens were harvested on the 22nd day and frozen at -70 C in 30% glycerol-10 mM Tris (Fisher Scientific, Fairlawn, NJ) buffer at pH 7.5.

## **Detection of Amyloid Deposits**

Whole or partial spleen were fixed in buffered 10% formalin then immersed in parrafin. Five um histiological sections prepared from the wax-embedded organs were then stained with alkaline Congo Red (6). The presence of amyloid deposits was then determined microscopically by the detection of green birefringence characteristic of AA fibrils under polarized light.

# **Preparation of AEF Extract**

AEF was extracted from the spleen of amyloidotic C57Bl/6J mice as previously described (5). Spleens were homogenized in ice-cold 30% glycerol-10 mM Tris buffer at pH 7.5 with a Tri-R Stir-R homogenizer (Tri-R Instruments, Rockville Centre, NY). Following sonication for 1 min. with a Sonic Dismembrator (Artek Systems Corporation, Farmingdale, NY), the preparation was mixed at 4 C for 1 hour then centrifuged at 250 000 x g, 4 C, for 3 hours. The supernatant was then dialysed against five changes of an 80fold volume of 0.01 M phosphate buffered saline (PBS) pH 7 in SpectraPor 3 (m. wt. cutoff 3 500 Da) dialysis bags (Spectrum Medical Industries Inc., Los Angeles, CA). The precipitate was recovered by centrifugation at 17 300 x g, 4 C, for 30 min. Following resuspension of the pellets in small volumes of PBS, sterile solubilisation buffer composed of 6% urea/1% ß mercaptoethanol/1 mM EDTA was added to the preparation in a 1 to 1 volume ratio. Mixing was achieved overnight at room temperature on a Roto-Torque rotator (Cole Parmer Instrument Company, Chicago, II) at 60 Hz. Following centrifugation at 10 000 x g and 10 C for 10 min. to remove unsolubilised particulates, the supernatant was dialyzed against five changes of 0.01 M PBS pH 7 in SpectraPor 3 (Spectrum Medical Industries) dialysis bags. Immediately following dialysis the preparation was frozen at -70 C then lyophilized with a Virtis lyophilizer (Gardiner, NY). This dry AEF preparation was then stored at -70 C until future use.

The AEF activity of all AEF preparations was ensured by their ability to induce amyloid deposition at an accelerated rate. Mice were injected intravenously with 100-300 ug dried AEF extract resuspended in PBS, in conjunction with a subcutaneous injection of  $0.5 \text{ ml } 2\% \text{ AgNO}_3$  (Fisher Scientific) solution in deionized water. Spleen sections were then tested for amyloid deposition within 48-72 hours.

Spleen extract containing no AEF was prepared according to the extraction protocol described above but using spleen from uninduced, healthy mice. The resulting "AEF-free" product was used as a control.

# **Preparative Gel Electrophoresis**

Separation of 3-5 mg of the lyophilized AEF extract described above was achieved by 12.5% sodium dodecylsulfate-polyacrylamide gel electrophoresis (SDS-PAGE) according to the method of Laemmli (7). Using Bio-Rad Protean II preparative gel apparatus (Bio-Rad, Richmond, CA), 1.5 mm thick SDS-polyacrylamide gels were prepared and polymerized at room temperature then stored overnight. Complete gels consisted of a 4.5% stacking gel and a 12.5% resolving gel. Both stacking and resolving gel solutions were thoroughly vacuum degassed prior to initiation of polymerization with 0.05% ammonium persulfate and 0.05% TEMED. Gels were pre-electrophoresed at a constant voltage of 250 V and approximate current of 175 mA. The running buffer consisted of 0.025 M Tris-HCl, 0.192 M glycine and 0.1% SDS, pH 8.3. Following addition of 1 ml each of double distilled  $H_2O$  (dd $H_2O$ ) and sample buffer, AEF preparations were boiled for 4 min. then applied to the sample well. The sample buffer was composed of 0.063 M l'ris-HCl pH 6.8, 2% SDS, 10% glycerol, and 0.001% bromophenol blue. Rainbow molecular weight markers (Amersham International, Amersham, UK) were run concomitantly in a single small well for reference. These consisted of myosin (200 000 Da), phosphorylase b (97 400 Da), bovine serum albumin (69 000 Da), ovalbumin (46 000 Da), carbonic anhydrase (30 000 Da), trypsin inhibitor (21 500 Da) and lysozyme (14 300 Da). The markers were prepared in a 1:1 dilution with sample buffer and boiled as was the experimental sample. During protein migration through the stacking gel, electrophoresis was run at 250 V and 25 mA per gel then at 375 V and 35 mA per gel through the resolving gel. The entire run took from 4 to 5 hours and the current was held constant throughout.

After completion of electrophoresis, vertical strips of 1.0-1.5 cm width were removed from the left and right sides of the gel and stained separately in Coomassie blue R-250 overnight. Small "tabs" were incorporated while cutting these peripheral strips to allow for proper realignment with the main, untreated gel piece following destaining and rehydration of the strips in ddH<sub>2</sub>O. Staining of the sides of the gel in this manner allowed for the visualization of separate protein bands. Using these as reference, distinct bands or regions were then excised from the entire gel. Protein contained within the major,
untreated portion of the gel was therefore not subjected to any unnecessary chemical modification caused by staining. Excised portions of the unstained gel were stored at -20 C until future use.

# Electroelution

Separated proteins were isolated from the preparative gels using the S&S Elutrap® electro-separation system (Schleicher & Schuell, Keene, NH) designed for rapid isolation and purification of macromolecular samples from fractionating gels. In brief, the system consists of a sample chamber limited at each end by an inert membrane with a dense matrix through which buffer ions and molecules less than 3-5 kDa can pass under the influence of an electric field. A second, microporous membrane placed at one end of the chamber, on the inner side of the first membrane, acts as a prefilter that prevents acrylamide and other particulates from entering the purified sample. Together these two membranes form a trap into which the sample migrates. The electric field acts as the driving force filtering the molecules through the membranes; when the voltage is switched off, the membranes seal the trap, preventing the diffusion of the sample out of the trap. Electroelution was carried out in CAPS buffer (10 mM CAPS in 10% methanol pH 11.0 titrated with 2 N NaOH) suitable for high protein transfer efficiency from acrylamide gels (8). After the thawed gel slices were cut into small pieces and placed in the buffer-filled chambers, a 50 V/1 A upper limit current was run through the system for at least 18 hours. Eluted proteins, in final volumes of 400 ul, were then collected individually from the membrane-bound traps. Protein was then precipitated from the CAPS buffer solution at -20 C overnight with a 1 to 4 volume of 1:1 ice-cold acetone: methanol solution. Alternatively, the CAPS-protein solution was dialyzed 3 times against ddH<sub>2</sub>O in SpectraPor 3 dialysis bags (Spectrum Medical Industries) then frozen overnight at -70 C for lyophilization. Proteins recovered from various regions of the preparative gels in this manner were then either injected into mice (as described above for our initial AEF preparation) to test for AEF activity, analysed by 2-D electrophoresis, or used in  $O_2^-$  production assays.

## Two Dimensional (2-D) Gel Electrophoresis

Isoelectric focusing (IEF), the first dimension: IEF was conducted using components of the Multiphor II electrophoresis system apparatus (Pharmacia LKB Biotech., Uppsala, Sweden) on gels consisting of an immobilized pH gradient, ranging from pH 5 to pH 8, obtained by the copolymerization of acrylamide, piperazine diacrylamide (PDA) as a crosslinker, and a series of Immobiline® buffering acrylamide derivitaves.

Prior to IEF the dehydrated gels were incubated for approximately 6 hours in rehydration buffer composed of 8 M urea, and 2.5% Pharmalyte® 3-10 (Pharmacia) as the

carrier ampholyte. Lyophilized AEF samples were dissolved in this same buffer with an additional 2% β-mercaptoethanol and 2% Triton X-100. All solutions were filtered before use with 0.2 um filters. β-mercaptoethanol was omitted from the rehydration step since it produced mercaptan bands at 50 and 70 kDa which were detectable upon silver staining. Samples were centrifuged at 13 600 x g for 15 min. and the supernatants loaded into sample cups. IEF was carried out under a protective layer of light mineral oil to avoid gradient drift caused by atmospheric CO<sub>2</sub>. The system was run at 300 V, 1 mA, 5 W for 3 hours (900 Vh) and 2 500 V, 1 mA, 5 W for 14 hours (35 000 Vh) for a total of 35 900 Vh. IEF gels were equilibrated twice for 10 min. in 125 mM Tris-HCl (pH 6.8) with 1% SDS and 10% glycerol prior to SDS-PAGE.

SDS-PAGE, the second dimension: Vertical, 1.5 mm thick, polyacrylamide/PDA (T=30%, C=2.6%) gradient slab gels were generated using a gradient maker (Hoefer Scientific Instruments, San Francisco, CA) with a dense solution consisting of 0.5 M Tris HCl (pH 8.8), 18% PA, 0.1% SDS, and 10% glycerol (mixing chamber) and a light solution consisting of 0.5 M Tris-HCl (pH 8.8), 12% PA, and 0.1% SDS (resevoir). Polymerization was initiated with the addition of 6 ul TEMED (0.0003% final concentration) to each chamber and 30 ul and 15 ul ammonium persulfate to the light and dense solutions respectively (5x10<sup>-5</sup> final concentration). Gels were poured using a 3channel peristaltic pump and the Bio-Rad Protean II apparatus (Bio-Rad). Complete gels included a 4% stacking gel as well as the resolving portion of the gel. IEF gels were applied flat onto the surface of the stacking gel. Electrophoresis was carried out at 25 mA/gel through the stacking gel, then at 50 mA/gel through the separating gel. Migration through the separating gel required approximately 3 hours. Rainbow molecular weight markers (Amersham International) included in the second dimension were diluted with 10 parts SDS-PAGE sample buffer (4 mM Tris-HCl pH 6.8, 4% SDS, 16% glycerol) with 10% B-mercaptoethanol and 0.006% bromophenol blue, then boiled for 3 min. prior to loading.

Once electrophoresis in the second dimension was completed the gel was silver stained according to the procedure devised by Bürk *et al* (9).

# RESULTS

# **SDS-PAGE** Analysis of AEF Extract

When denatured by heating in the presence of excess SDS, most polypeptides bind SDS in a constant weight ratio such that they have essentially identical charge densities and migrate in polyacrylamide gels according to their size (10). A plot of  $log_{10}$  polypeptide

64

molecular mass versus relative mobility (R<sub>f</sub>) reveals a straight line relationship. Using the migration distances of our marker polypeptides [myosin, 200 kilodaltons (kDa); phosphorylase b, 97.4 kDa; bovine serum albumin, 69 kDa; ovalbumin, 46 kDa; carbonic anhydrase, 30 kDa; trypsin inhibitor, 21.5 kDa; lysozyme, 14.3 kDa], a standard curve was constructed and used to calculate the apparent molecular mass of distinguishable bands resolved by electrophoretic analysis of our AEF extract. Separation of AEF extracted from amyloidotic spleen on 12.5% SDS-polyacrylamide gels revealed a relatively constant banding pattern comprised of 7 significant bands (Fig. 3.1). Some minor variability in staining intensity and inter-band distances did occur between gels most likely due to minor differences in electrophoretic conditions or between AEF samples prepared in different extraction batches. For this reason molecular weight estimations calculated from each individual prep gel were used to determine the average values reported in figure 3.1. The migration distances of proteins were measured relative to that of the dye front since some differences in protein migration (probably due to differences in temperature over the gel) occurred horizontally across the gels.

The Mr's of bands 1 to 7 (as indicated in Fig. 3.1) are 81.0 kDa  $\pm$  7.7, 73.3 kDa  $\pm$  6.7, 63.2 kDa  $\pm$  5.8, 54.6 kDa  $\pm$  7.4, 47.9 kDa  $\pm$  4.4, 15.6 kDa  $\pm$  6.4, and 12.4 kDa  $\pm$  7.0 respectively. A group of indistinguishable bands were detected in the region relative to the 30 kDa molecular weight marker. Previous studies conducted in our lab have shown that protein electroeluted from this 30 kDa molecular weight region do not exibit AEF activity.

Automated araino acid sequence analysis of band 7 has revealed an amino terminal end sequence identical to that of murine histone 3 protein of either H3.1 or H3.2 subtype (L. Hébert, unpublished data) (11, 12). Neither band 6 nor band 7 were found to exhibit AEF activity.

Electrophoretic analysis of "AEF-free" control extract prepared from the spleen of healthy animals revealed a similar banding pattern with the exception of band 4 which was not detected (Fig. 3.1).

# Isolation of Region Containing AEF Activity

As described in Materials and Methods, protein was electroeluted from specific regions of each preparative gel. Horizontal variability in protein migration and indistinct band edges prevented individual excision of bands 3, 4, and 5 in some cases. These three bands were therefore isolated collectively. I.v. injection of mice with 300 ug protein corresponding to bands 3, 4, and 5, (in conjunction with an injection of AgNO<sub>3</sub>, s.c., as an inflammatory stimulus) resulted in spleen amyloid deposition within 72 hours (Table 3.1). Animals treated with AgNO<sub>3</sub> and 300 ug protein electroeluted from either bands 1 or 2 did not



Figure 3 1 Separation of AEF extract on 12.5% polyacrylamide preparative gels. Strips excised from the vertical edges of preparative gels were stained with Coomassie blue R-250 (refer to Materials and Methods). Molecular weight markers were run in the small lanes appearing to the outermost left of each photograph. Bands 1 to 7 (as described in Results), as well as the region determined to contain protein responsible for AEF activity, are indicated. A. Electrophoretic separation of AEF extract prepared from the spleen of amyloidotic mice. B. Separation of "AEFfree" control extract prepared from the spleen of healthy animals. Note the absence of a detectable band 4 from the "AEF active" region in this case.

Table 3.1 Incidence of AEF Activity Indicated by Amyloid Depositionin Spleen Sections from Mice Treated with Protein Electroeluted fromPreparative Gels

Training Crone	Anarloid Depresition
	0.12
Band 1	0/3
Band 2	0/3
Bands 3, 4, and 5	3/3
Band 6	0/3
Band 7	0/3

Note: Each treatment group received 300 ug of the indicated protein as described in Materials and Methods.

Refer to Fig. 3.1 for protein band designations.

Amyloid deposition was indicated on Congo Red stained spleen sections by characteristic green birefringence under polarized light (as described in Materials and Methods). reveal amyloid deposition within 72 hours. These results indicate that the amyloid enhancing activity of our spleen extract is attributable to either bands 3, 4, and/or 5. We have labelled the region bound by bands 3 and 5 the "AEF-active" region of our electrophoretically separated sample (refer to Fig. 3.1).

# 2-D Gel Electrophoretic Resolution of the AEF-active Region

Following elution from preparative SDS gels, AEF active region proteins were analysed by two dimensional gel electrophoresis. This technique separates proteins according to two independent parameters, namely isoelectric point (pI) and molecular weight (13). Isoelectric focusing is carried out in the first dimension followed by SDS-PAGE in the second dimension. The similarity in protein pattern detected on both gels shown in Fig. 3.2 is indicative of the reproducibility of our results.

2-D analysis of protein electroeluted from the "AEF-active" region of multiple SDSpolyacrylamide preparative gels loaded with AEF extract revealed a protein pattern consisting of 13 distinguishable polypeptides (Fig. 3.2 A). The apparent pI of each polypeptide spot was estimated from migration distances based on the assumption of linear progression within the pH gradient in the first dimension which ranged from pH 5 to pH 8. They are, from spot 1 to spot 13 respectively: 6.2, 6.2, 6.9, 6.7, 5.6, 7.5, 7.1, 7.5, 7.1,  $5.6, \le 5.0, 5.3, \text{ and } \le 5.0$ .

2-D separation of protein eluted from the area corresponding to the AEF-active region (as determined relative to molecular weight markers) of SDS polyacrylamide preparative gels loaded with AEF-free extract is shown in fig. 3.2 **B**. Comparison by eye between **A** and **B** revealed that 3 distinct proteins (labelied 2, 3, and 4 in fig. 3.2) are detected in the AEF-active preparation only. All three proteins exhibit pI's just below 7.0 (6.2, 6.9, and 6.7 respectively) indicating that all three are slightly acidic. Their relative mobilities during SDS-polyacrylamide electrophoresis in the second dimension indicate their Mr's to be 12.0 kDa, 31.6 kDa and < 14.3 kDa respectively. Immunoblotting of both preparative gels and 2-D gels with anti-SAA antibodies (data not shown) showed no reaction thus eliminating the possibility of SAA contamination.

A number of spots (i.e. 1, 5, 6, 9, 10, 11, and 12) appear with greater intensity in the AEF-free control versus the AEF-active sample in Fig. 3.2. This is may be due to differences in the amount of material loaded and/or the quality of staining in each case. The large spot appearing in the lower right corner of each gel is likely an artifact of cathodic drift in the presence of heterogenous low M.W. protein products generated during electroelution and freeze/thawing of samples. Cathodic drift, which occurs due to the continuous passage of cathodic end components of the gel buffer into the resevoir buffer



Fig. 3.2 Two-dimensional gel electrophoretic resolution of the AEF-active region. Proteins are separated according to pl by isoelectric focusing in the first dimension and according to molecular weight by SDS PAGE in the second dimension. Gels were loaded with: A) protein electrocluted from the "AEF active" region of prep gels and B) protein electrocluted from the corresponding region of AEF-free controls Refer to Results for description of each sample † indicates protein spots which are detected in the AEF-active preparation (A) but not in the control preparation (B).

and electroosmosis, can cause major distortions during isoelectric focusing resulting in the loss of resolution of basic proteins (14).

## DISCUSSION

Despite the devotion of considerable research on the subject, the precise biological role of AEF in the pathogenesis of AA amyloidosis has yet to be determined. The elucidation of its amyloid-associated mechanism has been hindered to the greatest extent by the inavailability of a purified AEF preparation. Complete biochemical characterization of AEF in AA amyloidosis would not only indicate possible modes of action by comparison with known proteins, but would allow for the determination of whether AEF activity observed across the spectrum of amyloidotic disorders is attributable to a single factor, or rather a group of factors with similar function. The availability of purified AEF could also facilitate the development of *ex vivo* models of AA amyloid generation using tissue culture systems. Furthermore, should AEF appear to be an enzyme involved in amyloid fibril formation, studies examining its potency, kinetics, and active site may allow for therapeutic intervention with amyloid deposition. In addition, the production of AEF-specific antibodies could be useful for monitoring patients with pathological conditions which put them at risk of developing amyloidosis. Obviously, the possibilities for future research dependant on the determination of the nature of AEF are endless.

Electrophoretic analysis of our active AEF tissue extract under denaturing conditions revealed a relatively consistent composition of 7 significant bands. The estimated molecular weights of these bands (as indicated in Fig. 3.1) are contingent upon a constant SDS to protein binding ratio of 1.4 to 1 (15). Previous periodic acid Schiff's base staining and ionic-exchange HPLC analysis of our extract (Lise Hébert, unpublished observations) however, did not indicate the presence of sugars which are most likely to cause SDS binding modifications (16).

Histones are phylogenetically conserved, basic, chromosomal proteins found associated with nonhistone proteins and nucleac acids within eukaryotic nuclear chromatin (11). The inclusion of murine histone  $3 \pm 35$  prot in (band 7) in our AEF extract, despite solubilization with urea and  $\beta$ -mercaptoe at a nol, is most likely the result of its strong aggregative affinity for other proteins. The identity of band 7 as murine H3 is confirmed by the consistency of its Mr with the published 12.4 kDa molecular weight value for H3 (17).

With exception of band 4, electrophoretic analysis of "AEF-free" control extract revealed the same banding pattern as observed with tissue extracts containing AEF. The

absence of this 54.6 kDa band from the control prep indicates that it may be responsible for AEF activity. The remaining bands, namely 1, 2, 3, 5, 6, and 7, are common to both pathological and non-pathological tissue environments suggesting their presence does not necessarily result in amyloid fibril deposition. The proteins occuring in these latter bands however, may well act as cofactors to the process of amyloid deposition.

The preceeding conclusions are supported by our examinations of AEF activity of specific protein bands eluted from SDS-polyacrylamide preparative gels Only protein eluted from the region of the gels containing bands 3, 4, and 5 revealed AEF activity. Though it is likely that band 4 is responsible for AEF activity, it would be interesting to determine whether the presence of one or both bands 3 and 5 (i.e. as cofactors) are necessary for amyloid deposition to occur.

2-D analysis of protein eluted from the "AEF-active" region (refer to results section) of multiple prep gels revealed a more complex polypeptide composition than expected. The detection of 13 distinct polypeptides from a starting material consisting of only 3 apparent protein bands is due, in part, to the greater resolving power of 2-dimensional separation techniques over electrophoresis within a single dimension. Individual protein species with similar molecular weights are not distinguishable by SDS-PAGE analysis alone. However, if such species differ in inherent pI, a combined IEF/SDS-PAGE procedure can separate them. Furthermore, the SDS-PAGE dimension of our 2-D analysis consisted of a polyacrylamide gradient which has a higher resolving power than SDS-PAGE with constant polyacrylamide composition.

Additionally, protein detection on our 2-D gels is enhanced by the chosen staining method. Silver stain, though innappropriate for preparative gels due to large protein loads, was used. Silver staining methods have been reported to have a 100-fold increased sensitivity over Coomassie Blue methods (18). It is probable that some protein species detected in our 2-D studies were indiscernable on our Coomassie Blue stained preparative gels.

The decrease in Mr general to all protein species as detected on our 2-D gels (recall the "AEF-active" region of preparative gels spans 63.2 kDa to 47.9 kDa) is probably the result of partial degradation during electroelution (i.e. by base-catalyzed proteolysis in CAPS buffer) and lyophilization.

It is unlikely that the increase in number of detected protein species in our 2-D analysis is due to the dissassociation of multimeric proteins into their respective subunits since ß-mercaptoethanol (i.e. as a reducing agent) was included in sample buffers used for preparative gel electrophoresis.

Horizontal and vertical streaking observed on these gels may be the result of electroendosmotic effects produced by grafting of the immobilized pH gradient used in isoelectrofocusing, into the SDS gel matrix. Electroendosmotic alterations can hinder protein transfer from the first to the second dimension ultimately causing streaks (19).

The detection of spots 2, 3, and 4 in AEF-active preparations, but not AEF-free controls, suggests any one of these may be the protein, or a polypeptide derivative of the protein, responsible for AEF activity. 2-D gels, prepared according to the methods described here, are easily amenable to microsequencing via transfer to polyvinylidene difluoride membranes. Proteins blotted onto these chemically inert membranes can be sequenced directly without additional manipulations (20, 21). Microsequencing of the polypeptides of interest has already been initiated. Whole or partial protein sequences generated in this manner will be used to search for protein identity (by comparison with available sequences stored in data banks) as well as for preparing specific DNA probes for cloning as yet uncharacterized proteins.

## REFERENCES

(1) Werdelin, O., Ranlov, P. 1966. Amyloidosis of mice produced by transplantation of spleen cells from casein-treated mice. *Acta Pathol. Microbiol Scand.*; 68:1.

(2) Varga, J., Flinn, M.S., Shirahama, T., et al. 1986. The induction of murine amyloid with human splenic extract. Probable role of amyloid enhancing factor. *Virchows Arch B* 51:177.

(3) Morisette, C., Gervais, F. Induction of amyloid deposition by subfractions of the amyloid enhancing factor extracted from brain of patients with Alzheimer's disease. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) pp. 303. Kluwer:Dordrecht.

(4) Sipe, J.D., De Beer, F.C., Pepys, M., *et al.* 1990. Report of special sessions on bioassays and standardization of arryloid proteins and precursors. In <u>Amyloid and Amyloidosis 1990</u>. (Natvig, J. B., Forre, O., Husby, G., *et al.*, Eds.) pp. 883. Kluwer:Dordrecht.

(5) Axelrad, M., Kisilevsky, R., Willmen, J. et al. 1982. Further characterization of amyloid-enhancing factor. Lab. Invest. 47: 139.

(6) Puchtler, H., Sweat, F., Levine, M. 1962. On the binding of Congo Red by amyloid. *J. Histochem. Cytochem.* 10:355.

(7) Laemmli, U.K. 1970. Cleavage of structural proteins during the assembly of the head of bacteriophage T4. *Nature* 227:680.

(8) Matsudaira, P. 1987. Sequence from picomole quantities of proteins electroblotted onto polyvinylidene difluoride membranes. J. Biol. Chem. 262:10035.

(9) Bürk, R.R., Eschenbruch, M., Leuthard, P., *et al.* 1983. Sensitive detection of proteins and peptides in polyacrylamide gels after formaldehyde fixation. *Methods Enzymol.* 91:247.

(10) Hames, B.D. 1990. One dimensional polyacrylamide gel electrophoresis. In <u>Gel</u>
<u>Electrophoresis of Protein. A Practical Approach.</u> (Hames, B.D., Rickwood, D. Eds.) p.
1. Oxford University:Oxford.

(11) Dayhoff, M.O. 1972. Chromosomal proteins. In: <u>Atlas of Protein Sequence and Structure</u>, vol. 5. p. D-275. National Biochemical Research Foundation: Washington.
(12) Von Holt, C., Brandt, W.F., Greyling, H.J., et al. 1989. Isolation and characterization of histones. *Meth. Enzymol.* 170:431.

(13) O'Farrell, P.H. 1975. High resolution two-dimensional electrophoresis of proteins. J. Biol. Chem. 250:4007.

(14) Dunbar, B.S. 1987. <u>Two-Dimensional Electrophoresis and Immunological</u> <u>Techniques.</u> p. 186. New York: Plenum Press.

(15) Fish, W.W., Reynolds, J.A., Tanford, C. 1970. Gel chromatography of proteins in denaturing solvents. Comparison between sodium dodecyl sulfate and guanidine hydrochloride as denaturants. J. Biol. Chem. 245:5166.

(16) Tanford, C., Reynolds, J.A. 1976. Characterization of membrane proteins in detergent solutions. *Biochim. Biophys. Acta* 457:133.

(17) Sperling, R., Bustin, M. 1975. Dynamic equilibrium in histone assembly: self-assembly of single histones and histone pairs. *Biochem.* 14:3322.

(18) Switzer, R.C., Merril, C.R., Shifrin, S. 1979. A highly sensitive silver stain for detecting proteins and peptides in polyacrylamide gels. *Anal. Biochem.* 98:231.

(19) Gorg, A., Postel, W., Domscheit, A., *et al.* 1989. Methodology of two-dimensional electrophoresis with immobilized pH gradients for the analysis of cell lysates and tissue proteins. In: <u>Two-Dimensional Electrophoresis</u>. (Endler, A.T., Hanash, S.) pp.267. Germany: VCH.

(20) Matsuidaira, P. 1987. Sequence from picomole quantities of proteins electroblotted onto polyvinylidene difuoride membranes. J. Biol. Chem. 262:10035.

(21) Kennedy, T.E., Gawinowicz, M.A., Barzilai, A. et al. 1988. Sequencing of proteins from two-dimensional gels by using *in situ* digestion and transfer of peptides to polyvinylidene difuoride membranes: application to proteins associated with sensitization. *Aplysia. Proc. Natl. Acad. Sci. USA* 85:7008.

CHAPTER 4 MACROPHAGE RESPIRATORY BURST FUNCTION IN THE PRESENCE OF PURIFIED AMYLOID ENHANCING FACTOR

\_ \_\_

## **INTRODUCTION**

In chapter 2 we considered the effects of AEF on macrophage activation and function during amyloidogenesis. These experiments were conducted using a relatively crude extract with AEF activity. The studies described in chapter 3 concentrated on purifying AEF contained within this extract. This fourth and final chapter examines the effects of *ex vivo* AEF treatment of peritoneal macrophages as a continuum of data collected in chapter 2, using the purified AEF preparation devised in chapter 3.

Our experiments measuring reactive oxygen metabolite production by splenic macrophages revealed that AEF dampens  $O_2^-$  production by  $\land$  gNO3-induced macrophages and may similarly affect H<sub>2</sub>O<sub>2</sub> production. We speculated that AEF either acts in concert with factors induced or released during an inflammatory response or is able to exert an inhibitory effect only on those macrophages rendered responsive through inflammatory stimuli. In addition, we suggest that AEF may modulate O<sub>2</sub><sup>-</sup> production by acting at the level of membrane components of the phagocyte superoxide generating system, for example by interfering with NADPH oxireductase cytochrome C activity.

By investigating the effect of AEF on isolated proteose peptone-elicited macrophages we hoped to clarify the nature of the interaction between AEF and the macrophage indeed to determine whether the modulatory effects of AEF are dependent only upon the presence of AEF and the inflammation-induced macrophage.

# **MATERIALS AND METHODS**

Four treatment groups were included in each assay of oxygen radical production. These groups were treated with either AEF extract, with protein electroeluted from the "AEF-active" region of multiple SDS-PAGE preparative gels (as described in Chapter 2), or with a control, AEF-free preparation derived from the spleen of healthy mice. The fourth control group received no treatment.

# Animals

C57Bl/6J male mice were obtained from Jackson Laboratories (Bar Harbor, ME.) or bred in our own facilities. Food and water were provided *ad libitum*. Animals were sacrificed by CO<sub>2</sub> narcosis followed by cervical dislocation.

## Cells

Peritoneal exudate cells were harvested from mice by peritoneal lavage with 10 ml cold RPMI supplemented with glutarnine and gentamycin 3-4 days following

intraperitoneal injection with 1 ml 10 % proteose-peptone (Difco, Detroit, MN.). The proteose peptone had been previously determined to contain <0.1 ng lipopolysaccharide/ml as measured by a chromogenic *Limulus* assay (M.A. Bioproducts). Cells from at least 8 mice were pooled for each assay. The cells were washed twice in ice-cold media then adjusted to  $1.5 \times 10^6$  macrophages/ml in RPMI 1640 + 10% fetal cal<sup>4</sup> serum. Cell differentiation was determined on cytospin slides (Shandon, Sewickley, PA) stained with Giemsa (DiffQuik, Canlab, Montreal, Qué.). Cells were plated on 24-flat bottom well plates (Becton Dickinson Labware, Montreal, Qué.) in 1 ml aliquots then purified by adherance at 37 C for 90 minutes. In each case, cell viability was determined by trypan blue stain exclusion to be > 90-95 %.

# Measurement of Superoxide $(O_2)$ Production

 $O_2^-$  production was monitored using the method described by Cohen et al (1). In brief, adherant macrophages were washed with Hanks balanced salt solution (HBSS) then incubated in 2% fetal calf serum in RPMI 1640 with 5 to 20 ug lyophilized AEF sample for 4 hours. Dried AEF samples (described in chapter 2) consisted of solubilised AEF extract, protein electroeluted from the "AEF-active" region of multiple SDS-PAGE preparative gels, or a control, AEF-free preparation derived from the spleen of healthy mice. Following treatment with AEF, cells were washed, and incubated with fresh reactive media (0.02% glucose, 0.1% cytochrome C in HBSS) and PMA (phorbol myristate, Sigma, 0.5 ug/10<sup>6</sup>cells) for 60-90 minutes. The absorbance change at 550 nm was recorded from 1 ml aliquots of spent reactive media from each well. The protein content of each well was then determined following cell lysis with 0.5 N NaOH using the Bio-Rad protein assay kit (2). Results were recorded as nmoles O<sub>2</sub>-/mg protein/75 minutes.

## RESULTS

Results shown in Table 4.1 and figure 4.1 reveal that treatment with 5 or 10 ug of either our AEF extract or our AEF-free control sample had little effect on measurable  $O_2^-$  production by PMA-triggered macrophages. Notably, treatment of cells with similar doses of AEF protein eluted from the active region of our SDS preparative gels resulted in increases in  $O_2^-$  production as compared to control macrophages not treated prior to PMA triggering.

					1 1993 (1973)			
5 ug	84.59	(7.17)	89.15	(9.45)	100.70	(5.08)	107.63	(8.44)
	60.65	(9.16)	61.13	(13.91)	65.26	(9.58)	117.74	(6.74)
10 ug	215.10	(38.02)	258.76	(29.11)	260.87	(4.86)	228.04	(57.85)
	61.54	(25.06)	94.65	(22.45)	98.43	(7.96)	281.38	(108.46)
15 ug	370.68	(32.64)	293.79	(55.26)	172.37	(58.65)	568.09	(17.99)
	239.51	(17.82)	177.5	(31.54)	157.94	(13.20)	413.63	(80.59)
20 ug	101.73 79.59	(13.79) (40.00)	37.11 0	(11.32)	37.82 0	(2.36)	0 0	

# Table 4.1 Oxygen Radical Production by Peritoneal Macrophages Treated with either AEF Extract or Protein Eluted from the AEF-Active Region

Note: Mean values and standard deviations calculated from at least 3 assay samples. For descriptions of both control and AEF extract as well as protein eluted from the AEF active region refer to the Materials and Methods section of this chapter. Figure 4.1. Oxygen Radical Production by Peritoneal Macrophages Treated with either AEF Extract or Protein Eluted from the AEF-Active Region



Note: Fold increase/decrease values were obtained by comparing mean values calculated from at least 6 assay samples in each test group with untreated controls

For descriptions of both control and AEF extract as well as protein eluted from the AEF active region refer to the Materials and Methods section of this chapter.

With treatment doses of 15 ug, both the AEF extract and the AEF-free sample inhibited detectable  $O_2^-$  production. Cells treated with 15 ug eluted AEF active region protein however, again exhibited augmented  $O_2^-$  production.

When treated with 20 ug protein, cells in all groups produced little or no detectable  $O_2^{-}$ .

# DISCUSSION

Cells of the macrophage lineage are considered the most likely cellular candidate responsible for degradation of SAA preventing the accumulation of amyloid fibrils within normal tissue (3, 4). During sustained inflammation macrophages become "activated", resulting in enhancement or initiation of certain function including phagocytosis, bactericidal and respiratory burst activity, and antigen presentation (5-8). In the context of chronic inflammation, the activated macrophage is therefore relevant to the development of AA amyloidosis.

The processs of macrophage activation involves a regulated series of changes in physiology which culminate in the ability to execute complex enzymatic functions (5-9). The ability of a macrophage to perform certain specific functions, possibly including the degradation of SAA, may therefore be dependent on its activation state. In light of the fact that the appearance of endogenous AEF is directly correlated with the deposition of amyloid fibrils (10), we suspect that AEF may indeed affect the ability of the macrophage to be appropriately activated for the degradation of the amyloid precursor, ultimately leading to the deposition of amyloid fibrils.

Our previous work (Chp. 2, 11) has shown that spleen macrophages derived from animals treated with an exogenous source of AEF, in conjunction with an inflammatory stimulus, exhibit decreased reactive oxygen metabolite production in comparison to production by spleen macrophages derived from animals treated with the inflammatory stimulus alone. In addition, we found that when AEF was administered in the absence of AgNO<sub>3</sub>, (the inflammatory stimulus) no significant modification in metabolite release occured relative to steady state. The fact that the down-regulatory action of AEF is dependent upon a concomitant inflammatory stimuli suggests that AEF may act in concert with other factors induced or released during an inflammatory response or may exert an inhibitory effect only on macrophages rendered responsive through inflammatory stimuli. This would explain why AEF triggers AA fibril deposition only when administered in conjunction with an inflammatory stimulus. By exposing proteose-peptone elicited macrophages to AEF *ex vivo* we hoped to determine whether AEF-induced-modulation of  $O_2$  production is dependent only upon the presence of AEF and the inflammation-induced macrophage.

Our observation of significantly decreased  $O_2^-$  production in the presence of higher doses of our AEF samples, including the control AEF-free preparation, suggests that inhibition of metabolite production observed at 20 ug and 15 ug may be caused by toxicity of the samples to macrophages. Trypan blue exclusion studies performed post-incubation with treatment media revealed only a slight decrease in viability at these concentrations (data not shown), however the ability of the macrophages to produce  $O_2^-$  may have been impaired.

Furthermore, it is possible that the observed stimulation of reactive oxygen metabolite production in the presence of protein prepared from the "AEF-active" region of SDS preparative gels is due to SDS contamination. This is suggested by the fact that treatment with non-toxic doses of simple AEF extract, which contains active AEF but has not undergone final purification procedures involving SDS-preparative gels, does not result in enhanced  $O_2$  production. The triggering of the respiratory burst in macrophages involves a series of molecular events which take place at the level of the plasma membrane (12). Macrophages are known to respond to a variety of stimuli including particles subject to phagocytosis and soluble agents capable of interacting with the cell membrane (13) **SDS** is an ionic detergent that can denature and solubilize membrane proteins when present at high concentrations. It is conceivable then that contaminant SDS may affect the conformation of cell surface proteins involved in O2<sup>-</sup> production thereby triggering the respiratory burst. Despite extensive dialysis of samples following elution in CAPSmethanol buffer, it is probable that a small quantity of SDS remained. Using acetone:methanol precipitation to recover protein following electroelution eliminates contaminatant SDS from the protein quite effectively (14). This procedure however, further denatures the already extensively processed AEF preparation and was deemed innappropriate for the preparation of a biologically active sample. A possibility for further investigations into this question would be to conduct the O2<sup>-</sup> production assay on cells treated with protein electroeluted from the region of SDS gels of control, AEF-free sample corresponding to the "AEF-active" region of AEF sample-run gels. Preparation of such a control sample is greatly hindered by the fact that relatively little protein is contained within the appropriate region of the preparative gel. Inordinate amounts of healthy murine spleen are therefore required as starting material to produce the necessary microgram quantities of electroeluted, AEF-free end sample.

It is possible that the absence of AEF-associated down-regulation in  $O_2^-$  production observed in the experiments presented here is due to the fact that peritoneal macrophages

(versus spleen macrophages examined in chapter 2) were used. The ability of macrophages to produce oxygen metabolites can vary according to their site of isolation and extent of differentiation (15). Modifications within the oxygen metabolite production pathway occuring during site-specific macrophage differentiation could conceivably alter cellular susceptibility to the effects of AEF. Studies have shown however that coincubation of amyloidotic peritoneal macrophages and SAA results in amyloid fibril formation (16). We have therefore assumed that peritoneal macrophage function is in fact vulnerable to AEF-induced modulations associated with amyloid deposition.

Overall, data obtained during these experiments does not resemble data obtained with respect to the *in vivo* model of macrophage reactive oxygen metabolite production during AEF-induced amyloidogenesis described in chapter 2. This may indicate that elements other than the inflammation-associated macrophage and AEF are required to incur dampening of cellular  $O_2^-$  production or that active AEF is not present in sufficient concentration in our treatment preparations. In future, the complete characterization of AEF, allowing for its purification to homogeneity, will eliminate the possibility of intereference by unknown components included in current AEF preparations. Further oppurtunity for study lies in the investigation of the effects of adding specific factors known to be induced or released during an inflammatory response (i.e. acute phase proteins such as SAA or cytokines such as the interleukins 1 and 6 or tumor necrosis factor) to our *in vitro* assay system. These types of experiments may well lead to the elucidation of the elements involved in, as well as the mechanism characteristic of AEFinduced modulation of macrophage function.

82

## REFERENCES

(1) Cohen, H.J., Chovaniec, M.E. 1978. Superoxide generation by digitonin-stimulated guinea pig granulocytes. A basis for a continuous assay for monitoring superoxide production and for the study of the activation of the generating system J Clin Invest. 61:1081.

(2) Bradford, M. 1976. A rapid and sensitive method for the quantitation of microgram quantities of protein utilizing the principle of protein dye binding. *Anal Biochem* 72 248.
(3) Lavie, G., Zucker-Franklin, D., Franklin, E.C. 1980. Elastase-type proteases on the surface of human blood monocytes: possible role in amyloid formation. *J Immunol* 125:175.

(4) Fuks, A., Zucker-Franklin, D. 1985. Impaired Kuppfer cell function preceeds development of secondary amyloidosis. *J Exp Med* 161:1013.

(5) Berche, P., Gaillard, J-L., Sansonetti, P.J. 1987. Intracellular growth of *Listeria monocytogenes* as a prerequisite for *in vivo* induction of T-cell mediated immunity *J Immunol.* 138:2266.

(6) Babior, B.M. 1984. Oxidants from phagocytes. agents of defense and destruction. *Blood* 64:959.

(7) Nathan, C.F. 1983. Secretory products of macrophages J. Clin Invest 79.319
(8) Ezekowitz, R.A.B., Austyn, J., Stahl, P.D., et al. 1982. Surface properties of the Bacillus-Calmette-Guerin-activated mouse macrophage. Reduced expression of mannose specific endocytosis, Fc receptors, and antigen F4/80 accompanies induction of Ia J. Exp. Med. 154:60.

(9) Adams, D.O., Hamilton, T.O. 1987 Molecular transduction mechanisms by which IFN and other signals regulate macrophage development *Immunol Rev* 97.5.
(10). Axelrad, M., Kisilevsky, R. 1979. Biological characterization of amyloid enhancing factor. In: <u>Amyloid and Amyloidosis</u>. Proc. 3rd. Symposium on Amyloidosis (Glenner, G.G. Costa, P.P. deFreitas, F.F., Eds.) p. 527 Excerpta Med. Amsterdam.

(11) Reid, C., Hébert, L., Pozzulo, G, *et al.* 1993 Splenic macrophage activation and functions in amyloid enhancing factor-induced secondary amyloidosis Study of phagocytosis, killing, respiratory burst, and MHC II surface expression *1 Leuk Biol* 53 in press.

(12) Rossi, F., Bellavite, P., Berton, G., Grzeskowiak, M., *et al.* 1985. Mechanisms of production of toxic oxygen radicals by granulocytes and macrophages and their function in the inflammatory process. *Path. Res. Pract.* 180:136.

(13) Pick, E., Keisari, Y. 1981. Superoxide anion production and hydrogen peroxide production by chemically elicited peritoneal macrophages. Induction by multiple nonphagocytic stimuli. *Cell. Immunol* 59.301.

(14) Hunkapiller, M.W., Lujan, E. 1986. Purification of microgram quantities of proteins by polyacrylamide gel electrophoresis. In <u>Methods of Protein Microcharacterization</u>.
(Shively, J. Ed.) p. 89. Humana: Clifton

(15) Fantone, J.C., Ward, P.A. 1982. The role of oxygen-derived free radicals and metabolites in leukocyte-dependent inflammatory reactions. *Amer. J. Pathol.* 107, 397.
(16) Miura, K., Baba, S., Shirasawa, H., *et al.* 1992. Intraperitoneal amyloid formation by amyloid enhancing factor-rich macrophages in ascites fluid. *Virch Archiv. B Cell Pathol.* 62:245.

CHAPTER 5 GENERAL DISCUSSION AND CONCLUSIONS

,

## DISCUSSION

Amyloidosis describes a heterogenous collection of diseases characterized by the extracellular deposition of proteinaceous amyloid fibrils of great chemical diversity. Nevertheless, the shared tinctorial and structural characteristics of all amyloid deposits suggest that amyloidosis occurs due to a single, common pathophysiological process. The consistent detection of AEF in a variety of amyloidotic tissue and its apparent causative role in amyloidogenesis suggests that AEF is an integral component of this common pathogenic process (1–5). Physiological processing of precursor proteins of various amyloids by activated macrophages, RE cells, or microglia has also been consistently implicated in amyloidogenesis (4, 6-9).

Our *in vivo* studies focusing on the relationship between AEF and reticuloendothelial cells found at the site of amyloid deposition have shown that AEF selectively affects some macrophage activation-associated functions. Examination of functions known to be enhanced in the activated macrophage revealed that while AEF does not affect spleen macrophage phagocytic or microbicidal activity whether administered alone or in conjunction with an inflammatory stimulus, it does dampen reactive oxygen metabolite production as measured by O<sub>2</sub><sup>-</sup> production. When AEI<sup>+</sup> was administered in the absence of an inflammatory stimulus no lignificant modulation in metabolite release was observed. The apparent dependence of AEF activity on a concominant inflammatory stimulus suggests that AEF either acts in concert with factors induced or released during an inflammatory response, or exerts an inhibitory effect only on macrophages rendered responsive through inflammatory stimuli. This would explain why AEF triggers AA fibril deposition only when administered in conjunction with an inflammatory simulus.

The acute phase state, characterized by the presence of particular mediators (i.e. IL-6), has in fact been associated with amyloidotic tissue environments in non-AA amyloidosis (10, 11). Future investigations elucidating the specific aspects of the APR necessary for amyloid fibril formation in the presence of AEF may reveal oppurtune areas for therapeutic intervention in amyloidogenesis, for example through the use of antiinflammatory medications.

Further studies examining the effect of AEF on Ia surface expression (classically used as a marker of RE cell activation status [12-14]) revealed that increased la expression indicating inflammation-associated macrophage activation is blocked in the presence of AEF. As indicated by our respiratory burst studies, AEF seems to exert an inhibitory effect only in the event of a concomitant inflammatory stimulus. Taking these investigations a step further we then examined whether the observed suppression in macrophage Ia

expression played an integral part in altered SAA degradation leading to AA fibril deposition in the splcen of experimental animals. Our results show that restoration of surface Ia expression with IFN-gamma did not prevent production and deposition of AA fibrils in the spleen indicating that the mechanisms responsible for MHC antigen expression, though affected by AEF, are not directly involved in amyloidogenesis.

Our studies examining the hitherto elusive biochemical nature of AEF are on the brink of realization. Thus far SDS-PAGE analysis of AA amyloidotic spleen extracts coupled with concomitant AEF activity assays have pinpointed AEF activity to occur within a molecular weight range between 63.2 kDa and 47.9 kDa consisting of three distinct bands. Significantly, a single protein band with an Mr of 54.6 kDa was detected in AEF-active samples but not in AEF-free control preparations suggesting it as a possible AEF candidate.

2-D electrophoretic resolution of the region containing AEF activity revealed 13 distinct polypeptide spots suggesting that the three protein bands detected by in one dimensional SDS-PAGE analysis are comprised of greater than one protein species and that degradation events producing consistent polypeptide products has occurred. Three spots in particular have been identified to occur in active AEF samples but not AEF-free controls. Sequence characterization of these 3 spots, regardless of whether they represent whole or partial protein species, will be used to search for protein identity (by comparison with available sequences stored in databanks) as well as for preparing specific DNA probes for cloning as yet uncharacterized proteins. Both human and mouse cDNA libraries constructed from mRNA isolated from tissue known to produce AEF (i.e. the liver) are currently available. Obtaining cDNA clones corresponding to cellular transcripts of the AEF gene will be useful for obtaining the complete amino acid sequence of the protein, predicting specific structural or functional features of the protein by computer analysis. functional analysis of the protein in transfection experiments and site-directed mutagenesis, and elucidation of the control of the expression of this gene in normal and pathological situations.

Recent studies in our lab have led to the successful characterization of AEF isolated from the brain tissue of patients with Alzheimer's disease (unpublished results) and found it to correspond to myelin basic protein (MBP). Though our AA amyloid-associated preparation did not react with anti-MBP antibodies (data not shown), any similarities in sequence and/or function between MBP and AA amyloid-associated AEF (once characterized) will be highly indicative of specific features which attribute AEF activity to a protein. The synthesis and action of AEF is a key element in the pathogenesis of amyloidosis. Elevated AEF levels have been shown to necessarily accompany amyloid deposition in the murine model of AA amyloidosis (3) Determining both the mode of action of AEF as well as its biochemical character are essential to the elucidation of the amyloidogenic process common to all forms of amyloidosis. Once these objectives have been achieved it will be possible to investigate ways of inhibiting AEF synthesis or its action in triggering the process of fibril formation.

## REFERENCES

(1). Shirahama, E S., Miura, K , Ju, S-T., *et al.* 1990. Amyloid enhancing factor-loaded macrophages in amyloid fibril formation. *Lab Invest.*; 62<sup>,</sup> 61.

(2). Axelrad, M., Kisilevsky, R. 1979. Biological characterization of amyloid enhancing factor. In: <u>Amyloid and Amyloidosis</u> Proc. 3rd. Symposium on Amyloidosis (Glenner, G.G. Costa, P.P. deFreitas, F.F., Eds.) p. 527. Excerpta Med.: Amsterdam.

(3) Axelrad, M., Kisilevsky, R., Willmer, J., *et al.* 1982. Further characterization of amyloid enhancing factor. *Lab Invest* ; 47: 139.

(4) Varga, J., Flinn, M S A., Shirahama, T, *et al.* 1986 The induction of accelerated murine amyloid with human spleen extract *Virchows Arch B Cell Pathol.* 51:177.

(5). Alı-Khan, Z, Quirion, R., Robitaille, Y., *et al.* 1988. Evidence for increased amyloid enhancing factor activity in Alzheimer brain extract. Probable role of amyloid enhancing factor. *Acta Neuropathol.*, 77: 82.

(6) Lavie, G., Zucker-Franklin, D., Franklin, E.C. 1980. Elastase-type proteases on the surface of human blood monocytes. possible role in amyloid formation. *J. Immunol.* 125:175.

(7) Skogen, B, Thorsteinsson, L., Natvig, J B. 1980. Degradation of protein SAA to an AA-like fragment by enzymes of the monocytic origin. *Scand. J. Immunol* 11:533.

(8) Fuks, A, Zucker-Franklin, D 1985 Impaired Kupffer cell function precedes development of secondary amyloidosis. *J Exp Med.* 161:1013.

(9) Wisniewski, H.M., Vorbrodt, A.W., Wegiel, J., *et al.* 1990. Ultrastructure of the cells forming amyloid fibrils in Alzheimer disease and scrapie. *Am. J. Med. Genet. (Suppl)* 7:287.

(10) Bauer, J., Strauss, S., Schreter-Gasser, U., *et al.* 1991. Interleukin-6 and alpha-2macroglobulin indicate an acute phase state in Alzheimer's disease cortices. *FEBS Lett.* 285:111

(11) Vandenabeele, P., Fiers, W. 1991. Is amyloidogenesis during Alzheimer's disease due to an IL-1/IL-6-mediated 'acute phase response' in the brain? *Immunol. Today* 12:217.
(12) Adams, D.O., Hamilton, T.O. 1987. Molecular transductional mechanisms by which IFN and other signals regulate macrophage development. *Immunol. Rev.* 97:5.

(13) Lafuse, W.P. 1991. Molecular biology of murine MHC Class II genes. Crit. Rev. Immunol. 11 (3, 4):167.

(14) Nibbering, P.H., Langermans, J.A.M., Van de Gevel, J.S., Van der Hulst, M.B., Van Furth, R. 1991. Nitrite production by activated murine macrophages correlates with

their toxople+ mastatic activity, Ia antigen expression, and production of H2O2. *Immunobiol.* 184(1):93.

## **CLAIMS TO ORIGINAL RESEARCH**

1. I have established that AEF selectively affects certain activation-associated macrophage functions. Our *in vivo* studies show that while AEF, in the presence or absence of silver nitrate, has no apparent effect on the ability of spleen and liver macrophages to phagocytose or kill *L* monocytogenes, it blocks enhanced respiratory burst function (as measured by  $O_2^-$  production).

I have demonstrated that treatment with AEF in conjunction with an inflammatory stimulus (AgNO<sub>3</sub>) blocks the increase in number of splenic macrophages expressing Ia seen with the inflammatory stimulus alone. Furthermore, treatment with IFN-gamma was shown to restore decreased Ia expression in animals given AEF and AgNO<sub>3</sub>. I have also shown that this restoration of la expression did not prevent amyloid A fibril deposition.
 I have identified three candidate proteins (or polypeptide derivatives of these proteins, present in AEF-active preparations only, which may be responsible for AEF activity. I have presented these proteins on 2-D gels easily amenable to future microsequencing via transfer to polyvinylidene difluoride membranes.

91

# ACKNOWLEDGMENTS

I would like to extend my appreciation and thanks, first and foremost, to my supervisor, Dr. Francine Gervais, for providing me with the oppurtunity to work and learn in her laboratory.

I sincerely thank Dr. C. Pietrangelli for her citical reading of the manuscript presented in Appendix I. Her advice and support has proved invaluable to me.

I would like to acknowledge Emilie Strmen, Janet Etherington, and Ping Wang for their technical assistance in my work. I would like to give special thanks to Eric Leblond for his work on 2-D analysis.

Finally, I could not finish without thanking my sister Lucy, without whom this work would never have reached the printed page.

# **APPENDIX** 1

Manuscript of Paper Entitled:

Splenic Macrophage Activation and Functions In Amyloid Enhancing Factor-Induced

Secondary Amyloidosis

Study of Phagocytosis, Killing, Respiratory Burst, and MHC II Surface Expression

# Splenic macrophage activation and functions in amyloid enhancing factor-induced secondary amyloidosis Study of phagocytosis, killing, respiratory burst, and MHC class II surface expression

Caroline Reid, Lise Hebert, Gina Pozzulo, and Francine Gervais

. Methodown for the Stale of Hold  $R_{\rm est}$  is a construction of Hold  $= R_{\rm est}$  is the Monde O  $_{
m est}$  is  $\epsilon_{
m est}$  .

Abstract: Secondary amyloidosis is a systemic disease characterized by the extracellular tissue deposition of insoluble fibrillar amyloid A protein Aberrant metabolism of serum amyloid A protein by reticuloendothelial cells is thought to result in the accumulation of fibrils within the tissue. Treatment of mice with amyloid-enhancing factor (AEF) in conjunction with an inflammatory stimulus (i.e., AgNO<sub>3</sub>) induced amyloid deposition within 48-72 h. The activation state of a macrophage largely defines its

vymatic capabilities. In the studies reported here, we examined the effect of AEF on spleen macrophage activation using both functional and phenotypic assays. We found that while AEF in the presence or absence of AgNO<sub>3</sub> has no apparent effect on the ability of spleen and liver macrophages to phagocytose or kill Listeria monocytogenes, it appears to block enhanced respiratory burst function (as measured by O2<sup>-</sup> production) observed with AgNO3 alone AEF therefore seems capable of inhibiting certain macrophage activation-associated functions while not affecting others. Our activation phenotype studies, using surface la expression, reveal that AEF blocks the increase in number of splenic macrophages expressing Ia seen with AgNO3 alone Treatment with interferon- $\gamma$  was found to restore decreased Ia expression in animals given AEF + AgNO3 but did not prevent amyloid A fibril deposition. J. Leukoc. Biol. 53. 651-657, 1993.

Key Words. reticuloendothelial cells · Ia · bactericidal action • reactive oxygen metabolites • FACS analysis

## INTRODUCTION

Amyloidosis is a generic term for a diverse group of diseases characterized by the extracellular tissue deposition of insoluble protein fibrils, often resulting in failure of the involved organs. Secondary amyloidosis is typified by the deposition of amvloid A protein (AA) derived from the carboxylterminal cleavage of its acute-phase serum precursor, serum amyloid A (SAA) [1, 2]. Secondary amyloidosis appears in conjunction with certain malignancies, chronic inflammatory disorders, or prolonged infection [3]. Experimental secondary amyloidosis is induced in laboratory animals by sustained inflammation or chronic bacterial or parasitic infections [4].

Amyloid-enhancing factor (AEF) is believed to play an integral role in the pathogenesis of amyloidosis. Although its complete biochemical characterization remains elusive, AEI is routinely extracted from the tissue of amyloidotic animals,

where its activity is evident approximately 21-43 h before? the histological detection of any lord fibral [9]. The fulling of characteristic of AFE is at ability to needer ite AA fibral deposition. Although most experimental induction protocolrequire everal week for the development of detectable anyloid deposits, idministration of AEE in communication with in inflammator standar really in fibril deposition within 24 to 72 h [6]. The ability of an exocenous come of XEE to shorten the induction period required for une-loid deposit tion suggests that it icts by precluding the lag time required for the endogenous production of AFT [ ]. Furthermore genetic analysis of involoid actituat and an ceptible trainof mice suggest, that asceptibility to the levelopment of epermitental secondary anyloidosi. is controlled by a mole dominant gene [3]. This observations in conjunction with the "induce that is a fame an A.J. mice can be overcome by an travenous (iv) injection of in exogenoir once of MT [9] further supports the role of AEL as a major determinant in the development of secondary anyloidosis.

Renculoendothelial cells have long been recognized in dose association with anyloid deposit in iter [10]. Some if peripheral blood monocytes are known to be expible of mediating SXX degradation in vitro through the action of cell surface enzymes [IE-12]. Studies, unred at chieidating the exact relationship between macrophages or monocytes and the deposition of anyloid fibril, have led to the discover of a mononuclear leukocytes from healthy individuals accupable of mediating complete SAA degradation, where is the cells of amyloidotic patients produce an intermediate protein fragment, both the sarae size as AA and untraunologic dly cross-reactive with XA [11]. In addition, normal marine Kupfler cells have been found to degrade SXX completely. whereas Kupfler cells from anyloid induced animal produce the AA intermediate [13]. This observation has led to the hypothe is that tissue imploid deposition in , be due to inappropriate or incomplete SAA degradation by cell of the monocyte/macrophage lineage, resulting in the second lation of AA

In light of the obsious importance of both ALE and reticuloendoth chalcells in the formation of amylor l deposit.

Abbreviations XX include X protein XFF include the include the second stations XX include X protein XFF induced in the BSX boxine sector in domain  $H \to \gamma$  interferon  $\gamma$  by an unconductive G. MHC major histocompatibility complex PBS photphate buffered drive PMA photbol inversities active  $\beta AA$  error and ord X. PF predict pathogen free

Reprint requests Francine Gerzus Montre il General Hospital Research Institute FH 11, Rm 513, 9540 Cedar & C. Montre il Oucher HBG 1784 Received January 27, 1993, accepted March 3, 1993.

racional caracteria meltion the production of cadomining AFE could cause ibertant SA's degradation by macrophage. Circumstantial condence apporting our hypothes a provided by the result of presion studies which hoved that the appearance of ALL cas correlated and be the enne of theil accumulation in two [5]. Furremore premain , ork in our liborator detected a de ser en plene merophig, urbe major histocomputbalic sample (MHC) day H(Li) e pression that conceded with the open inner of endownour. All during induction of e penneral condar andodo i hi edironi inflamma tion protocol (unpubliched reality). All could interfere with maciophase demandati e proce e clashlocking e ramon re along 5 mahilit of the marophage to degrade the SAA completel. We therefore e-immed macrophage activition on a functional le dan sell as on a phenotypic level follow nee the induction of coordary analoidosis using MT in combinations (the her nitrate (A (NOs) as an inflammators annub.

Corran morrophage producacity dues such as phagoesto-E, bactericidal function - ad respiratory burst functions are enhanced during any ation [11/15] and are therefore good functional indicators of macrophage activation status. Our e perment compare the in seo, ability of liver and spleen cell to physicitose and full Insteria monocytogenes as well as the invitio ordere but tasponse (is indicated by O.7 production) of splenic macrophages between anyloidotic minutes treated with AgNO<sub>6</sub> in conjunction with AEE and animal, treated with AgNO<sub>4</sub> alone. In addition, we examined the differential expression of Ia antigen by macrophages rolated from the apleers of mice treated with AFE. and AgNO<sub>3</sub> by using flow extoractiv and immunofluorescence malysis. The enhanced expression of MHC class II molecules may be used as a marker of macrophage activation [17, 18] Ora ann was to determine whether treatment with MEE and alver nutrate causes a change in macrophage surface la expression compared with treatment with the inflammatory stimulus alone and it so whether any such change could reflect an alteration in the degradation pattern of the SAA precursor leading to AA anyloid fibral deposition.

## MATERIALS AND METHODS

### Animals

Specific pathogen-tree (SPF) C57Bl/6NHsd male mice, 8–12 weeks old, were either obtained from Harlan Sprague Dawley (NHsd, Indianapolis, IN) or bred in our own SPF facilities Food and water were provided ad libitum. Animals were sacrificed by CO<sub>2</sub> narcosis followed by cervical dislocation.

# Preparation of amyloid-enhancing factor and induction of amyloidosis

AFF was prepared from the spleers of C57BL/6 mice, injected subcutaneously (s.c.) for 21 days with 13% casein (United States Biochemical, Cleveland, OH) AEF was estracted from homogenized spleens in we-cold 30% glycerol-10 mM firs HCl, pH 7.5, buffer as previously described [6]. A 0.5-rig portion of the lyophilized AEF product (an amount previously determined to induce significant amyloid deposition within 24-48 h [9]) was resuspended in 0.2 ml of sterile phosphate buffered saline (PBS) and administered via tv. injection concommant with 0.5 ml s.c. of 2% AgNO<sub>3</sub> solution

#### Detection of secondary amyloid deposits

At least a portion of each spleen was fixed in buffered 10% ferro dimembedded in partitin, sheed in 5-µm sections, and then standed with alkaline Congo Red [19]. The presence of am loid deposits was determined microscopically by the derection of zero n birch ingence characteristic of AA fibrils under polarized light.

#### In vivo phagocytosis and bactericidal activity

Three days after the infection of MT flore  $NgNO_3$  alone, or both MT and  $NgNO_3$  experimental animals were inoculated with  $3^{-100}$  colony-forming units of *T\_mono\_stogenes*. The livers and spleens of infected animals were tested for their ability to phagoextose and kill *T\_mono\_stogenes* by determining the number of viable bacteria in individual organs 30 min or 6 has respectively after infection [14]. Spleens and livers were removed aseptically and homogenized in sterile solute for about 1 min with a Tri-R Stir R homogenizer (Tri-R\_instruments\_Rockville\_Centre\_NY). Serial 10-fold dilutions of these homogenizes were plated on tryptose agar (1% Bacto-agai. Diteo: Laboratories\_Detroit\_NH) and colony counts were routinely executed after 18-24-h incubations at 37. C\_Five animals per group were tested.

### Spleen macrophage O<sub>2</sub><sup>-</sup> production

Responses were examined in splenic macrophages from naive mice as well as mice treated with ALF alone,  $AgNO_3$ alone, or ALF and  $AgNO_3$  days previously. Internal experimental positive controls consisted of proteose peptone (PP)-elicited peritoneal macrophages.

Spleen cells were flushed from isolated spleens with icecold RPMI 1640 supplemented with 0.3 mg/ml glutamine and 50.0 ng/ml gentamicin. Red blood cells were lysed by hypetonic shock on ice. Cell differentiation was determined on cytospin slides (Shandon, Sewickley, PA) stained with Giemsa (DiffQuik, Canlab, Montreal, Quebec). Each cell suspension was adjusted to  $1.5 \times 10^{6}$  macrophages/ml in RPMI 1640. + 10% fetal calf serum. Macrophages were plated in 6- or 24-flat bottom well plates (Linbro, Flow Laboratories, McLean, VA) and purified by adherence overnight at 37°C, 5% CO<sub>2</sub>. Each cell suspension was assaved in triplicate. In all instances, cell viability as established by trypan blue exclusion test was >90–95%. The cell monolayers were >90% macrophages as determined morphologically by Giemsa stain and phagocytosis of *L. monocytogenes*.

Production of  $O_2^-$  was measured following treatment with phorbol myristate acetate (PMA,  $0.5 \mu g/10^6$  cells) for 75 min by the method of Cohen and Chovaneic [20]. The protein content of each well was then determined following cell lysis with 0.5 N NaOH using the Bio-Rad protein assay kit [21]. Results are reported as nmol  $O_2^{-7}$ /mg protein/75 min  $\pm$ SEM. Statistical analysis was done by a two-tailed Mann-Whitney U test with a present probability level of  $\alpha = 0.05$ . Mean values represent data from three experiments.

### la surface expression

After aseptic removal, each spleen was flushed with ice-cold RPMI 1640 supplemented with glutamine and gentamicin

Constraint a configuration for the second second

vished by centrifugation (4.C) and resuspended in 10 milot ce-cold RPMI 4640. Enumeration of cells and determinition of viability (by stain exclusion) were accomplished using a hemoeytometer and Turks or trypan blue stam respectively. Slides for cell differential counts were prepared is previously described. Following resuspension at a final concentration of 1-2 × 10° cells ml with 0.5% beying scrum al burnin (BSA) in PBS (1 = 10° cells were objusted as suit polystyrene round bottom tubes (Bector) Dickinson Libware Montreal) for minimunofluore cent labeling. After their bation with goat immunoglobulin GalgG-Sigma-St. Lone. MO) to reduce nonspecific hinding by blocking hereceptors. alls were labeled successively with monoclenil intimume Mac-1 antibody (clone M170) Bochringer Mainheim Br ochemical Laval Oucbee) phycocrythrin conjugated coar F(ab), antirut lgt, it altrig Laboratories. Sim Li meiscov and fluorescem isothiocyanite conjugated murine intelle-IgG (Cedarlanc Laboratories). Each incubation with label ing antibody (at 1.20 dilution is determined optimal by serial titration) was carried out for 15 min on ice. The qu propriate control labelings were prepared to correct for autofluorescence nonspecific binding, and inhibition of binding due to steric hindrance for each experiment group. abeled cells were fixed in ice-cold paratormaldehyde and ept at 4°C until analyzed on a EVCScan flow extometer with the Lysis II Program (Becton Dickinson) the following day When indicated, interferon  $\gamma$  (IFN- $\gamma$ ) treatment consisted of 5000 units of rat recombinant II N-7 (Amgen **Thousand** Oaks, CA) administered (p) on divs 0 and 2 of the experimental design

## RESULTS

# In vivo phagocytosis and bactericidal activity during AEF-induced secondary amyloidosis

To determine whether the ability of splenic and hepatic mac rophages to phagocytose and kill L monocytogenes is modified during AA fibril deposition, groups of mice were injected muc injected with ALE or AoNO mone

Phagoeversti, ov spicen and two cells was merimeet so minuteer the result ection of B = 0,  $z = z_{\rm ect}$ , z = 0. By a velopment of any subsequences with monified bases of aptike compared with many namely or animals or animals treated with AH or XeNO, done (Fig. 1). Approximately 95 — of the bacterial incombine way take imply the placen and fixed is the first 30 para there and so

The base modulate proty of manophases can a model that inter the nanomonal  $2^{-1}$  of wheta accumulation modulated on the layer of ATH ANO, treated much despite unifold deposition after 5 day compared with neurominals or much  $2^{-1}$  or leavish ATH or A NO, done that D. All groups were shown to be expublic of elemine approximatch  $3^{-1}$  90 set the bacterial load within the fit  $1^{-2}$  following interferent No, is informational discrivity value of served during the lance time period in the placen of both control and experimental much d

### O<sub>2</sub> production in AEF induced secondally amyloidosis.

To determine whether the reprintery barst function of splena interophases is directed during XX fieral leposition due to XEE O = production by cells rolated from any days direction through the VEE O = or eather user taking was measured (Table 1). Splenic interophase from mission period with X(NO), done canon unsloted genic treatment) released a terminent. There concentration of O = metabolise (following) tanulation in vitro with PMA within the cool trained from the unite ited control (roup. Remark bbl) when animals were treated with both XEE and XCNO), the dolir of their spleen macrophase to release O = set (and their spleen macrophase) to release O = set (and their spleen macrophase) to release O = set (and their spleen macrophase) to release O = set (and their spleen macrophase) to release O = set (and their spleen macrophase) to release O = set (and their spleen macrophase) to release O = set (b) and the AgNO3 treatment group. These with the original that in an amyloid-inducing treatment. XEE partially, block (XeNO) induced O, production by plean macrophase.

This pattern of partial inhibition, it is occurred at the level of H<sub>2</sub>O<sub>5</sub> production. Again, the auxilordogenic recutment



Fig. 1 In vivo phagocytosis and bactericidal activity measured in the her of mice  $\alpha$  and  $\beta + \beta$  and  $\beta + \beta$  compared with mice treated with M F and/or  $AgNO_3\beta$  days before being interval with  $\beta \neq \beta^*$  *I* monocytogene. The in  $\alpha$  to phagocyte itality  $\beta$  the liver cells was determined 30 min after infection and the in vivo bictericidal activity, we say measure 6 h bits  $\beta$ , described in Material and Method. Normal mice displayed phagocyte and bactericidal activities similar to those observed in mice treated with M F and/or  $Ag^*O_3$ .

LABLE F	()	Production 5. Macrophages in ALI-Induced		
Secondar, Am doide as				

to constant of the	O production' minorany procession of minor (SEM)
- antrol	92 17 + 9 20
LE dota	$110.52 \times 17.60$
5 .O none	5116 July - 12 14
All and Synch.	217 83 -> 32 29
· · ·	

(1) production was measured according to the method described by Cotion and Costancie [20]. Results obtained represent means (+ SEM or three (+ permittate) (+ e) from an triplicate).

arche unit, les (a = 0 i) than the obtained with NeNO (done

as accompanied by a decrease in the AgNO<sub>3</sub>-induced release of the relevant metabolite. Data obtained for  $H_2O_2$  release however, did not reach significance.

# In surface expression during AEF-induced secondary amyloidosis

The injection of a combination of ALL and  $NgNO_3$  (as an inflaminatory stimulus) is known to induce anyloid fibril deposition in mice within 24-48 h [5]. AgNO3 alone was injected into mice in a parallel group to provide a basis of compartion of the level of activation of splenic macrophages harvested from animals subjected to an inflammatory stimulus in the absence of AEF. Using a double immunofluorescence protocol, we determined the level of surface Ia antigen expression on Mac-I\* spleen cells by flow cytometry, as a means of monitoring the activation state of the macrophages, 3 days after treatment of mice in vivo. Results presented in Figure 2 reveal that maximal surface expression of Ia antigens by splenic macrophages occurred in response to the injection of AgNO3 alone, which does not produce amyloid deposits in the tissues. When this treatment was combined with an injection of anyloid-inducing AEF, however, there was complete abrogation of the augmented macrophage Ia expression. Control preparations of AEF-free spleen extracts



Fig.3 Restoration of splern macrophage Erexpression in mice treated with NLI and AgNO<sub>3</sub> by an vivo  $HN\gamma$  treatment. Splern cells isolated from mice 3 days after in vivo treatments indicated in the figure (described in Materials and Methods) were prepared and analyzed as described in the legend of Figure 2. One to three groups (of at least three mice rach) were rested for each treatment.

did not interfere with the inflammation-induced Ia expression, suggesting that the observed down-regulation in surface Ia is a specific effect of AEI AEF treatment without concomitant AgNO<sub>3</sub> treatment caused a slight but insignificant increase in Ia expression compared with untreated animals (Fig. 2).

Having established that AEF may indeed down-regulate the normal macrophage response to activating agents, at least at the Ia level, we examined the possibility that the observed down-regulation may be related to a change in the degradation pattern of SAA leading to AA amyloid fibril deposition. By combining an injection of IFN- $\gamma$  (in a dose previously determined to be sufficient to induce increased Ia expression by splenic macrophages in C57BI/6 mice, unpublished results) with our AEF-AgNO<sub>3</sub> treatment protocol, we



Fig. 2. Modulation of spleen macrophage Ia expression from normal steady state in mice-rendered annioidotic by treatment with AEF and AgNO<sub>3</sub> computed with mice-treated with AgNO<sub>3</sub> or AEF alone, or a control, AEF-free spleen extract and AgNO<sub>3</sub> Spleen cells were isolated from mice 3 days after the mile-treatments. Cells were labeled with Ia and Mac-1 specific antibodies, as described in Materials and Methods, prior to analysis by flow extometry. This double antibody label system distinguishes macrophages expressing Ia antigen on their surface (Mac-1'Ia') from B cells (Mac-1'Ia') and surface Is negative macrophages (Mac-1'Ia'). Granulocytes and natural killer cells, which express only small quantities of Mac-1 antigen, are weakly Mac-1' relative to macrophages [22] and therefore do not stain intensity enough to be included in Mac-1' determinations. Scatter gates were set to exclude dead cells and debtis. Results are expressed as a percentage of Mac-1' cells that are Ia'. As indicated on the casis, analysis of spleen cells from untreated, control mice text fled that Ib 5% of Mac-1' spleen cells are Ia' under normal steady state (NSS) conditions. One to three groups (of at least three mice each) were tested for cash treatment.



Fig. 4 Detection of amyloid deposition by Congo Red stain. Amyloid fibril deposits within splice actions obtained from an  $x(1 + xg^{S}C)$ , treated minial exhibit characteristic green birefringence under polarized light (X) is hen stained with Congo Red. Splice actions from normal minial do not exhibit birefringence under polarized light (C) (B and D) Congo Red-stained splice sections from amyloidotic industriated mice respectively a observed under a normal microscope. Splicen sections were prepared as described in Materials and Methods.
" anarat e cap	+n loid dep e view
STE alore	0,3
	12/-1
FE and the Ca	979
Council All to contract and AshOr	() / 3
	47.3
	() f)
$A = X_{i} A = A = A = A = A = A = A = A = A = A $	+1/+2

Account groups is described in Materials and Methods

action are cancel with Conco Red as described in Materials and demody. Ym acach buildepolation was detected by enaracteristic witch or extra sense under polarized fight.

. creable to restore la expression to level seen with AgNO3 alone (Fig. 3). Histological examination of portions of spleen retained from mice treated with all three agents, however reveal that despite the IFN- $\gamma$ -induced restoration of macrophage Leespre sion in mice administered ALF and AgNO3, pleme anyloid fibril deposition sull occurred (Fig. 4a, Table 2). The, finding suggests that AEF induced down-regulation of spleme macrophage activation observed at the level of Ia expression is not related to aberrant SAA degradation leading to AA anyloid fibril deposition

## DISCUSSION

Based on studies reported in the literature [11, 13], we postulate that the degradation of amyloid precursors by maciophages is a critical event that prevents the deposition of anyloid fibrils in normal tissues. The process of macrophage activation involves a regulated series of changes in physiology that culminate in the ability to execute complex enzvinatic functions [18-22] The ability of a macrophage to perform certain specific functions, possibly including the degradation of SAA, may therefore be dependent on its activation state. In light of the fact that the appearance of endogenous AEF is directly correlated with the deposition of anyloid fibrils [5] and our preliminary observation that splenic macrophage surface Ia expression is decreased during the induction of amyloid fibril deposition (unpublished results), we suspected that AEF may indeed affect the ability of the macrophage to be appropriately activated for the degradation of the amyloid precutsor, leading to the deposition of amyloid fibrils

Because the mechanism by which SAA and macrophages interact remains  $100^{11}$  nown, we focused our studies on functions known to be enhanced in the activated macrophage Macrophage phagocytic and microbicidal activity was monitored by the in vivo uptake and killing of the facultative intracellular pathogen *L. monocytogenes*, which has been shown to be processed by macrophages [14]. Our results indicate that AEF, whether administered alone or in conjunction with AgNO<sub>3</sub>, does not affect either capacity.

Primed and activated macrophages exhibit increased release of reactive oxygen metabolites [23] Treatment of inimals with an inflammatory agent, such as AgNO<sub>3</sub>, results in enhanced macrophage oxidative metabolism [15] Our experiments measuring reactive oxygen metabolite production by splenic macrophages reveal that XEF damps  $O_2^-$  production by AgNO<sub>4</sub>-induced macrophages and may similarly affect  $H_2O_2$  production. At this point, one can only speculate about the mechanism by which AEF modulates  $O_2^-$  production. It is conceivable, however, that AEF inhibition occurs at the level of membrane components of the phagocyte superovide-generating system, for example, by interfering with NADPH oxireductase evidebrome i activity. When AFF was administered in the absence of AgNO<sub>3</sub> no significant modification in metabolite release was observed as compared with teady state, suggesting that the downregulatory action of AEF is dependent on a concomitant inflammatory stimulus. It is probable that AEF either acts in concert with other factors induced or released during an inflammatory response er exerts an inhibitory effect only on macrophages rendered responsive through inflammatory stimuli. This would explain why AEF triggers XA fibril deposition only when administered in conjunction with an inflammatory stimulus.

Having thus determined that AFE may selectively affect • some macrophage activation-associated functions but not others, we chose to examine the effect of ALL on macrophage activation at the phenotypic level. Increased MHC class II (Ia) surface expression is considered a marker of macrophage activation [18, 24, 25] and is therefore useful for monitoring macrophage activation status. We first compared surface Ia expression by splenic macrophages haivested from inice subjected to amyloid deposit-inducing and -noninducing treatment protocols. Our data indicate that AgNO<sub>3</sub> treatment alone ineffective in causing amyloidosis, results in splenic macrophage activation as reflected by enhanced Ia expression over normal steady state. This increased expression is damped or blocked in the presence of AEF. The absence of any such 'block" in Ia expression in mice treated with AEF-free spleen extract (obtained from control, untreated mice) in conjunction with AgNO3 suggests that decreased Ia expression is attributable to AEF and not some contaminant in our AEF preparation. Furthermore, it is interesting to note that AEF treatment alone revealed a slight increase in Ia expression levels over normal steady state. As indicated by our respiratory burst studies. AEF seems to exert an inhibitory effect only in the event o' a concomitant inflammatory stimulus Again, this observation may be related to the fact that ALF is amyloidogenic only when administered in conjunction with an inflammatory stimulus

Having thus linked decreased splenic macrophage Ia expression to AEF, we looked at whether the observed suppression of macrophage Ia expression in the presence of AEF played an integral role in altered SAA degradation leading to AA fibril deposition By counteracting AEF induced down-regulation of splenic macrophage la expression with IFN- $\gamma$ , we were able to show that the effect of ALF on macrophage activation status during the induction of amyloidosis at the level of Ia expression is not necessary for the production and deposition of AA fibrils in the spleen. It is possible that IFN- $\gamma$  increases Ia expression by a different pathway than that which may be blocked by AEF Studies of the regulation of macrophage surface Ia expression suggest that at least four activation-associated transductional signal cascades (e g , cAMP cascade, Ca2+, and protein kinase C cascades) may be important for its regulation [18, 26]. It is therefore conceivable that Ia expression could be restored by IFN- $\gamma$  despite continued inhibition of an alternative cascade by AEF Should this alternative transduction cascade also mediate SAA degradation (i.e., by transcriptional regulation of SAA metabolic enzyme genes), the end result could be that AEF blockage of one macrophage activation pathway (causing aberrant SAA degradation) could persist despite replenished Ia expression due to IFN- $\gamma$ 

These studies have established that, in the presence of an inflammatory stimulus, AEF selectively modifies certain aspects of the activated splenic macrophage. It is probable  Lat AEF induces anyloid fibril deposition by altering the acrophage function responsible for normal or aberrant egradation of SAA. Studies such as those described here will prove most revealing when the exact nature of this function has been characterized.

## ACKNOWLEDGMENT

We thank Mrs. Ping Wang for her technical help. This work was supported by a grant from the Medical Research Counil of Canada (MT-10250).

## REFERENCES

- 1 Hoffman, J.S., Eucsson, I.H., Eriksen, N. Walsh, K.V., Benditt, E.P. (1984) Murine tissue annyloid protein AV. NH5 terminal sequence identity with only one of the two serum amyloid protein (apo-SAA) gene products  $J_{-}Ixp_{-}Med_{-}159$ , 641-646
- 2 Meek, R.L., Hoffman, J.S. Benditt, F.P. (1986) Anyloidogenesis one serum anyloid A isotype is selectively removed from the circulation. J. Exp. Med. 163, 499-510.
- 3 Glenner, G.G. (1980) Amyloid deposits and amyloidosis -- the beta-thrilloses I. N. Ingl. J. Med. 302, 1283-1292
- 4 Cohen A.S., Cathcart, E.S. (1972) Casein-induced experimental anyloidosis Review of the cellular and immunological aspects Methods Ichuv Exp. Pathol. 6, 207–242
- 5 Axelrad, M., Kisilevsky, B. (1980) Biological characterization of anyloid-enhancing factor. In *Amploid and Amyloidous*. Proc. 3rd Symposium on Amyloidosis (G.G. Glenner, P.P. Costa, and F.F. deFreitas, eds). Excerpta Medica, Amsterdam, 527-533.
- 6 Axelrad, M., Kisilevsky, R., Willmen, J., Chen, S.J., Skinner, M. (1982) Further characterization of anyloid-enhancing factor Lab Invest 47, 139-146
- 7 Kisilevsky, R., Boudreau, L. (1983) Kinetics of anyloid deposition. The effects of anyloid-enhancing factor and splenectomy. *Lab. Invest.* 48, 53–59.
- 8 Wohlgethan, J.R. Cathcart, F.S. (1979) Anyloid in A/J mice is determined by a single gene. *Nature* 278, 453-454
- 9 Gervais, F., Hebert, I., Skamene, E. (1988) Amyloid enhancing factor production and response in amyloid-susceptible and resistant strains J. Leukoc. Biol. 43, 311-316
- 10 Gueft, B, Ghidoni, J J (1963) The site of formation and ultrastructure of amyloid Am J Pathol 43, 837-854
- 11 Lavie, G, Zucker-Franklin, D, Franklin, EC (1980) Elastasetype proteases on the surface of human blood monocytes possi-

ble role in amyloid formation. J. Journalet, 125, 1 5 480.

- 12 Skogen, B. Thorsteinsson, I. Nuvig. J.B. (1980) Degradation of protein SAX to an AX like fragment by enviros of the monocytic origin. Solut. J. Interface, 11, 533–540.
- 13 Euks, V., Zucker Er inklin, D. (1985). Impared Kuppler cell function precedes development of secondary invloidosis. J. Exp. Met. 161, 4013–4058.
- 14 Berche, P., Guillard J.L., Suisonetti P.J. (1987). Intracellular growth of *Lateral meteorherica* as a prerequisite for in vivo in duction of Feell mediated immunity. *J. Imm. Soc.* 138, 2206–22771.
- 15 Babior BAL (1984) Oxidants from phagocytes agents of defence and destruction Blue 64, 959 966
- Nathan C.I. (1987) Secretors products of macrophases *J. Clim. Incest.* 79, 319–326.
- 17 Ecolowitz, R. V.B., Austyn, J., Stuhl, P.D., Gordon, S. (1982). Surface properties of the bacillus Calmette Guerni activited mouse macrophase. Reduced expression of mannosc specific endocytosis. Fe receptors, and antiech 14:80. accompanies in duction of 157 / 159. Met. 154, 60. 76.
- 18 Adams D.O. Hamilton, T.O. (1987) Molecular transductional mechanisms by which H.N. and other signals regulate macrophage development. *Internuoil Res.* 97, 5–27.
- Puchtler H. Sweit, F. Levine, M. (1962) On the binding of Congo Red by anyloid *J. Histochem.* (*Moham.* 10, 35) 364.
- 20 Cohen, H. J., Chovancie, M.L. (1978) Superoxide generation by digitonin stimulated gime epig granulocites. A basis for a continuous assay monitoring superoxide production and for the study of the activation of the generating system. J. Clin. Presst 61, 1088–1096.
- 21 Bradford, M. (1976). Vrapid and acoustive method for the quantitation of microgram quantities of protein utilizing the principle of protein dve binding. *Inal. Basehom.* 72, 248–251.
- 22 Ho, M.K., Springer T.N. (1982) Mac1 antigen quantitative expression in macrophage populations and tissues, and un minoflucrescent localization in the splecie. *J. Immunol.* 128, 2281–2286.
- 23 Johnston, R.B., Jr. (1981) Enhancement of phagocytosis associated oxidative metabolism a remainlestation of macrophage activation. *Lymphokines*, 3, 33.
- 24 Lahise, W.P. (1991) Molecular biology of murine MHC class II genes. Crit. Rev. Immunol. 14, 467–193.
- 25 Nibbering, P.H., Eangermans, J. V.M., Van de Gevel, J.S., Vin der Hulst, M.B., Van Furth, R. (1991) Nutite production by activated murine macrophages correlates with their toxoplas mastatic activity. Ta antigen expression, and production of H<sub>2</sub>O<sub>2</sub>. *Immunobiology* 184, 93-105.
- 26 Uhing, R.J., Adamis, D.O. (1989) Molecular events in the activation of murine macrophages. Agents Actions 26, 9-14.